

CONSENT CALENDAR
July 26, 2022

To: Honorable Mayor and Members of the City Council

From: Dee Williams-Ridley, City Manager

Submitted by: Don E. Ellison, Interim Director of Human Resources

Subject: Memorandum Agreement: International Brotherhood of Electrical Workers

Local 1245

RECOMMENDATION

Adopt two Resolutions:

- 1. Adopting a successor three-year Memorandum Agreement (MA) with the International Brotherhood of Electrical Workers Local 1245 (hereinafter referred to as the "Union") with a term of July 1, 2022 through June 30, 2025, authorizing the City Manager to execute and implement the revised terms and conditions of employment set forth in the Memorandum Agreement; and
- 2. Approving a new salary resolution for Representation Unit C that implements the new salary levels negotiated in the new labor agreement and rescinding Resolution No. 69,962-N.S.

FISCAL IMPACTS OF RECOMMENDATION

The terms of the new MA provide for a 3% salary increase effective the first full pay period after Council adoption; 3% effective July 9, 2023; and 1% July 7, 2024, for a cumulative wage increase of 7% over three (3) years; and one-time lump sum payment of \$1,000. The cost for the new MA is approximately \$221,895 over the term of the agreement (Fiscal Year 2023 through Fiscal Year 2025).

The MA also provides an elimination of the PEPRA CalPERS Cost Share of 1.25% and 3% premium pay to an employee who possesses and maintains a "General Electrician" state certification from the Department of Industrial Relations.

The funding for this MA comes from the General Fund and other special revenue funds. The cost of this labor contract will be included in the fiscal years 2023 through 2025 Budgets.

The terms of the new Memorandum Agreement include a "me too" clause that if the City reaches agreement with the largest bargaining unit or extends the Unrepresented employees an across the board Cost of Living Adjustment (COLA) increase, wage

CONSENT CALENDAR July 26, 2022

increase, lump sum payment, or other benefit increase during the term of this agreement, the City agrees to apply the same increases to the Union.

CURRENT SITUATION AND ITS EFFECTS

The City's labor contract with the Union expired and was fully terminated as of June 30, 2022. In an effort to reach agreement on a successor Memorandum Agreement, representatives of the City and representatives of the Union held approximately 7 negotiating sessions beginning in April 14, 2022. The parties reached agreement on all outstanding economic issues on July 13, 2022.

While the labor contract expired and fully terminated on June 30, 2022, the laws governing collective bargaining agreements provide that the terms and conditions set forth in the expired contract remain in full force and effect until modified through the collective bargaining process. The collective bargaining process has now been completed and the parties have reached tentative agreement on all outstanding issues.

BACKGROUND

There are 15 budgeted positions represented by IBEW Local 1245 in one representational unit (Unit C). IBEW Local 1245 represents employees in the electrical occupations in the Public Works Department.

The City's philosophy during negotiations with IBEW Local 1245 was to follow City Council policies to protect the City's short and long-range economic health. City Council policies for labor negotiations include, but are not limited to, the following:

- Assure that labor organizations and their members are treated fairly and with respect.
- 2. Negotiate in good faith and within the process established by the parties including honoring the traditional confidentiality of the negotiation process.
- 3. Negotiate contracts based on a "Total Compensation Package" model (changes in current and future salary, and health and welfare benefits) within the City's overall financial conditions.

Major provisions of the new labor contract are as follows:

Term	July 1, 2022 through June 30, 2025
11 Salary Increase	3%, 3%, 1%; \$1000 lump sum
37.11 PEPRA Ramp	Elimination of the PEPRA Pension contribution of 1.25% as
Down 1.25	soon as administratively possible

ENVIRONMENTAL SUSTAINABILITY AND CLIMATE IMPACTS

There are no identifiable environmental effects or opportunities associated with the subject of this report.

CONSENT CALENDAR July 26, 2022

RATIONALE FOR RECOMMENDATION

Representatives of the City worked within policies set by the City Council for guiding contract negotiations and staff met with the City Council in closed session to discuss and receive the policy direction and economic authority to settle this contract. The overall settlement must be within the City's ability to pay based on projected revenue as well as demands for services across the spectrum of programs the City provides the community. I believe this settlement is in keeping with City Council's direction to staff and is fair and equitable to the members of the Union.

ALTERNATIVE ACTIONS CONSIDERED

None

CONTACT PERSON

Don E. Ellison, Interim Director of Human Resources, Human Resources, 510-981-6807

Attachments:

- 1: Resolution Revised Memorandum Agreement: IBEW Local 1245
 - Exhibit A: Memorandum Agreement between the City of Berkeley and the International Brotherhood of Electrical Workers Local 1245 A.F.L. C.I.O. July 1, 2022, 2022 through June 30, 2025 (Edited Version)
 - Exhibit B: Memorandum Agreement between the City of Berkeley and the International Brotherhood of Electrical Workers Local 1245 A.F.L. C.I.O. July 1, 2022 through June 30, 2025 (Clean Version)
- Resolution: Approving a New Classification and Salary Resolution for International Brotherhood of Electrical Workers and Rescinding Resolution No. 69,962-N.S. Exhibit A: International Brotherhood of Electrical Workers, Local 1245 Classification/Salary Resolution Period: August 7, 2022 through June 30, 2025
- 3. Tentative Agreement Between the City of Berkeley and the International Brotherhood of Electrical Workers Local 1245 A.F.L. –

Page 4 of 183

RESOLUTION NO. ##,###-N.S.

MEMORANDUM AGREEMENT: INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 1245

WHEREAS, the City is obligated under the provisions of California Government Code Section 3500 – 3510, commonly referred to as the Meyers-Milias-Brown Act, to meet and confer in good faith and attempt to reach agreement with representatives of recognized bargaining units on matters within the scope of representation including, but not limited to wages, hours and other terms and conditions of employment; and

WHEREAS, representatives of the City and the International Brotherhood of Electrical Workers, Local 1245 have met and conferred in good faith and have reached agreement on a new Memorandum Agreement that incorporates all changes and modifications in wages, hours and other terms and conditions of employment agreed to by the parties.

NOW THEREFORE, BE IT RESOLVED by the Council of the City of Berkeley that the City Manager is hereby authorized to execute the new Memorandum Agreement for the period July 1, 2022 through June 30, 2025 with the International Brotherhood of Electrical Workers, Local 1245, including changes in certain benefits on dates specified in the Memorandum of Understanding which is attached hereto, made a part hereof and marked Exhibit B.

BE IT FURTHER RESOLVED that the City Manager is hereby authorized to execute and implement said Memorandum Agreement including all changes in wages, hours, and other terms and conditions of employment. A fully executed original of said contract is filed in the Office of the City Clerk.

BE IT FURTHER RESOLVED that this revised resolution provides for amendments to the Memorandum Agreement adopted on July 26, 2022.

Exhibit A: Memorandum Agreement between the City of Berkeley and the International Brotherhood of Electrical Workers Local 1245 – A.F.L. – C.I.O - July 1, 2022 through June 30, 2025 (Edited Version)

Exhibit B: Memorandum Agreement between the City of Berkeley and the International Brotherhood of Electrical Workers Local 1245 – A.F.L. – C.I.O. – July 1, 2022 through June 30, 2025 (Clean Version)





MEMORANDUM AGREEMENT

Between the

CITY OF BERKELEY

and the

INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 1245 - A.F.L. - C.I.O.

June 28, 2020 July 1, 2022 to June 30, 2022 2025

Page 6 of 183

RESOLUTION NO. 68,670-N.S.

MEMORANDUM AGREEMENT: INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 1245

WHEREAS, the City is obligated under the provisions of California Government Code Section 3500 – 3510, commonly referred to as the Meyers-Milias-Brown Act, to meet and confer in good faith and attempt to reach agreement with representatives of recognized bargaining units on matters within the scope of representation including, but not limited to wages, hours and other terms and conditions of employment; and

WHEREAS, representatives of the City and the International Brotherhood of Electrical Workers Local 1245 have met and conferred in good faith and have reached agreement on a new Memorandum Agreement that incorporates all changes and modifications in wages, hours and other terms and conditions of employment agreed to by the parties.

NOW THEREFORE, BE IT RESOLVED by the Council of the City of Berkeley that the City Manager is hereby authorized to execute the new Memorandum Agreement for the period October 21, 2018 through June 27, 2020 with the International Brotherhood of Electrical Workers Local 1245, including changes in certain benefits on dates specified in the Memorandum Agreement which is attached hereto, made a part hereof and marked Exhibit A.

BE IT FURTHER RESOLVED that the City Manager is hereby authorized to execute and implement said Memorandum Agreement including all changes in wages, hours, and other terms and conditions of employment. A fully executed original of said contract is filed in the Office of the City Clerk.

The foregoing Resolution was adopted by the Berkeley City Council on November 27, 2018 by the following vote:

Ayes: Davila, Droste, Hahn, Harrison, Maio, Wengraf, Worthington and Arreguin.

Noes: None.

Absent: Bartlett.

Jesse Arreguin, Mayor

Attest:

Mark Numalnville, City Clerk

Page 7 of 183

Table of Contents (click section titles to jump to page)

ARTICLE 1 - ADMINISTRATION	1
SECTION 1: RECITALS	1
SECTION 2: PARTIES TO THE AGREEMENT	2
2.1 Exclusive Representation	2
2.2 Management Rights	2
SECTION 3: NO DISCRIMINATION	2
SECTION 4: UNION SECURITY	3
4.1 Union Security	3
4.3 New Employee Orientation	3
SECTION 5: DEDUCTION OF UNION DUES	5
5.1 Deductions	5
5.2 Revocation	5
5.3 Authorization Forms	5
5.4 Notice	5
SECTION 6: UNION REPRESENTATIVES	
6.1 Number of Representatives	6
6.2 Maximum Representatives	
6.3 Notification	6
6.4 Union/Management Meetings	6
6.5 Union Training	6
SECTION 7: SEPARABILITY OF PROVISIONS	7
SECTION 8: EXISTING MEMORANDUM AGREEMENT	7
SECTION 9: FINALITY OF RECOMMENDATIONS	7
SECTION 10: DURATION	7
ARTICLE 2 - SALARIES, HOURS OF WORK AND COMPENSATION ISSUES	7
SECTION 11: SALARIES	8
11.1 Salary Rates	8
11.2 Step Placement	<u>8</u> 8
11.3 Maximum Salary Rate	9
11.4 Effect of Leaves	9
11.6 Y-Rate	9
11.7 Pay Frequency	9
11.8 Higher Classification	10
11.9 Temporary Appointment	11
11.10 Step Increases for Provisional Employees	11
11.11 End of Year Pay Period	11
11.12 Overtime Underpayment.	11
11.13 Equity Studies	11
11.14 Step Increase	11
11.15 Effective Date of Salary Adjustment	<u>11</u> 44
11.16 Reclassification	
11.17 Premium Pay Differential - Working at Heights of 60 Feet or More	12

Page 8 of 183

11.18 Commercial Driver's License Premium	12
11.19 Longevity Pay	<u>1212</u>
SECTION 12: HOURS AND DAYS OF WORK	
12.1 Hours and Days of Work	
12.2 Sunday and Graveyard Shifts	
12.3 Shift Assignment	
12.4 Classes during Shift Assignments	
12.5 Rest Period	
12.6 Daylight Saving Time	
SECTION 13: OVERTIME	
13.1 Rate	<u>15</u> 14
13.2 Department Head Discretion	<u>15</u> 14
13.3 Payoff of Overtime	15
13.4 FLSA Workweek	15
13.5 Emergency Overtime	15
13.6 Meals	<u>16</u> 15
13.7 Standby Pay	16
13.8 Emergency Overtime Assignments	<u>17</u> 16
13.9 Overtime Offers	
SECTION 14: SHIFT DIFFERENTIAL	18
14.1 Swing Shift	18
14.2 Night Shift	18
SECTION 15: NATURAL DISASTER/DECLARED EMERGENCY	18
15.1 Call Outside Normal Working Hours	<u>19</u> 18
15.2 Shift Differential	<u>19</u> 18
15.3 Reduction in Hours	<u>19</u> 18
15.4 Change in Assignments	19
15.5 Hours Worked in Excess of 12-hours	19
ARTICLE 3 - LEAVES	19
SECTION 16: VACATION	19
16.2 Vacation Approval	19
16.3 Use of Incremental Vacation	<u>20</u> 19
16.4 Accrual Rate as of December 8, 1987	<u>20</u> 19
16.5 Accrual, Use, and Limitation for Employees with Less than 6-Months of Serv	ice20
16.6 Vacation Eligibility for Part-time, Intermittent, Leave Without Pay or	<u>21</u> 20
Reemployed Employees	<u>21</u> 20
16.7 Maximum Vacation Accumulation	21
16.8 Prorated Vacation due to Extended Absence, Reemployment or Reins	tatement <u>22</u> 21
16.9 Payment of Vacation upon Termination or Extended Leave after Six Months	of Service22
16.10 Calculation and Payment or Reimbursement of Vacation Credits upon T	Cermination or
Extended Leave	
16.11 Lump Sum Payment of Vacation	<u>23</u> 22
16.12 Calculation of Vacation Based on Actual Paid Hours	23
16.13 Vacation Accounting Procedures	23

Page 9 of 183

16.14 Voluntary Time Off	23
SECTION 17: HOLIDAYS	23
17.1 Holidays	23
17.2 Floating Holidays	24
17.3 Floating Holiday Accrual upon Employment	24
17.4 Limitations	<u>25</u> 24
17.5 Holidays for Employees whose Work Week is Other than Monday through Friday	<u>25</u> 24
17.6 Work during a Holiday	
17.7 Holidays for Part-Time Employees	25
SECTION 18: SICK LEAVE	
18.1 Eligibility	<u>2625</u>
18.2 Sick Leave Accrual	
18.3 Pro-rated Sick Leave for Part-Time Employees	26
18.4 Sick Leave for Intermittent Employees	
18.5 Maximum Accumulation	
18.6 Payment upon Retirement/Termination	
18.7 Annual Payment for Hours in Excess of Maximum for Employees Working Hal	
More for Employees Hired on or Before June 30, 2013	
18.8 Restoration of Sick Leave if Reemployed within Two Years	
18.9 Sick Leave Use/Family Illness	
18.10 Prohibition for Use with Outside Employment	
18.11 Notification Requirement	
18.12 Cessation of Accrual	
18.13 Workers' Compensation	
18.14 Light Duty	
18.15 Control Program for Sick Leave Use	
18.16 Bonus Time for Unused Sick Leave	
18.17 State Disability Insurance	32 31
SECTION 19: FUNERAL LEAVE	
SECTION 20: MILITARY AND MARITIME LEAVE	33
SECTION 21: PARENTAL LEAVE	34 33
21.1 Eligibility:	
SECTION 22: LEAVE OF ABSENCE WITHOUT PAY	
22.1 Department Head Authority	
22.2 Union Sponsored Training	
22.3 Exhaustion of Accrued Time	
SECTION 23: JURY DUTY LEAVE	
23.1 Court Time	
ARTICLE 4 - HEALTH AND WELFARE BENEFITS	
SECTION 24: HOSPITAL-MEDICAL, DENTAL, AND VISION COVERAGE	
24.1 Medical Coverage	
24.2 Maximum Medical Premium Payments	
24.3 Dental Coverage	
24.4 Vision Coverage	

Page 10 of 183

24.5 Change in Insurance Carrier	<u>39</u> 38
24.6 Domestic Partner Coverage	
24.7 Part-Time Employees and Prorated Benefits	<u>40</u> 39
24.8 Commencement of Medical, Dental, and Vision Benefits	
24.9 Health Insurance In-Lieu Payments	<u>40</u> 39
24.10 Group Life Insurance	<u>40</u> 39
24.11 Pre-Tax Status	<u>40</u> 39
SECTION 25: RETIREE MEDICAL COVERAGE	<u>41</u> 40
25.1 Amendment of Retiree Health Premium Assistance Plan III, effective January 2	3, 1998,
Restated and Amended effective March 22, 2011	4140
25.2 Eligibility	<u>42</u> 41
25.3 Pre Age 65 Retiree Health Insurance	
25.4 Retiree Benefits for Employees Age 65 and Over	
25.5 Termination by City of Retiree Medical Benefit	
25.6 Retiree Medical Benefit for Employees Retiring Between the Ages of 50 and 55	
25.7 Employees Retiring with a CalPERS Approved Disability Retirement	
25.8 City Funding of Retiree Health Benefit	
25.9 Actuarial Study	
ARTICLE 5 – TERMS AND CONDITIONS OF EMPLOYMENT	474 6
SECTION 26: PROBATIONARY PERIOD	<u>4746</u>
26.1 Duration and Effect of Military Leave on Probationary Period	<u>47</u> 46
26.2 Provisional Appointments to a Higher Classification	
26.3 Report Requirement	
26.4 Rejection during Probationary Period	
26.5 Rejections Subject to Disciplinary Appeal	
26.6 Probationary or Temporary Employee Assignments	
SECTION 27: TRANSFER	
SECTION 28: PROMOTION	<u>48</u> 47
SECTION 29: DEMOTION	<u>49</u> 48
SECTION 30: SUSPENSION	<u>49</u> 48
SECTION 31: DISCHARGE	<u>49</u> 48
SECTION 32: RESIGNATION	<u>50</u> 49
SECTION 33: REINSTATEMENT	<u>50</u> 49
SECTION 34: NOTICE OF TERMINATION	<u>50</u> 49
ARTICLE 6 - GRIEVANCE AND DISCIPLINARY APPEAL PROCEDURE	<u>50</u> 49
SECTION 35: GRIEVANCES	<u>50</u> 49
35.1 Definition	<u>50</u> 49
35.2 Procedure	<u>51</u> 50
35.3 EEO Complaints	<u>54</u> 53
35.4 Compensation Complaints	
35.5 Changes or Interpretations	
35.6 Extension of Timelines	
35.7 Days	
35 8 Grievances Challenging Disciplinary Action	<u></u> 5554

Page 11 of 183

ARTICLE 7 – GENERAL PROVISIONS	<u>5756</u>
SECTION 36: GENERAL PROVISIONS	<u>57</u> 56
36.1 Personal Conduct	<u>57</u> 56
36.2 Rain Gear	<u>57</u> 56
36.3 Tools	<u>57</u> 56
36.5 Training	<u>58</u> 57
36.6 Use of Automobiles	<u>59</u> 58
36.7 Safety	<u>60</u> 59
36.8 YMCA Group Membership	<u>60</u> 59
36.9 Annual Performance Evaluation	
ARTICLE 8 - PUBLIC EMPLOYEES RETIREMENT SYSTEM	<u>61</u> 60
SECTION 37: PAYMENT OF EMPLOYEES' PERS CONTRIBUTIONS	<u>61</u> 60
37.1 Miscellaneous Designation	<u>61</u> 60
37.2 "Classic Employees" Definition	<u>61</u> 60
37.3 "New Member" Definition	
37.4 CalPERS Retirement Formula for "New Members" as Defined Under the	6160
Public Employees' Pension Reform Act of 2013 (PEPRA)	
37.5 CalPERS Retirement Formula and Employer Paid Member Contribution for C	
Employees (i.e. current employees and employees who do not qualify as "New Members" is	
PEPRA)	
37.6 CalPERS New Members Normal Share of Cost	<u>61</u> 60
37.7 Reporting of Contributions	6160
37.8 City Contracts with CalPERS	
37.9 Index Level 1959 Survivor Benefit	
37.10 Classic Members' Pension Contribution through 20516 a Contract Amendment	6362
37.11 New Members' Pension Contributions through 20516 a Contract Amendment	
SECTION 38: SUPPLEMENTAL RETIREMENT/DISABILITY INSURANCE PLAN.	
ARTICLE 9 - LAYOFF PROCEDURE	6564
SECTION 39: LAYOFF	6564
39.1 Announcement of Layoff	6564
39.2 Seniority Service Date	6564
39.3 Establishment of Seniority Lists	
39.4 Employee Retreat Rights	
39.5 Employee Notification	
39.6 Flexible Placement Program	
39.7 Reemployment Lists	
39.8 Career Exempt Employees	
39.9 Appeal Procedures	
39.10 Reemployment Priority Rights	
SIGNATURE PAGE	
EXHIBIT A - Hourly Salary Schedule	
EXHIBIT C	
EXHIBIT D – Service Vacation Leave	
EXHIBIT E - HOURS AND DAYS OF WORK	

Page 12 of 183

EXHIBIT F	- IMPLEMENTING TH	IE REDUCED W	ORK SCHEDULE	<u>79</u> 77
EXHIBIT G				76

MEMORANDUM AGREEMENT Between CITY OF BERKELEY and the INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 1245

ARTICLE 1 - ADMINISTRATION

SECTION 1: RECITALS

- 1.1 This Memorandum Agreement is entered into pursuant to the Meyers-Milias Brown Act (Government Code, Sections 3500-3511), as amended, and has been jointly prepared by the parties.
- 1.2 The City Manager is the representative of the City of Berkeley (hereinafter referred to as "the City") in employer-employee relations as provided in Resolution No. 43,397-N.S., adopted by the City Council on October 14, 1969.
- 1.3 International Brotherhood of Electrical Workers, Local 1245, is the recognized employee organization for Representation Unit C (Electrical Occupations), which organization has been certified as such pursuant to said Resolution No. 43,397N.S. The employee positions in such Representation Unit are hereinafter set forth in Exhibit A attached hereto and made a part hereof, and International Brotherhood of Electrical Workers, Local 1245 (hereinafter referred to as "the Union") is recognized as the sole representative of employees assigned to such positions.
- 1.4 The parties have met and conferred in good faith regarding wages, hours, and other terms and conditions of employment of the employees in said Representation Unit C, have exchanged freely information, opinions, and proposals, and have endeavored to reach agreement on all matters relating to the employment conditions and employer-employee relations of such employees.
- 1.5 This Memorandum Agreement shall be presented to the City Council as the joint recommendation of the undersigned.
- 1.6 Section titles in this Memorandum Agreement are for identification purposes only and are not to be used for the purpose of interpreting either the intent or the meaning of the language of any section.

SECTION 2: PARTIES TO THE AGREEMENT

2.1 Exclusive Representation

The Union is the exclusive representative of all employees within Representation Unit C (Electrical Occupations) and shall continue to be recognized as such unless, in accordance with the provisions of Resolution No. 43,397-N.S. or said Resolution may be amended, the Union is no longer certified as the Recognized Employee Organization for employees in Representation Unit C.

2.2 Management Rights

The Union recognizes that the management of the City and its business and the direction of its working forces are vested exclusively in the City, and this includes, but it not limited to, the following: to direct and supervise the work of its employees, to hire, promote, demote, transfer, suspend, and discipline or discharge employees for just cause; to plan, direct, and control operations; to lay off employees because of lack of work or for other legitimate reasons; to introduce new or improved methods or facilities, provided, however, that all of the foregoing shall be subject to the provisions of this Agreement, arbitration decisions, or letters of agreement, or memorandums of understanding clarifying or interpreting this Agreement.

The City shall not by reason of the execution of this Agreement (a) abrogate or reduce the scope of any present plan or rule beneficial to employees, such as its vacation and sick leave policies or its retirement plan, or (b) reduce the wage rate of any employee covered hereby, or change the conditions of employment of any such employee to his disadvantage. The foregoing limitation shall not limit City in making a change in a condition of employment if such change has been negotiated and agreed to by City and Union.

SECTION 3: NO DISCRIMINATION

The City and Union agree that they will not discriminate against employees based on race, creed, color, ethnicity, ancestry, religion, political affiliation, gender, sexual orientation, age, national origin, marital or domestic partner status, gender identity or gender expression, parental status, pregnancy, disability or medical condition, Acquired Immune Deficiency (AIDS/HIV) or AIDS related condition, or any other status protected by applicable state or federal law, or protected Union activity. Furthermore, the City and Union agree to comply with all applicable federal, state and local laws pertaining to nondiscrimination and equal employment opportunity.

The Union agrees that it will support programs for making members of minority groups and women aware of employment opportunities within the City's jurisdiction, and that it will work with the City to increase recruitment efforts of such minorities and women into City service. The Union recognizes and supports the City of Berkeley's commitment to equal employment opportunity.

SECTION 4: UNION SECURITY

4.1 Union Security

- 4.1.1 An employee included in Unit C, may become and remain a member in good standing of the Union; or pay to the Union an initial fee equal to the regular initiation fee and, thereafter, a monthly fair share service fee equal to the regular monthly union dues and general assessments.
- 4.1.2 **Contracting Out:** The City will not hire any contractors, firm or individual which will result in the layoff of any employees covered under this Agreement.
- 4.1.3 **Subcontracting:** If the City intends to subcontract any electrical or communication work, the City will notify the Union in advance of any bid solicitation whose value is estimated to be \$25,000 or more. The City will notify the Union in advance by mailing a copy of the City Council Bid Notifications Report to the Union Stewards. Information will be sent to the Union at the same time it is forwarded to the City Council and will include nature of project, the name of department requesting the work, and the budget code. Upon written request from the Union, the City will provide a copy of the actual bid specifications.

For electrical or communication work estimated to cost less than \$25,000, the City will provide the Union Stewards a copy of the Council Awarded Contracts Report on a monthly basis.

- 4.2.3 Indemnification: The Union shall indemnify and save harmless the City, its officers and employees, from and against any and all loss, damages, costs, expenses, claims, attorneys' fees, demands, actions, suits, judgments and other proceedings arising out of any claims made by Unit C employees for deductions made in reliance on information provided by the Union.
- 4.2.4 **Union Reports:** The City shall furnish the Union, on a monthly basis, the name, date of hire, salary, classification and work location of all newly-hired employees subject to this Agreement.

4.3 New Employee Orientation

This provision shall apply to new hire employees appointed to classifications covered by this Agreement.

4.3.1 Onboarding: The parties acknowledge that the City provides a new employee orientation (onboarding) to each new employee hired by the City. As such, the Union will be provided with not less than 10 calendar days' advanced notice of the time, date, and location of the onboarding of any new employee represented by the Union. The Union will be given 30-minutes at the start of the new employee onboarding in a room designated by the City for no more than one (1) representative to present Union membership information. The City representative will excuse him or herself during the Union portion of the onboarding. The Union agrees in its portion of the onboarding not to engage in speech that could cause disruption or material interference with City activities.

The City will provide 30 minutes of Union Release Time to the Union representative presenting the Union membership information during the scheduled onboarding. The Union shall provide the Union representative's immediate supervisor with the Union representative's name at least five (5) days prior to the onboarding. The Union representative shall be released for this purpose unless unusual operation needs interfere with such release in which case the Union representative's immediate supervisor will provide a written explanation of why release could not be approved. If the Union representative is not released due to department operational needs, the Union representative may arrange an alternative date and time to meet with the newly hired employee within the first two (2) weeks of employment, subject to the 30-minutes onboarding and Union Release Time requirements as stipulated above.

4.3.2 Information Provided: On a quarterly basis (March, June, September, and December), the City will provide the Union with a digital file via email to the email address designated by the Union. The Union acknowledges and understands that the City is working diligently and in good faith to update its contact information database functionality to incorporate all the fields of contact information listed below. As a result, the City may not initially be able to provide the employee's work telephone number, personal phone number, and personal email address until the completion and implementation of the City's new Enterprise Resource Management Application (ERMA) system.

The City will provide the Union with the following information on file, to the extent the City has it on file:

- Name.
- Job Title.
- Department.
- Work Location.

- Home telephone number.
- Home address.
- Personal cellular telephone number (new hires hired on or after October 1, 2017).
- Work telephone number (upon implementing ERMA).
- Personal email addresses on file with the City (upon implementing ERMA).

Notwithstanding the foregoing, limited to the express purpose of the requirements of Government Code section 3558 only, an employee may opt out via written request to the City (copy to the Union) to direct the City to withhold disclosure of the employee's: Home address; home telephone number; personal cellular telephone number; personal email address; and date of birth.

SECTION 5: DEDUCTION OF UNION DUES

5.1 Deductions

The City shall deduct, once monthly, the amount of Union regular and periodic dues, service fees, or insurance premiums as may be specified by the Union under the authority of written notification by the Union certifying that the employee has provided written affirmed consent for deduction of union membership dues or fair share service fees.

Such deductions, together with a written statement of the names and amounts deducted, shall be forwarded promptly to the Union office.

5.2 Revocation

Upon receipt of a written notice from the Union to the Human Resources Department, the City shall cease deducting membership dues or fair share service fees from the designated employee.

5.3 Authorization Forms

The City may request verification of employee's authorization form giving affirmed consent to deduct dues or fees only if a legitimate dispute arises about the existence or term of the authorization.

5.4 Notice

Union agrees to give reasonable notice of dues change.

SECTION 6: UNION REPRESENTATIVES

City of Berkeley

IBEW Local 1245

6.1 Number of Representatives

The City shall allow two (2) representatives of the Union reasonable time off from work, without loss of compensation or other benefits, to represent its members in disputes which involve the interpretation or application of those rules, regulations, and resolutions which have been or may hereafter be adopted by the City Council to govern personnel practices and working conditions, including such rules, regulations, and resolutions as may be adopted by the City Council to effect memoranda of understanding which may result from the meeting and conferring process, and to represent its members in meeting and conferring in good faith for amendments to this Memorandum Agreement in the future, subject to the conditions set forth in Sections 6.2 (Maximum Representatives) and 6.3 (Notification).

All release time shall be recorded on time sheets and time cards with appropriate codes.

6.2 Maximum Representatives

Two (2) Union representatives shall be the maximum number who will be allowed concurrent time off.

6.3 Notification

The representatives shall notify the first non-bargaining supervisor a minimum of one workday in advance before leaving their work assignments except for emergency situations which require the immediate attention of said representatives (an emergency is defined as a safety problem or hazardous condition), but in no case shall an employee leave his or her job without notification of the first non-bargaining supervisor. Such request shall include the location, and area of activity, the approximate time needed and the general nature of the union business involved. Such request shall not be unreasonably denied. Failure to comply with notification and time recording rules may be grounds for denial of leave.

6.4 Union/Management Meetings

Quarterly meetings shall be held between the union and management. Release time shall be granted for up to two union representatives. Agendas for such meetings shall be set one week in advance between the Business Representative of the Union and the City Manager or their designees. Meetings within department may be held at the discretion of the department head.

6.5 Union Training

Conditional upon prior approval of the course content and upon receipt of certification of completion, if the employee is an elected official or steward of the Union, the City shall reimburse the employee for up to one-half of his or her regular work time spent in such training at the employee's permanent rate of pay, not to exceed twenty (20) hours of paid leave in a calendar year.

SECTION 7: SEPARABILITY OF PROVISIONS

In the event that any provisions of this Memorandum Agreement are declared by a court of competent jurisdiction to be illegal or unenforceable, that provision of this Memorandum Agreement shall be null and void, but such nullification shall not affect any other provisions of this Memorandum Agreement, all of which other provisions shall remain in full force and effect.

SECTION 8: EXISTING MEMORANDUM AGREEMENT

Pursuant to Section A of the Memorandum of Understanding dated August 9, 1972 by and between the City and the Union, this Memorandum Agreement shall supersede all existing Memoranda of Understanding between the City and the Union. There is no guarantee that working conditions and practices will be continued if they are not included in this Memorandum Agreement or have not been or are not hereafter specifically authorized by ordinance or by resolution or the City Council.

SECTION 9: FINALITY OF RECOMMENDATIONS

The recommendations set forth herein are final. No changes or modifications shall be offered, urged, or otherwise presented by the Union, or the City Manager prior to October 20, 2018; provided, however, that nothing herein shall prevent the parties to this Memorandum Agreement from meeting and conferring and making modifications herein by mutual consent.

SECTION 10: DURATION

This Memorandum Agreement shall be effective upon ratification by the City Council except for those provisions which have been assigned other effective dates as herein set forth, and shall remain in full force and effect up to and including June 30, 20252. This Memorandum Agreement and all its rights, obligations, terms and provisions shall expire and otherwise be fully terminated at midnight June 30, 20252.

ARTICLE 2 - SALARIES, HOURS OF WORK AND COMPENSATION ISSUES

SECTION 11: SALARIES

11.1 Salary Rates

Salary rates for the period of June 28, 2020July 1, 2022 through June 30, 20222025, for all classes of positions in Representation Unit C shall be set according to the classifications and salary ranges assigned to those classifications listed in Exhibit "A" and attached hereto and made part hereof.

11.1.1 **Salary Increase upon Council Approval:** Effective July 25, 2021, tThe salary ranges for those the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of four percent (4.0%) and shall be show in Exhibit A.as follows:

Effective the first full pay period following council approval, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of three percent (3.0%).

Effective July 9, 2023, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of three percent (3.0%).

Effective July 7, 2024, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of one percent (1.0%).

11.1.2 **One-Time Payment**: A onetime non-pensionable lump sum payment of \$1,000.001,000.00 for each career bargaining unit employee in this unit shall be conferred and paid on August 13, 2021the first full pay period after council approval. Eligible employees must be current City employees in paid status for the full pay period this lump sum is conferred and shall not apply to unit members who are in unpaid status. Such payment shall be subject to required payroll tax deductions. The parties agree that tThis lump sum provision does not create or bind the City to any precedent or past practice.

If during any fiscal year during the term of this agreement, the City reaches agreement with the largest bargaining unit or extends the unrepresented employees an across the board Cost of Living Adjustment (COLA) increase, wage increase, lump sum payment, or other benefit increase the

City agrees to apply these same increases to the salary ranges and/or benefits for those classifications covered by this MA as listed in Exhibit B.

11.2 Step Placement

Employees occupying a position in the competitive service shall be paid a salary or wage within the range established for that position's class as set forth in Exhibit "B". The minimum rate for the class shall apply to employees upon original appointment to the position except as may be determined by the City Manager within the negotiated wage ranges. Employees who have been laid-off and are rehired to the same classification shall be placed on the same wage step they were in when laid-off. Transfers shall not affect an employee's salary rate. Employees appointed to any of the positions set forth in Exhibit "B" and employed or working on a part-time basis shall be paid in proportion to the time worked and described in their appointment.

11.3 Maximum Salary Rate

No salary advancement shall be made so as to exceed the maximum rate established for the class to which the advanced employee's position is allocated. Advancement shall be in accordance with the compensation plan of the City and shall depend upon increased service value of an employee to the City as exemplified by recommendations of the department head, performance record, special training, length of service, and other pertinent evidence.

11.4 Effect of Leaves

An employee's pay increase shall not be affected by any leave of absence without pay if the employee is off the payroll for less than one hundred sixty (160) consecutive hours. If the employee is off the payroll for more than one hundred sixty (160) consecutive hours, the total amount of time off shall be made up before the employee shall be entitled to such pay increase.

11.6 Y-Rate

Any employee occupying a position which is reallocated to a class, the maximum salary for which is less than the incumbent's present salary, or occupying a position in a class, the salary rate or range for which is reduced, shall continue to receive his present salary. Such salary shall be designated as a Y rate. When an employee on a Y rate vacates his or her position, subsequent appointments to that position shall be made in accordance with Section 11.1 (Salary Rates).

11.7 Pay Frequency

Payment of salaries shall be made bi-weekly. Each pay period shall begin at 12:01 a.m. Sunday up to and including 12:00 midnight Saturday two weeks following. Each payment shall be made not later than the Friday following the end of each

payroll period and shall include payment for all earnings during the previous payroll period.

- 11.7.1 **Full-Time:** Bi-weekly payment to full-time employees shall be made on the basis of the hourly rate, as set forth in Exhibit B, multiplied by the number of hours worked during the pay period.
- 11.7.1.1 **40 Hour Week**: For employees on a forty (40) hour week schedule, the annual salary shall be the product of the hourly salary times 2,080 hours.
- 11.7.1.2 **37.5 Hour Week**: For employees on a thirty seventhirty-seven and one-half (37.5) hour work week schedule, the annual salary shall be the product of the hourly salary times 1,950 hours.
- 11.7.1.3 **Monthly Salary**: The monthly salary shall be the quotient of the annual salary divided by 12.
- 11.7.2 **Part-Time:** Bi-weekly payment to part-time employees in a class for which a monthly salary rate has been herein established shall be made on the basis of the hourly rate multiplied by the number of hours worked during the pay period. The hourly rate for such part-time employees shall be computed in the same manner as for full-time employees.
- 11.7.3 **Intermittent:** Bi-weekly payment to intermittent employees in a class for which a monthly salary rate has been established shall be made on the basis of the hourly rate multiplied by the number of hours worked during the pay period. The hourly rate for such intermittent employees shall be computed in the same manner as for full-time employees.
- 11.7.4 **Bi-Weekly Daily Rate:** Bi-weekly payment to employees in a class for which a daily rate rather than a monthly rate has been herein established shall be made on the basis of the actual days worked during the biweekly pay period multiplied by the daily rate.
- 11.7.5 **Bi-Weekly Hourly Rate:** Bi-weekly payment to employees in a class for which an hourly rate rather than a monthly rate has been herein established shall be made on the basis of the hourly rate multiplied by the number of hours worked during the pay period.

11.8 Higher Classification

The Department Head will work all employees within their career classifications. The departments may specifically assign an employee to work in a higher class. Such assignments shall be in writing and shall indicate the reasons, length and

duties of the assignment. Assignments over one (1) week shall be approved in advance by the City Manager or his or her designee. Blanket authorizations for department will be issued on a yearly basis for applicable positions. To be eligible for higher-class pay, the employee must work a minimum of four (4) hours, meet the minimum qualifications, and perform the duties of the higher class. Employees meeting these requirements will be compensated at the lowest step of the higher classification which provides at least a five percent (5%) differential. The journey level employees will rotate the assignment on a monthly basis and perform the necessary duties. In the event the offer of higher class is declined by the journey level employee assigned for that month, volunteers will be asked according to the rotation list. If there are no volunteers, the journey-level employee assigned for that month will be required to perform the necessary duties.

11.9 Temporary Appointment

When a temporary vacancy is to be filled in a classification for which there is an existing eligibility list, the City shall attempt to make the temporary appointment from that list.

11.10 Step Increases for Provisional Employees

An employee who holds a provisional appointment in a classification shall receive step increases in such classification as if the employee held a permanent appointment thereto.

11.11 End of Year Pay Period

For all salary and benefit purposes, the parties agree that the last day of the last pay period ending in the calendar year shall be the end of the year. For excess leave only, the end of the year shall be treated as the last day of the pay period nearest March 31st.

11.12 Overtime Underpayment

The City will include any payment of overtime, which was inadvertently omitted or miscalculated, in the following pay period's check.

11.13 Equity Studies

A list of comparison jurisdictions is established for the purpose of salary equity studies: Alameda County, Contra Costa County, Oakland, Palo Alto, Richmond, San Francisco, San Jose, San Leandro, Santa Clara, and Vallejo. Other jurisdictions may be added as required by agreement between the parties. Job classifications which fall below the median for these jurisdictions may be reviewed. It is the policy of the City that within available funding limits, equity adjustments which are in the interests of the service will be considered.

11.14 Step Increase

Step increases shall be effective for payroll purposes only on the first day of the pay period nearest to the actual anniversary date. Personnel records will maintain actual dates and will be used to resolve any discrepancies or questions that may arise.

11.15 Effective Date of Salary Adjustment

Any general future salary adjustments will be set to become effective on the first day of the pay period.

11.16 Reclassification

In the event the City reclassifies a position based on an employee initiated review from a lower level classification to a higher level classification, an incumbent occupying such position shall be reclassified without competitive examination provided he or she has performed the duties of the new classification for one (1) year and has not received an unsatisfactory evaluation during that period. All other employees shall be required to pass an examination for the higher classification and shall serve the normal probationary period. If a position is reclassified, the employee shall receive salary compensation retroactive to the date the position description questionnaire is submitted and verified as having been received by the department head. Position description questionnaires will be provided by the City. Employees may request one (1) job audit one (1) year after an initial job audit has been completed on their position. Thereafter one (1) job audit will be permitted every two (2) years.

Reclassification or reallocation of positions shall not be used as a mechanism, the sole purpose of which is to improperly circumvent the provisions of this Agreement, including provisions relating to layoff, transfer, demotion or promotion.

11.17 Premium Pay Differential - Working at Heights of 60 Feet or More

Assignments for working at heights of 60 feet or more will be on a voluntary basis. However, if all qualified employees decline the assignment, the City will be free to contract the work out to an outside service provider.

Employees who work at heights of 60 feet or more, including but not limited to performing maintenance of lights at San Pablo Park, shall receive Height Premium Pay Differential of one and one-half times (1½) times the normal hourly rate for all time performing such work. The Height Premium Pay Differential shall be reported to CalPERS as Height Premium Special Assignment Pay. However, any hours worked on overtime are excluded from CalPERS reported "compensation earnable" in California Government Code Section 20635.

11.18 Commercial Driver's License Premium

Employees required to possess and maintain a valid California Class A or Class B Commercial Driver's License as a condition of employment shall receive three

percent (3%) differential to hourly rate. This differential shall be reported to CalPERS as Special Class Driver's License Pay. However, any hours worked on overtime are excluded from CalPERS reported "Compensation earnable" in California Government Code Section 20636.

11.19 Longevity Pay

Effective the first full pay period after Council approval on its Regular Agenda, employees completing nineteen (19) years of service shall receive a three percent (3.0%) differential beginning with the anniversary date of beginning of the twentieth (20th) year of service and shall apply to all hours in a paid status. This Longevity Pay shall be reported to CalPERS as Longevity Pay Incentive Pay.

11.20 State Certification Premium

An employee who possesses and maintains a "General Electrician" state certification from the Department of Industrial Relations shall receive an additional three percent (3%) to their base rate of pay. The City shall retain the right to add the general electrician state certification as a minimum qualification for the Electrician and Lead Electrician classifications. If the certification is made a minimum qualification, active employees, employed in classifications within this unit at the time of implementation, shall be exempt from this qualification for the remainder of their employment.

SECTION 12: HOURS AND DAYS OF WORK

12.1 Hours and Days of Work

Hours and days of work shall be governed by rules established by the City Manager. (At the present time, the hours and days of work are as shown in Exhibit E (Hours and Days of Work)

12.2 Sunday and Graveyard Shifts

Regularly scheduled Sunday shifts and "graveyard" shifts, as defined in Section 14 (Shift Differential), shall be for eight (8) consecutive hours, including up to onehalf (½) hour for lunch.

12.3 Shift Assignment

Within a given classification, shift assignments shall first be offered to employees by classification seniority on a voluntary basis. In the event shift assignments are not filled voluntarily, such assignments will be made on the basis of inverse classification seniority.

12.3.1 Types of Shifts:

- a. **Swing shift** means authorized work schedules regularly assigned in which at least four (4) hours worked are between the hours of 5:00 p.m. and 12:00 midnight of each workday.
- Night shift means authorized work schedules regularly assigned in which at least four (4) hours worked are between the hours of 12:00 midnight and 7:00 a.m. of each workday.
- c. **Day shift** means any authorized work schedules between the hours of 7:00 a.m. and 5:00 p.m.

12.4 Classes during Shift Assignments

If an employee assigned to the swing or graveyard shift wishes to attend classes during the day, the employee shall notify the first non-bargaining supervisor of his or her intention to enroll in classes during the day and the completion date of the classes. The City will review the work program and advise the employee within 14 days whether or not the day classes will interfere with the shift assignments. If no conflict exists, the City will not change the shift assignment until after the semester ends provided that the employee continues to be actively enrolled in the class.

12.5 Rest Period

Each employee shall be entitled to a rest period of fifteen (15) minutes during each one-half ($\frac{1}{2}$) shift worked.

12.6 Daylight Saving Time

- 12.6.1 **Spring**: In the Spring when transitioning to Daylight Saving Time (DST), employees working during the one (1) hour transition from Standard Time to DST will be paid only for actual hours worked. Employees working on a shift which includes the one (1) hour transition may be granted an option by the Department Head or his or her designee, to work an additional hour or use compensatory time, floating holiday, or vacation to make up the lost work hour.
- 12.6.2 **Fall**: In the Fall when transitioning from DST, employees working during the one (1) hour transition will be paid for all hours worked including overtime at one and one-half (1½) times the straight-time rate of pay for hours worked in excess of the regular workweek as set forth in Section 13 (Overtime) of this Agreement.

SECTION 13: OVERTIME

Employees required to work in excess of their basic work week or in excess of 8 hours in one day (excepting voluntary regular schedules of 9 or 10 hours) during any one week shall be compensated for such overtime services as follows:

13.1 Rate

The overtime rate shall be one and one-half (1½) times the regular hourly salary for the first four (4) hours of such excess and at two (2) times the regular hourly salary for the balance of such excess.

13.2 Department Head Discretion

Whether an employee shall be compensated for overtime by compensatory time off or by payment shall be at the sole discretion of the employee's department head.

13.2.1 **Compensatory Time Off:** Compensatory time off may be earned in lieu of overtime pay at the rate of one and one-half (1½) hours for each overtime hour worked up to a maximum of sixty (60) hours of such compensatory time. Accumulation of compensatory time off in excess of sixty (60) hours may be allowed at the discretion of the department head. Utilization of compensatory time shall be at the discretion of the employee's department head. The times during the calendar year at which an employee shall take his or her compensatory time off shall be determined by the department head with due regard for the wishes of the employee and particular regard for the needs of the service. As used herein, sixty (60) hours is equivalent to ninety (90) hours of time off work.

13.3 Payoff of Overtime

In the event that an employee resigns or is terminated, he or she shall be entitled to compensation for his or her accumulated overtime.

13.4 FLSA Workweek

For the purpose of computing overtime, the work week shall be defined as beginning at 12:01 a.m. Sunday and ending at 12:00 midnight Saturday. Any applied benefited leave shall be included in the calculation of overtime.

13.5 Emergency Overtime

Employees who are called from their living quarters for emergency work or duty on days other than normal workdays, or on normal workdays outside of their regular work hours, shall be paid emergency overtime compensation for actual time worked;

provided, however, that in any case of emergency overtime as herein provided the minimum time for which such overtime compensation shall be paid shall be four (4) hours; and provided further that if such overtime work is performed prior to the beginning of the regularly scheduled work period and such overtime continues into such regularly scheduled work period without a break in service, compensation shall be paid only for the actual time worked.

13.5.1 Overtime Earned Rest Period

If an employee has worked four (4) hours or more of overtime between the hours of 10:30 p.m. and 6:30 a.m. preceding the beginning of his/her regular work hours on a workday, he/she shall be entitled, except in an emergency, to a rest period to include two (2) hours of his/her workday without loss of his/her regular straight-time pay. If the overtime work period extends for six (6) hours or more of overtime between the hours of 10:30 p.m. and 6:30 a.m. preceding the beginning of his/her regular work hours on a workday, he/she shall be entitled, except in an emergency, to a rest period to include five (5) hours of his/her workday without loss of his/her regular straight-time pay. In the event that an employee, due to operational need, is required to work during an earned rest period during regular work hours, the employee will be paid for the earned rest period in addition to wages earned for hours worked.

13.6 Meals

The City will provide meals to employees during emergency assignments and when the employees work continuously for two (2) or more hours immediately before or after their regular shift of eight (8) or nine (9) hours for a total of ten (10) or eleven (11) hours. The employee shall earn an additional meal for every four (4) hours of continuous work after the first meal is earned or taken.

The City will reimburse the employees for the costs of the meal up to twenty five dollars (\$25) if no meals are provided.

13.7 Standby Pay

An employee assigned to standby shall receive ten (10) hours of overtime compensation in addition to any overtime worked while on standby.

13.7.1 **Standby Week:** For the purpose of this Section 13.7.1 (Standby Week), a week shall mean the seven (7) consecutive calendar days following assignment to standby service. In the case of emergencies, the Senior Supervisor or Superintendent may assign another employee to complete the standby week for the originally assigned employee. In that case, the ten (10) hours standby compensation will be pro-rated between the employees.

- 13.7.2 **Missed Assignment while on Standby:** If an employee misses an overtime assignment while on standby duty, he or she will not receive the standby pay for that day. If he or she misses a second call during the same standby period, he or she will forfeit all the standby pay for that period.
- 13.7.3 **Consequences:** If an employee develops a pattern of failing to perform the service when called, they will be subject to disciplinary action and the Union reserves the right to appeal any action.
- 13.7.4 **Standby Vehicle:** The City will provide the employee on standby duty with a vehicle. Said employee shall be responsible for taking all reasonable steps to insure the safety of the tools and equipment on that vehicle.

13.8 Emergency Overtime Assignments

- All emergency overtime assignments including calling an employee other than the employee on scheduled standby status will be paid the minimum overtime compensation as defined in Section 13.5 (Emergency Overtime). In the event that the employee on scheduled standby is not called, and another employee has been called to perform the emergency overtime assignment, both the employee who performed the assignment and the employee on scheduled standby status will receive the minimum overtime compensation.
- 13.8.1 Scheduled Overtime during Standby: The employee who is on scheduled standby status may perform scheduled overtime tasks on a voluntary basis. The standby premium shall not be pro-rated. If an emergency call occurs during the time that the employee is performing scheduled overtime tasks that call shall be paid at the time and one-half rate and not at the call back rate set forth in Section 13.5 (Emergency Overtime). If other City electrical employees are concurrently performing scheduled overtime tasks, the Senior Electrical Supervisor may, at his or her discretion, assign the work to electrical employees present within the City performing the scheduled overtime tasks at the regular overtime rate of pay. The employee who is on scheduled standby status who chooses to perform scheduled overtime shall defer to other employees who choose schedule overtime assignments pursuant to Section 13.9 (Overtime Offers).

13.9 Overtime Offers

The Department will offer Scheduled Overtime on a voluntary basis to employees with the least amount of overtime worked in accordance with the following provisions:

13.9.1 **Posting of Overtime:** The Department will post a Scheduled Overtime Sign-up List on the Division's bulletin board once every two weeks.

- Employees who wish to be considered for scheduled overtime during that time period must sign the list.
- 13.9.2 **Scheduled Overtime Definition:** Scheduled Overtime shall be defined as an overtime assignment scheduled at least 24 hours in advance.
- 13.9.3 **Overtime Report:** The Department will prepare and post a Report of Overtime Worked for employees on a calendar year basis beginning with a zero balance on January 1. The report will be updated after each pay period.
- 13.9.4 **Overtime Assignments:** As scheduled overtime assignments occur, the Department will offer the assignment to individuals on the list for the applicable time period beginning with the person on the sign-up list with the least amount of overtime worked during the calendar year. Notification of scheduled overtime assignments shall be made during work hours at the work place, and employees must be present at the work place to receive the overtime work assignment.
- 13.9.5 **Overtime Assignment Tie Breaker:** If employees have equal amounts of overtime, the scheduled overtime assignment shall be offered to employees in the same order as their names appear on the sign-up list.

SECTION 14: SHIFT DIFFERENTIAL

14.1 Swing Shift

Employees who regularly work a full shift of seven and a half (7½) hours or more on swing shift as defined in Section 12.3.1 (Types of Shifts), which includes more than four (4) hours between the hours of 5:00 p.m. and 12:00 midnight, shall be paid their regular salary plus seven and a half percent (7½%) of their regular monthly salary per month.

14.2 Night Shift

Employees who regularly work a full shift of seven and a half (7½) hours or more on night shift as defined in Section 12.3.1 (Types of Shifts), which includes more than four (4) hours between the hours of 12:00 midnight and 7:00 a.m. (night shift), shall be paid their regular salary plus ten percent (10%) of their regular monthly salary per month; provided, however, that in the case of any such employee who is regularly assigned to such night-shift work for less than an entire work week, the additional payment shall be made only for the portion of the work week worked on the night-shift assignment.

SECTION 15: NATURAL DISASTER/DECLARED EMERGENCY

If an emergency is formally declared by the City, county, state or national authority:

15.1 Call Outside Normal Working Hours

If an employee is called outside of normal working hours, the employee gets time and one-half (1½) the normal rate of pay for the first whole shift regardless of the number of hours worked. If the employee is not called from home the regular rules apply (i.e., overtime for hours worked above eight in a day).

15.2 Shift Differential

Thereafter, the first eight hours at regular rate and 7½% for hours worked between 5:00 p.m. and midnight; 10% for hours worked from midnight to 7:00 a.m. For hours greater than eight in a shift, the employee gets time and one-half (1½) the normal rate of pay but no shift differential on those hours above eight.

15.3 Reduction in Hours

There will be no reduction in the number of hours in the regular work week schedule.

15.4 Change in Assignments

If an employee is at work during the day when an emergency is declared and is assigned to work a night shift and is sent home, the employee will be compensated as follows:

- 15.4.1 If an employee has been at work for up to but not more than two (2) hours the employee will be paid for two (2) hours at the straight time rate.
- 15.4.2 If an employee has been at work for more than two (2) hours but not more than four (4) hours the employee will be paid for four (4) hours at the straight time rate.
- 15.4.3 If an employee has been at work for more than four (4) hours the employee will be paid for eight (8) hours at the straight time rate.

15.5 Hours Worked in Excess of 12-hours

An employee working more than twelve (12) hours in one continuous shift will be compensated at two (2) times the normal hourly rate.

ARTICLE 3 - LEAVES

SECTION 16: VACATION

16.2 Vacation Approval

The times during the calendar year at which an employee shall take his or her vacation shall be determined by the department head with due regard for the wishes of the employee and particular regard for the needs of the service. Wherever practical, employees working in the same classifications within a division shall be given preference of vacation time by seniority. If the requirements of the service are such that a department head cannot permit an employee within his or her department to take an annual vacation leave or any part of such leave within a particular calendar year, the City Manager may permit such employee to take the deferred vacation during the following year.

16.3 Use of Incremental Vacation

Employees may, with advance supervisory approval, use vacation leave in increments of one hour.

16.4 Accrual Rate as of December 8, 1987

Effective December 8, 1987, the vacation accrual rate shall be modified as follows:

Vacation Accrual Schedule	Authorized Annual Workweeks of Vacation	Vacation Leave Credits in Work Days per Month of Service	Vacation Leave Credits in Hours Earned per Month of Service
Through the first three (3) calendar years of service (except as provided in Section 16.4 (Accrual Rates as of December 8, 1987) and 16.5	2 work weeks	0.833	6.667
Fourth (4 th) through eleventh (11 th) calendar years of service (except as provided in Section 16.5 below)	3 work weeks	1.25	10.00
Twelfth (12 th) through seventeenth (17 th) calendar years (except as provided in Section 16.5	4 work weeks	1.667	13.33
Eighteenth (18 th) through twenty-fourth (24 th) calendar years (except as provided in Section 16.5	5 work weeks	2.083	16.667
Twenty-fifth (25 th) and subsequent calendar years (except as provided in Section 16.5	6 work weeks	2.5	20.00

The authorized annual vacation leave for prior years is set forth in Exhibit D attached hereto and made a part hereof.

16.5 Accrual, Use, and Limitation for Employees with Less than 6-Months of Service Each employee, during that portion of the calendar year in which he or she is originally appointed and during the next succeeding year, shall be entitled to vacation leave credits at the rate of 0.833 work days for each calendar month of service. Each such employee shall be entitled to take during these two (2) calendar years only such annual vacation leave as he or she earns.

16.6 Vacation Eligibility for Part-time, Intermittent, Leave Without Pay or Reemployees

For an employee who has worked on a part-time or intermittent basis or who has been on leave of absence without pay for a total of six (6) months or more or who has been terminated and subsequently reemployed, the actual years of service with the City shall be used for the purpose of computing length of service in determining eligibility for vacation at the three (3), four (4), five (5) and six (6) weeks' rate.

- 16.6.1 **Prorated Vacation for Intermittent or Part-Time Employees:** Employees working on an intermittent or part-time basis who have worked half-time or more in the preceding calendar year without termination of employment shall be entitled to a prorated vacation leave based upon the actual years of service with the City and upon the actual amount of time worked in the preceding calendar year.
- 16.6.2 **Effect of Military Leave on Vacation Eligibility:** For the purpose of computing length of service in determining eligibility for vacation at the three (3), four (4), five (5) or six (6) weeks' rate, time spent on extended military leave shall be counted as time spent in the service of the City.

16.7 Maximum Vacation Accumulation

Employees may accrue vacation earned up to a maximum cumulative total of 320 hours. Once an employee accrues 320 hours vacation, the employee shall not accrue any additional vacation leave hours until his or her vacation leave balance is below the maximum 320 hours. Under no circumstances will an employee be allowed to accrue more than 320 hours vacation leave. As provided below, the City will advise employees and their supervisors when the employee has attained an accumulation of 280 hours. Supervisors should be flexible in granting vacation requests from employees who reach 280 hours. Provided further that with regard to employees who have reached 280 hours, no vacation request by such employee shall be unreasonably denied. No employee shall be denied vacation leave such that it causes him or her to reach the 320 hour limit.

Notification: To assist employees to remain below the 320 hours vacation leave maximum accumulation, the parties agree that no later than October 1 of each

year of this Agreement, the City will provide the Union and Department Heads with a report identifying accumulated vacation hours of all employees. The City also agrees to notify all employees who have accrued 280 hours of vacation leave, as of this date, that they must take vacation leave to reduce their vacation leave to avoid reaching the 320 hour maximum accrual.

16.8 Prorated Vacation due to Extended Absence, Reemployment or Reinstatement

An employee who has returned from extended military leave or any other extended leave of absence without pay or who has been reemployed or reinstated shall be entitled, during the calendar year in which he or she returns to the City service, to a prorated vacation based upon the total years of service with the City and upon the total number of months of actual service with the City during the said calendar year. For succeeding calendar years, his or her vacation leave shall be as provided elsewhere in this Section 16 (Vacation).

An employee who is granted a leave of absence without pay and who is off the payroll for less than one hundred sixty (160) consecutive hours shall be entitled to a full vacation. If such an employee is off the payroll for one hundred sixty (160) consecutive hours, his or her vacation shall be as provided in the preceding paragraph of this Section 16.8 (Prorated Vacation due to Extended Absence, Reemployment or Reinstatement) or Section 16.9 (Payment of Vacation upon Termination or Extended Leave after Six Months of Service).

16.9 Payment of Vacation upon Termination or Extended Leave after Six Months of Service

If an employee, is terminated, or is granted an extended military leave or other extended leave of absence without pay, such employee, or his or her estate, shall be paid for vacation credits in excess of the actual amount of vacation leave taken or such employee, or his or her estate, shall reimburse the City for the actual amount of vacation leave taken in excess of vacation leave credits, as the case may be.

16.10 Calculation and Payment or Reimbursement of Vacation Credits upon Termination or Extended Leave

Upon termination, extended military leave, or other extended leave of absence without pay, vacation leave credits shall be totaled, and the actual amount of vacation leave taken, including any that may have been taken during the year in which the termination, extended military leave, or other extended leave of absence without pay occurs, shall be deducted from the total credits. If the credits exceed the actual amount of vacation leave taken, such employee, or his or her estate, shall be paid for the excess of credits on the basis hereinafter set forth. If the actual amount of vacation leave taken exceeds the credits, such employee, or his or her estate, shall reimburse the City on the same basis.

City of Berkeley

IBEW Local 1245

The basis for such payment by the City or for such reimbursement to the City shall be as follows:

The employee's hourly pay rate at date of termination, extended military leave, or other extended leave of absence without pay, multiplied by the excess of credits over vacation leave actually taken or excess of vacation leave actually taken over credits, as the case may be.

16.11 Lump Sum Payment of Vacation

Upon termination, extended military leave or other extended leave of absence without pay, payment for excess of vacation leave credits shall be made in one lump sum at time of termination, extended military leave, or other extended leave of absence without pay, or as soon thereafter as possible; provided, however, that an employee may elect to use excess vacation leave credits prior to termination, extended military leave, or other extended leave of absence without pay, to the extent permitted by this Section 16 (Vacation), and receive a lump sum payment for the balance of leave credits, if any. An employee, or his or her estate, shall not be paid for vacation leave credits in excess of eight (8) calendar weeks. Notwithstanding the foregoing, accumulated but unearned vacation credit at the time of retirement shall be paid off in a lump sum.

16.12 Calculation of Vacation Based on Actual Paid Hours

All vacation benefits shall be calculated upon actual paid hours of work. This provision shall go into effect immediately upon implementation of necessary data processing and programming changes.

16.13 Vacation Accounting Procedures

The City may revise vacation accumulation provisions in order to standardize accounting procedures without effect on the amount of employees' vacation, subject to review and approval by the union.

16.14 Voluntary Time Off

The City will implement a voluntary time off (VTO) program, which may include a yearend holiday closure. It is understood by the parties that employee participation is completely voluntary and that employees who wish to work during such periods will be permitted to do so.

SECTION 17: HOLIDAYS

17.1 Holidays

Recognized holidays for employees in Representation Unit C shall be:

17.1.1	New Year's Day
17.1.2	Martin Luther King's Birthday (3rd Monday of January)
17.1.3	Lincoln's Birthday
17.1.4	Washington's Birthday (observed on the third Monday in February)
17.1.5	Malcolm X's Birthday (observed on Monday or Friday nearest May 19th)
17.1.6	Memorial Day (observed on the last Monday in May)
17.1.7	Juneteenth National Independence Day – observed on the Monday or
	Friday nearest June 19
17.1.8	Independence Day
17.1.9	Labor Day (observed on the first Monday in September)
17.1.10	Indigenous Peoples' Day (observed on the second Monday in October)
17.1.11	Veteran's Day
17.1.12	Thanksgiving Day
17.1.13	Day after Thanksgiving Day
17.1.14	Christmas Day
17.1.15	Effective January 9, 2000, each employee shall be granted three (3)

17.2 Floating Holidays

Any employee shall be granted three (3) floating holidays each calendar year. The days selected shall be by mutual agreement between the employee and the department head (or his or her designee). In the event mutual agreement cannot be reached on the selection of floating holidays, the employee shall have the three (3) days added to his or her accrued vacation time. Employees may take Floating Holidays in one (1) hour increments.

floating holidays each calendar year.

17.2.1 Additional Floating Holidays

For employees who were required to remain in the workplace from March 17, 2020 - June 1, 2020, the City will provide 8 hours of floating holidays for every 40 hours of regularly scheduled hours worked in the workplace up to a maximum of 32 hours of floating holiday. The City will credit these floating holiday hours in the first full pay period after adoption of the MOU.

City will use a specific pay code for the additional floating holiday, hours will be available June 30, 2021. These additional 32 hours of floating holiday shall have no cash value and may not roll over to vacation if not used by June 30, 2021.

17.3 Floating Holiday Accrual upon Employment

In the first calendar year of employment, an employee hired before May 1 shall be entitled to three (3) floating holidays; and an employee hired on or after May 1, but before September 1, shall be entitled to two (2) floating holidays; and an employee hired

on or after September 1, shall be entitled to one (1) floating holiday in that first calendar year.

17.4 Limitations

No monetary award shall be authorized for unused accumulated Floating Holidays for employees who terminate employment prior to the completion of six (6) months of continuous City service.

17.5 Holidays for Employees whose Work Week is Other than Monday through Friday

Employees whose work week is Monday through Friday shall be allowed all holidays with pay which fall within such work week. Those employees whose work week is other than Monday through Friday shall be entitled to the same number of holidays, with pay, during each calendar year as are allowed to employees whose work week is Monday through Friday. The procedure for allowing holidays for employees whose workweek is other than Monday through Friday shall be established by the City Manager. The provisions of this Section 17.5 (Holidays for Employees whose Work Week is Other than Monday through Friday) are not applicable to intermittent employees.

17.6 Work during a Holiday

An employee required to work on any day which is a holiday for employees whose work week is Monday through Friday shall be paid for the number of hours worked during such day at the rate of one and one-half (1½) times the straight-time rate, based upon his or her regular monthly salary, or shall be granted compensatory time off in an amount equal to one and one-half (1½) times the number of hours worked on such holiday. Any employee who works on Christmas day or Thanksgiving Day shall be paid double time for that day. The hours worked on such holiday and paid at the rate herein provided shall not be credited in computing the hours worked in the week for overtime purposes.

The holiday pay provided for shall be in addition to an employee's regular salary. In the event that the time worked on such a holiday is also overtime, as provided in Section 13 (Overtime) of this Memorandum Agreement, payment will be made for the hours worked either as overtime under said Section 13 (Overtime), or as holiday pay under this Section 17 (Holidays), but will not be made under both Sections.

17.7 Holidays for Part-Time Employees

Regularly scheduled part-time employees 1) who are assigned to a regular schedule of twenty (20) or more hours per week; 2) who have worked for the City as regular or part-time employees for one (1) or more years; and 3) who during such period of time have worked a minimum of 1,000 hours shall be entitled to the holiday pay for the number of hours which such employee would have worked had the holiday not

occurred, provided such holiday occurs on a day which such employee is regularly scheduled to work.

SECTION 18: SICK LEAVE

18.1 Eligibility

Any employee shall be entitled to take sick leave with full pay in case of sickness, disability, or serious illness within the immediate family of the employee in accordance with the provisions of Sections 18.2 (Sick Leave Accrual) to 18.12 (Cessation of Accrual).

18.2 Sick Leave Accrual

Each employee shall be credited with one (1) working day of sick leave with full pay for each month of service.

For the purposes of this Section 18 (Sick Leave), a month of service shall mean thirty (30) consecutive calendar days in the case of employees working on a fulltime or part-time basis, and shall mean 163 hours of work in the case of employees working on an intermittent basis.

18.3 Pro-rated Sick Leave for Part-Time Employees

An employee working on a part-time basis shall be entitled to use earned sick leave only on a pro rata basis; for example if an employee works half-time, he or she shall be paid for time off on sick leave on a half-time basis.

18.4 Sick Leave for Intermittent Employees

An employee who works on an intermittent basis shall be entitled to use earned sick leave only for those days on which he or she would have worked if he or she had not been sick; provided, however, that an employee working on an intermittent basis who works only when called shall be entitled to use earned sick leave only when he or she becomes sick after reporting to work in response to such call.

18.5 Maximum Accumulation

Such sick leave as provided in Section 18.2 (Sick Leave Accrual), when not used, shall be cumulative; but the accumulated unused period of sick leave shall not exceed two hundred (200) working days, regardless of the length of service. When the maximum of two hundred (200) working days has been reached, and thereafter part of said maximum has been used, the used part of said maximum may subsequently be replenished at the applicable rate provided in Section 18.2 (Sick Leave Accrual).

18.6 Payment upon Retirement/Termination

All accumulated sick leave shall be canceled when an employee terminates or is terminated, except as provided below for employees hired on or before June 30, 2013.

- 18.6.1 Employees Hired on or Before June 30, 2013: For employees hired on or before June 30, 2013 who retire or voluntarily terminate with a vested pension, and with between twenty (20) years and twenty-eight (28) years of benefited service shall be entitled to receive payment in an amount equal to thirty eight percent (38%) of their accrued sick leave days up to a maximum of the two hundred (200) day maximum accumulation. Provided further that any employee retiring on permanent disability arising out of and incurred in the course and scope of his employment with the City shall be entitled to receive payment at retirement for thirty eight percent (38%) of accumulated unused sick leave days but not, in any event, more than thirty eight percent (38%) of the two hundred (200) day maximum accumulation.
- 18.6.2 Payment After 28 Years of Service for Employees Hired on or Before June 30, 2013: Employees who were hired on or before June 30, 2013, who terminate with at least twenty-eight (28) years of benefited City of Berkeley service or employees retiring on permanent disability arising out of and incurred in the course and scope of their employment with the City with at least twenty-eight (28) years of benefited service shall be entitled to receive payment in an amount equal to fifty percent (50%) of their accrued sick leave days up to a maximum of two hundred (200) unused sick leave days.

18.6.3 Sick Leave Payout to 401(a) Plan for Employees Hired on or Before June 30, 2013: The City and the Union have met and conferred on an Internal Revenue Code Section 401(a) plan and trust agreement to address the liquidation of sick leave at time of retirement. This plan and trust agreement was originally negotiated with the Berkeley Fire Fighters Association/I.A.F.F. Local 1227 and has been submitted to the Internal Revenue Service for a Determination Letter and a Private Letter Ruling which are pending. If the City receives a positive response from the Internal Revenue Service, the plan and trust agreement will be extended to the employees in the bargaining unit. This will provide the employees with an irrevocable option to defer accrued but unused sick leave at time of retirement into a 401(a) plan or be paid out the balance of the accrued but unused sick leave less withholding of applicable federal and state taxes.

IBEW Local 1245

18.7 Annual Payment for Hours in Excess of Maximum for Employees Working Half-Time or More for Employees Hired on or Before June 30, 2013

Employees who regularly work one-half (½) time or more and who have attained the one hundred fifty (150) day maximum sick leave accumulation shall be entitled to receive payment for one-third (1/3) of the first twelve (12) days of sick leave for which they become eligible but do not use and would otherwise forfeit because of the one hundred fifty (150) day maximum limitation. Determination of eligibility for such payment shall be made on an annual calendar year basis, and payment for such sick leave for any calendar year shall be made not later than January 22 of the following year. Such payment shall be made at the employee's salary rate in effect on the preceding December 31 and shall be made only in units of whole days and not for any fraction of a day.

18.8 Restoration of Sick Leave if Reemployed within Two Years

Accumulated sick leave, which has been canceled by reason of an employee's termination, shall be credited back to such employee if the employee returns to City employment within two (2) years of such termination.

18.9 Sick Leave Use/Family Illness

Sick leave shall not be considered as a privilege which an employee may use at his or her discretion but shall be allowed only in case of his or her sickness or disability or in the case of serious illness within the immediate family of the employee. Not more than fifteen (15) working days in any calendar year may be taken as sick leave because of the illness of a member of the employee's immediate family, except for serious medical conditions covered under the provisions of Administrative Regulation 2.4 (Family Care Leave). The immediate family of an employee, for the purpose of this Section, shall be defined as: spouse, domestic partner, son, daughter, parent or dependent.

18.10 Prohibition for Use with Outside Employment

No sick leave shall be allowed for time off for an injury incurred while working for another employer, provided that such injury is covered by the Workers' Compensation laws of the State of California, or other provision for payment for time off because of such injury is made by such other employer. In the event such injury is not covered by the Workers' Compensation laws of the State of California and no other provision for payment for time off because of injury is made by such other employer, sick leave in accordance with the provisions of this Section shall be allowed only if such outside employment has been approved by the City.

18.11 Notification Requirement

In order to receive compensation while absent on sick leave, the employee shall notify his department head prior to or within four (4) hours after the time set for beginning his or her daily duties, or as may be approved by the head of his or her department. The Union and the City recognize it is advantageous to both parties that calls should normally be made prior to the beginning of the work shift. Leave for non-

emergency doctors' appointments shall be requested in advance. In specific instances the City may by written notice require an employee (s) to call in prior to the beginning of their shift in order to be eligible for sick leave.

18.12 Cessation of Accrual

An employee who is granted a leave of absence without pay and who is off the payroll for less than one hundred-sixty consecutive hours shall receive his or her earned sick leave credit. If the employee is off the payroll for one hundred-sixty (160) consecutive hours or more, he or she shall not earn sick leave credit for each two successive pay periods that he or she is off the payroll.

18.13 Workers' Compensation

All probationary and permanent employees of the City and provisional employees who have worked a total of four (4) months or more for the City, who have suffered any disability arising out of and in the course of their employment as defined by the Worker's Compensation Insurance and Safety Act of the State of California, and who are receiving or shall receive compensation from the insurance carrier for such disability, and during the first seven (7) days after such disability when compensation is not paid by the insurance carrier shall be entitled to remain absent from duty with pay until such time as they are able to return to duty or some other final disposition is made of their case; provided, however, that provisional employees who have worked a total of four (4) months or more for the City shall be entitled to remain absent from duty with pay for the period they would have been permitted to remain provisional employees under their employment at the time of injury.

- All employees, other than probationary and permanent employees and provisional employees who have worked a total of four (4) months or more, shall be entitled to such compensation as may be allowed them by the Workers' Compensation Insurance and Safety Act of the State of California.
 - 18.13.1 Workers' Compensation Payments: Payments from the insurance carrier for disability arising out of and in the course of employment shall be paid to the employee. The amount of such payment or payments shall be deducted from the monies, which the employee received from the City. Payments from the insurance carrier, plus the monies paid to the employee by the City, shall be equivalent to the employee's regular full pay.
 - 18.13.2 Workers' Compensation Leave and Salary Continuation: Payments under the Workers' Compensation law for temporary disability, or a recurrence thereof, arising out of and in the course of employment shall be paid for a period not to exceed 365 days at a maximum payment of the employee's pre-disability net pay but shall not exclude any salary

adjustments to which the employee is entitled. Thereafter, the employee will continue to receive only the temporary disability payments provided under state law, and the City will cease to pay the difference. However, salary continuation payments above the statutorily required temporary disability payments shall not be reported by the City to CalPERS as compensation. No time worked shall be part of the 365 calendar days of salary supplement paid by the City. This change shall not affect employees who are currently off the job with a work-related injury which occurred prior to July 1, 1979.

- 18.13.3 **Calculation:** The City shall continue to calculate salary continuation at pre-disability gross pay. The City may calculate salary continuation payments at pre-disability net pay at such time when they develop the capacity to administer it equitably. Any change in calculation shall not reduce employees' combination of disability payments and salary continuation payments below employees' pre-disability net pay.
 - The change in calculation shall not affect employees who are off the job with a work-related injury prior to the new calculation method being implemented.
- 18.13.4 Workers' Compensation Related Absence of Less than Four (4) Hours: An employee who is absent from work for a medical appointment or physical therapy for less than four (4) hours will have ½ day charged against his or her 365 day maximum period for receipt of salary continuation pre-disability net pay.
- 18.13.5 Workers' Compensation Related Absence of Four (4) Hours or More:
 An employee who is absent from work for a medical appointment or
 physical therapy for 4 hours or more will have 1 day charged against his
 or her 365 day maximum period for receipt of salary continuation
 predisability net pay.
- 18.13.6 **Calculation Based on Actual Paid Hours:** All sick leave benefits shall be calculated upon actual paid hours.

18.14 Light Duty

Assignments for temporarily disabled employees.

An employee who is absent by reason of industrial disability may be returned to work and given temporary light duties within his or her ability to perform. The duration of any such period of temporary work should be determined by the City. Employees should be compensated at the rate of pay of their regular classification while

engaged in such temporary duties, and such work assignments are to incorporate the following provisions:

- 18.14.1 **Modified Duty Accommodation:** The City shall accommodate, when feasible, employees covered by this memorandum under the provisions of workers' compensation, and such work assignments are to incorporate the following provision:
 - 18.14.1.1 The assignment shall be consistent with medical limitations as determined by the physician of record.
 - 18.14.1.2 The Assignment shall be within the City of Berkeley and may include hours and days of work other than the employee's regular assignment.
- 18.14.2 Modified Duty Accommodation for Non-Industrial Disabilities: The City may accommodate an employee disabled with a non-industrial disability by providing a modified work assignment in that employee's classification. To be eligible for such a modified assignment, the employee must provide the Human Resources Department with a medical statement from his or her treating physician that clearly states the medical limitations and abilities of the employee. If modification of that position does not serve the best interests of the City, other classifications may be considered, subject to the approval of the Human Resources Director. Compensation will be provided at the level of the classification in which the temporarily disabled employee works during the disability. The employee must meet standards of satisfactory performance for the duration of the work assignment.
- 18.14.3 Modified Duty for Pregnancy-Related Disability: In the case of a medically certified, pregnancy-related disability, in which the normal duties clearly threaten the health and safety of the employee or the unborn child, the Human Resources Department will endeavor to place the employee in a position which best serves the interest of the City with no loss of pay, but in no event will such placement exceed 5 months in duration.
- 18.14.4 **Light Duty Assignments**: Availability of light duty job assignments shall be discussed upon request at the monthly Labor-Management Meetings.

18.15 Control Program for Sick Leave Use

The City may establish a reasonable program for the control of abuse of sick leave and absenteeism, subject to Union review and comment.

18.16 Bonus Time for Unused Sick Leave

For every six (6) months of perfect sick leave attendance the employee will receive eight (8) hours of bonus time. A leave of absence from work pursuant to workers' compensation is counted as an absence from work in the same manner as sick leave for the purpose of this bonus. This bonus time will be prorated for part-time employees. Such bonus time can be used for any leave purpose covered by this Memorandum Agreement. Such bonus time shall be counted as vacation leave credits for purposes of determining eligibility for carry-over and cash payment.

18.17 State Disability Insurance

Except as provided in 18.17.1.3 below, any employee who is absent due to personal illness for more than 7 calendar days (or for any period of time if hospitalized) may apply for State Disability Insurance Benefits.

- 18.17.1 **State Disability Integration:** After such employee has been absent from work due to personal illness for six (6) consecutive work days, if the employee applies and is eligible for State Disability Insurance, the City shall integrate the employee's pay with the employee's State Disability benefits in the following way:
 - 18.17.1.1 The City will determine the weekly SDI benefit amount based on the amount of wages earned with the City of Berkeley in the SDI base period.
 - 18.17.1.2 The weekly SDI benefit will be subtracted from the employee's normal weekly wages and the amount necessary to bring the total of State Disability plus wages to 100% will be deducted from any accumulated sick leave, vacation leave and compensatory time available to the employee. The integration with vacation leave and compensatory time is optional but will be automatically implemented after sick leave has expired unless written notification is received from the employee, as discussed below.
 - 18.17.1.3 The employee must notify the payroll clerk, in writing, to stop integration of State Disability Insurance payments with vacation leave or compensatory time. Upon receipt of notification, the payroll clerk will cease integration of any future leave for that incident of illness.
 - 18.17.1.4 The employee must show the State of California form (Disability Insurance Notice of Computation) to his or her payroll clerk to verify dates covered by SDI and the amount to

be paid. The employee must inform their payroll clerk of all SDI payments. Any employee entitled to State Disability Insurance shall receive in addition thereto such portion of his or her accumulated leave as will meet, but not exceed, the standard earnings of the employee for his or her normal workweek, up to a maximum of five (5) days.

SECTION 19: FUNERAL LEAVE

In the case of death within the immediate family of an employee (who is not excluded from such benefit pursuant to Chapter 4.04.120 of the Berkeley Municipal Code (Personnel Ordinance) which is attached hereto as Exhibit C (City of Berkeley Municipal Code Chapter 4.04.120) and made a part hereof, such employee shall be entitled to remain absent from duty with pay in order to attend the funeral or memorial service for a period not exceeding three (3) working days or, in the case of a funeral or memorial service conducted out of the State of California, for a period not exceeding five (5) working days. For the purpose of this Section, the immediate family of an employee shall be defined as: husband, wife, domestic partner, mother, father, sister, brother, child, grandmother, grandfather, grandchildren, mother-in-law, father-in-law, brother-in-law, sister-in-law, son-inlaw, and daughter-in-law, step-parent, step-sibling, step-child, aunts and uncles, nieces and nephews.

Leave of absence with pay because of death in an employee's immediate family is allowed for the purpose of attending the funeral or memorial services, and such leave shall not be charged against vacation or sick leave that an employee may be entitled to but shall be in addition thereto. Employees may request, and the City will make reasonable efforts to accommodate requests, for employees to supplement bereavement leave by using accrued vacation, compensatory time, or floating holiday. All accrued leave (and/or sick leave, if applicable) shall be utilized prior to taking a leave of absence without pay.

Funeral Leave for Persons Other than Immediate Family: In special cases, with the approval of the department head, the City Manager may grant a death leave to allow an employee to attend funeral or memorial services because of a death of a person not included within the definition of the immediate family.

SECTION 20: MILITARY AND MARITIME LEAVE

Military and Maritime Leave shall be governed by the provisions of the Federal Uniformed Services Employment and Reemployment Rights Act (USERRA) and any regulations promulgated to implement the Act and the California Military and Veteran's Code.

If an employee voluntarily extends his or her military leave in excess of two (2) weeks, the amount of pay received by the employee for his or her military duty shall be deducted from his or her regular pay for such period.

SECTION 21: PARENTAL LEAVE

21.1 Eligibility:

Employees with one (1) or more years of employment with the City (or equivalent in the case of part-time employees) shall be entitled to parental leave as follows:

- A continuous parental leave of up to one year will be granted to any employee with one year (full-time) or more years of employment with the City (or equivalent in the case of part-time employees) upon the birth of a child or the legal adoption of a child who is five years or younger, provided that:
 - 21.1.1 **Commencement of Parental Leave:** The one year parental leave must commence no later than 13 months from the date of birth or adoption and must expire no later than 25 months from the date of birth or adoption, and.
 - 21.1.2 **Notice Requirement:** Employees exercising their rights under this provision must provide the City at least 45 calendar days notice prior to the anticipated commencement date of the parental leave, unless a shorter notice is approved for good cause.
 - 21.1.3 **Use of Sick Leave:** The employee, at his or her option, may request that all or any portion of sick leave (up to a maximum of two hundred (200) days) or vacation leave that he or she has accumulated be paid in the same manner as it would if he or she had been absent due to illness or on vacation during the leave. In the event both parents are employed by the City, nothing in the Personnel Rules and Regulations shall prohibit both employees from taking simultaneous parental leave.
 - 21.1.4 **Verification Requirement:** The foregoing leave shall be granted upon medical certification of pregnancy or the presentation of legal evidence of adoption.
 - 21.1.5 Life and Health Insurance Coverage Upon Exhaustion of Leave:

During approved parental leave, after all earned leaves are exhausted, (except sick leave) the City agrees to maintain life and health insurance coverage for duration of the approved leave subject to any regular participation requirement of the employee. Thereafter the City agrees to continue coverage for the employee at the employee's expense.

SECTION 22: LEAVE OF ABSENCE WITHOUT PAY

22.1 Department Head Authority

Upon the request of the employee, a department head may grant a leave of absence to an employee within his or her department without pay for a period not to exceed fifteen (15) working days. No leave without pay shall be granted for more than fifteen (15) working days except upon the written request of an employee and approval of the City Manager. Failure on the part of an employee on leave to report promptly at its expiration shall be cause for discharge.

22.2 Union Sponsored Training

A leave of absence without pay shall be granted at the request of an employee and the Union for the purpose of the employee's attending a training course sponsored by the Union. The maximum duration of such leave shall not exceed two (2) consecutive payroll periods in a calendar year.

22.3 Exhaustion of Accrued Time

Leaves of absence without pay shall be granted only after the employee has utilized all accrued vacation and any other time owed to the employee, except sick leave.

SECTION 23: JURY DUTY LEAVE

An employee who is called or required to serve as a trial juror shall be entitled to be absent from work with pay during the period of jury service or while required to be present in court as a result of a call to jury duty. An employee is required to be present at work when not serving as a trial juror or as a member of a jury selection panel. An employee will notify his or her supervisor of any unusual constraints (e.g., time to call in, time to report for jury service) made by the court that affect the employee's ability to simultaneously fulfill his or her jury duty service and employment obligations; and the supervisor will attempt to accommodate the employee based on the operational needs of the department. Absence from work to perform jury duty service shall apply to employees who work swing and graveyard shifts for those days on jury duty. An employee who serves jury duty on his or her day off shall be granted an equivalent number of days off during his or her normal workweek. Employees are required to submit a written proof of jury duty service issued

Page 48 of 183 2020-2022 2022-2025 Memorandum Agreement

City of Berkeley

IBEW Local 1245

by the court in order to receive payment for Jury Duty Leave. The employee will keep any payment received for jury service including mileage reimbursement.

23.1 Court Time

The City will guarantee a minimum of three (3) hours pay for every court appearance required by an employee in the conduct of official City of Berkeley job duties on the employee's scheduled day off and three (3) hours minimum if on a workday but outside scheduled working hours. In addition, employees assigned to court phone standby in the conduct of official City of Berkeley job duties will be compensated by earning compensatory time as follows: Duty day, outside of scheduled working hours, one hour minimum compensatory time and hour for hour thereafter. Day off, two-hour minimum compensatory time and hour for hour thereafter.

ARTICLE 4 - HEALTH AND WELFARE BENEFITS

SECTION 24: HOSPITAL-MEDICAL, DENTAL, AND VISION COVERAGE

24.1 Medical Coverage

The City shall pay the cost of health insurance coverage for employees who are not excluded from such benefit pursuant to Chapter 4.04.120 of the Berkeley Municipal Code (Personnel Ordinance) which is attached hereto as Exhibit C (City of Berkeley Municipal Code Chapter 4.04.120) and made a part hereof, and who have such coverage under any group health insurance plan authorized by the City Council. The present level of benefits under the Kaiser Plan shall be maintained at City expense for the duration of this Memorandum Agreement. Medical benefits are extended to full-time employees, spouse of the employee or domestic partner, and IRS dependent up to age 26.

The City is committed to providing at least one fully paid employer provided health insurance option for employees and eligible dependents. The Union acknowledges the City's policies as stated above.

24.2 Maximum Medical Premium Payments

- 24.2.1 Effective January 1, 2003, the City will transfer employees who chose to maintain Kaiser medical plan coverage from the Kaiser V-5 Plan into the Kaiser S-1 Plan. Effective January 1, 2003, the City will transfer employees who chose to maintain Health Net HMO coverage from the Health Net W-2 Plan to the Health net C9A Plan.
- 24.2.2 For employees hired on or after January 1, 2003, the maximum amount the City shall be required to pay for medical insurance premiums shall be the applicable Kaiser rate (i.e., single party, two-party, or family) regardless of the City sponsored health plan selected by the employee.
- 24.2.3 For employees in a probationary or career benefited status as of January 1, 2003, the City will continue to pay 100% of the health care premium costs (employee and any dependents) for the health plan the employee is enrolled in as of this date as long as the employee remains employed in the bargaining unit. After January 1, 2003, if the employee transfers health coverage to a different health plan, the employee will assume responsibility for paying the difference, if any, between the Kaiser monthly premium rate (i.e., single-party, two-party, or family) and the plan chosen by the employee from that date forward.

- 24.2.4 Effective January 1, 2009, the amount the City contributes each calendar year toward the payment of health insurance premiums, in accordance with Section 24.2 (Maximum Medical Premium Payment), will increase by the lesser of twenty percent (20%) (single, two-party, family) or the amount of the Kaiser HMO premium amount (single, twoparty, family) in effect on that date.
- 24.2.5 Effective January 1, 2014, for those employees who are enrolled on October 31, 2012 in the Health Net Point of Service (POS) health plan (Payroll Benefit Code Description HJ, HK, and HL) the City shall pay medical insurance premiums at the applicable rate for the Health Net HMO plan (i.e., single; 2-party, or family). If the employee chooses to remain enrolled in the Health Net POS Plan on or after January 1, 2014, the employee will assume responsibility for paying the difference between the Health Net POS and the Health Net HMO monthly premium rate (single; 2-party, or family). Provided further that if at any time after January 1, 2014 the employee transfers health coverage from Health Net POS to the Kaiser health plan, the maximum amount the City shall be required to pay for medical insurance premiums shall be the applicable Kaiser rate (i.e., single 2-party; or family) even if the employee subsequently enrolls in a different City sponsored health plan.
- 24.2.6 Meet & Confer: The Parties agree to meet and confer with the City, commencing no sooner than January 1, 2017. The negotiations shall be on developing comparable and less expensive health plan options as a means of reducing or ensuring that the City shall not be required to pay any penalties associated with the Excise Tax. This meet and confer process will be subject to normal rules of collective bargaining, including applicable impasse, strike or lock-out procedures.
- Any changes resulting from a review of employee health insurance or other health related benefits will be subject to meet and confer between the Union and the City, except as otherwise provided in Section 24.5 (Change in Insurance Carrier).

24.3 Dental Coverage

The City shall provide a dental care program for employees who are not excluded from such benefit by Chapter 4.04.120 of the Berkeley Municipal Code (Personnel Ordinance) which is attached hereto as Exhibit C (City of Berkeley Municipal Code Chapter 4.04.120) and made a part hereof. The present level of benefits under the Dental Program shall be maintained at City expense for the duration of this Memorandum. Dental benefits are extended to full-time employees, spouse of the employee or domestic partner, and IRS dependents up to age 26.

- 24.3.1 Effective January 1, 2000, the dental plan is improved from 90% coverage to 100% coverage.
- 24.3.2 Effective January 1, 2003, the annual maximum dental benefit will be increased to \$4000 per year, the lifetime maximum orthodontia benefit will be increased to \$4000, and the annual cleanings will be increased from two (2) to four (4). Effective January 1, 2016, the number of annual cleanings will be decreased from four (4) to three (3).

24.4 Vision Coverage

Effective January 1, 2003, the City shall provide a Vision Care Program for employees covered by this Agreement. The annual maximums for this benefit are as follows:

Benefit	Benefit Frequency
Exam	12-Months
Lenses	12-Months
Frames	24-Months
Contact Lenses**	12-Months
**Note: Benefits for Contact Lenses are in lieu of benefits for lenses and	
frames.	

- The maximum amount the City shall be required to pay for the Vision Care Program shall be the applicable Vision Services Plan (\$25 Plan B) rate (i.e., employee only, employee plus spouse, employee plus one (1) child; employee plus family).
- If during the term of this Agreement the premiums for such Vision Care Program are increased, the amount the City contributes shall increase no more than five percent (5%) above the previous calendar year's contribution amount towards the payment of the monthly premium.
- Recognizing that the "vision exam" portion of the Vision Care Program duplicates existing benefit provided under Medical Coverage Section 24.1 (Medical Coverage) and 24.2 (Maximum Medical Premium Payments), the Union and the City agree to revisit the Vision Care Program during the term of this Agreement with the expectation that it may be possible to identify a plan that provides a "Hardware Only" benefit (Lenses every 12-months; Frames every 24-months). Such "Hardware Only" plan shall be at a lower City and employee cost than the present Vision Services Plan (\$25 Plan B).

24.5 Change in Insurance Carrier

Before the City acts to change an insurance carrier during the term of this Agreement, the City shall give the Union thirty (30) days notice of its intention to change carriers and shall, upon written request, meet with the Union to discuss the reasons for

such change. The final determination of insurance carriers shall be at the sole discretion of the City.

24.6 Domestic Partner Coverage

If an employee chooses to complete and submit an Affidavit of Domestic partnership and sign up for medical and/or dental benefits for his or her domestic partner, the employee shall be subject to federal and state income tax withholding.

24.7 Part-Time Employees and Prorated Benefits

All career and grant-funded, benefited employees working less than full time shall receive prorated rather than full fringe benefits and shall pay, by payroll deduction, a pro rata portion of the health and dental insurance premiums.

24.8 Commencement of Medical, Dental, and Vision Benefits

Medical, dental, and vision benefits shall begin the first day of the calendar month following the date of hire, and end the last day of the month an employee is in pay status.

24.9 Health Insurance In-Lieu Payments

Effective January 1, 2019, for employees who show proof of alternate medical coverage, the City will compensate the employee \$576.00 per month, prorated for less than full-time employees, as provided in Section 24.7 (Part-Time Employees and Prorated Benefits). In order to include in-lieu payments in the regular rate of pay for the correct overtime calculation, the City must provide in-lieu payments on the bi-weekly payroll cycle. This benefit shall be frozen at this amount for the term of this agreement.

24.10 Group Life Insurance

Effective January 1, 2016, the City shall provide group life insurance, by a carrier of the City's choice, in the amount of \$100,000 which shall include a standard accidental death and dismemberment provision of a like amount. In addition, the employee may purchase additional life insurance up to a maximum of \$300,000 at a rate offered by the City's insurance carrier and subject to any medical exam as required by the insurance carrier. Life insurance shall become effective the first day of the calendar month following the date of hire, and shall continue until the last day of the calendar month in a pay status.

24.11 Pre-Tax Status

When employees are required to contribute to the cost for medical, dental, or vision insurance, those contributions will continue to qualify as pre-tax expenses under the provisions of IRS Section 125.

SECTION 25: RETIREE MEDICAL COVERAGE

The City and Union have agreed that the City will make available retiree health insurance coverage under certain terms and conditions described below. The retiree medical benefit described below is the plan tentatively agreed to during multi-union bargaining during the summer of 1998. The terms and conditions of this benefit shall be set forth in a separate document which shall contain a full plan description and shall control the administration of the retiree medical plan.

The City will begin to provide the retiree medical coverage set forth in this Section on July 1, 1998. An employee's entitlement to any and all benefits provided by the City under this retiree medical cover plan are subject to the funding limitations set forth in sub-Section 25.8 (City Funding of Retiree Health Benefit).

25.1 Amendment of Retiree Health Premium Assistance Plan III, effective January 23, 1998, Restated and Amended effective March 22, 2011

Employees who retire on or after June 21, 2015, shall be permitted, at their discretion, to enroll in non-City sponsored health plans. After Council approval of the successor Memorandum of Understanding, the City shall amend the Retiree Health Premium Assistance Plan III as soon as practicable to allow enrollment in non-City sponsored health plans. In the event a retiree elects to enroll in a nonCity sponsored health plan, the City shall make medical insurance premium payments directly to the health insurance provider in an amount equal to what the City would contribute to the City sponsored health plan. Retiree shall be solely responsible for all aspects of the requirements to enroll in a non-City sponsored health plan and maintain eligibility for such a plan; the City's sole obligation is to pay the medical insurance premium contribution required under this Section, as directed by the retiree to a non-City sponsored health plan. The City shall not be responsible for any excess cost differentials associated with the direct payment of premiums to non-City sponsored plans. The City will only make payments through its third party administrator to provide medical insurance premium payments for an individual plan and will not make payments for a group plan. The retiree and/or surviving spouse or domestic partner that enroll in non-City sponsored health plans shall be solely responsible for paying the administrative set up fee, the monthly administrative fee, and/or any other fees established by the third party administrator, and said fees will be deducted directly from the retiree's monthly contribution. No cash payments will be paid directly to the retiree and/or the retiree's spouse/domestic partner. There shall be no cash in lieu payments made under this benefit.

The City and the Union agree that the City will also amend the Retiree Premium Assistance Plan III to allow eligible retirees who retired prior to June 21, 2015 to enroll in a non-City sponsored health plan.

25.2 Eligibility

An employee is eligible for the retiree health insurance coverage set forth in subSection 26.2 (Provisional Appointments to a Higher Classification) below if he or she meets all the following criteria:

- 25.2.1 retires on or after July 1, 1998,
- 25.2.2 is vested with CalPERS,
- 25.2.3 has at least eight (8) years of CalPERS qualifying service with the City,
- 25.2.4 is at least age 55.

25.3 Pre Age 65 Retiree Health Insurance

25.3.1 **Beginning July 1, 1998**: The City shall make available health insurance coverage to the employee and his or her spouse or domestic partner. The City will pay on the employee's behalf no more than \$166.26 per month for an employee electing single party health coverage and no more than \$332.52 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the following chart:

Years of CalPERS Qualifying Service	Percent of City Contribution
8	30%
9	40%
10	50%
11	58%
12	66%
11	58%
12	66%
13	74%
14	82%
15	90%
16	92%
17	94%
18	96%
19	98%
20	100%

The employee will pay the difference between the City's monthly contribution and the actual monthly insurance premium charged by the health plan he or she has elected for retiree medical coverage. If during the term of this Agreement, the premiums for such health insurance are increased, the

amount the City contributes shall increase no more than 4.5% above the previous year's contribution. No increases in the amount the City contributes shall occur before July 1, 1999. Thereafter, any increase in the amount contributed by the City will occur on July 1 each year thereafter.

- 25.3.2 **Effective June 28, 2009**: For employees who retire on or after June 28, 2009, the City will pay on the employee's behalf no more than \$358.19 per month for an employee electing single party health coverage and no more than \$716.38 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.3 **Effective June 27, 2010**: For employees who retire on or after June 27, 2010, the City will pay on the employee's behalf no more than \$424.31 per month for an employee electing single party health coverage and no more than \$848.61 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.4 **Effective June 26, 2011**: For employees who retire on or after June 26, 2011, the City will pay on the employee's behalf no more than \$468.40 per month for an employee electing single party health coverage and no more than \$936.80 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.5 **Effective June 24, 2012**: For employees who retire on or after June 24, 2012, the City will pay on the employee's behalf no more than \$589.48 per month for an employee electing single party health coverage and no more than \$1,178.96 per month for an employee electing two-party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.6 **Effective June 23, 2013**: For employees who retire on or after June 23, 2013, the City will pay on the employee's behalf no more than \$666.00 per month for an employee electing single party health coverage and no more than \$1,332.01 per month for an employee electing two-party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.

- 25.3.7 **Effective June 22, 2014**: For employees who retire on or after June 22, 2014, the City will pay on the employee's behalf no more than \$720.97 per month for an employee electing single party health coverage and no more than \$1441.94 per month for an employee electing two-party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.7.1 Retiree Health Premium Assistance Plan Benefit as of June 30, 2017: As of June 30, 2017, the maximum amount the City will pay the health care service provider, which were increased in accordance with Section 25.3 (Pre Age 65 Retiree Health Insurance), total \$1,517.40 for two party coverage for the retiree and spouse or domestic partner or \$758.70 for single party coverage.

25.4 Retiree Benefits for Employees Age 65 and Over

Once an employee or retiree reaches age 65, he or she is eligible for Medicare. As a result his or her eligibility for the retiree medical benefits set forth in subsection 25.3 (Pre Age 65 Retiree Health Insurance) ceases. On reaching age 65, the City will make available health insurance coverage in addition to Medicare. When an employee or retiree reaches age 65, the City will contribute no more than \$50.00 per month on the employee's behalf for single party health insurance coverage and no more than \$100.00 per month for two party health coverage. If during the term of this Agreement, the premiums for such health insurance are increased, the amount the City contributes shall increase no more than 4.5% above the previous year's contribution.

- The City will take such actions under the provisions of Section 218(g) of the Social Security Act to permit employees who are not currently paying employee portion of the Medicare Tax with a one-time opportunity to choose to be covered by the Medicare Tax. If the employee chooses to be covered by the Medicare Tax the choice cannot be revoked at a later date.
- The City shall include in its next actuarial request, the impact on the City if the City were to amend its Retiree Health Premium Assistance Plan III to provide the following benefit levels:

Single Party Health: \$200.00 Two Party Health: \$400.00

25.4.1 Retiree Health Premium Assistance Plan Benefit as of June 30, 2017: As of June 30, 2017, the maximum amount the City will pay the health care service provider, which were increased in accordance with Section 25.4

(Retiree Benefits for Employees Age 65 and Over), total \$62.59 for two party coverage for the retiree and spouse or domestic partner or \$31.30 for single party coverage.

25.5 Termination by City of Retiree Medical Benefit

Failure of the retiree or surviving spouse to pay their monthly share of the health insurance premium will result in termination of the retiree medical benefit and relieve the City of any further obligation to provide any further benefits under this Section.

25.6 Retiree Medical Benefit for Employees Retiring Between the Ages of 50 and 55

An employee who is at least 50 years of age, but less than 55, and has at least eight years of CalPERS qualifying employment with the City will retain eligibility for the retiree medical benefits provided in sub-Section 25.3 (Pre Age 65 Retiree Health Insurance) when the employee reaches age 55 if the employee is enrolled in a group health plan coverage from the date of his or her termination from City employment until the employee's 55th birthday. If for any reason the employee has a lapse in group health care coverage the employee forfeits his or her eligibility for the retiree health plan benefits upon reaching age 55 and the City has no further obligation to provide any benefits under this Section to the employee and/or his spouse or domestic partner.

25.7 Employees Retiring with a CalPERS Approved Disability Retirement

If an employee retires from the City before age 55 with a CalPERS approved disability retirement, the employee will retain eligibility for the retiree medical benefits provided in sub-Section 25.3 (Pre Age 65 Retiree Health Insurance) when the employee reaches age 55 if the employee is enrolled in a group health plan coverage from the date of his or her termination from City employment until the employee's 55th birthday. If for any reason the employee has a lapse in health care coverage the employee forfeits his or her eligibility for the retiree health plan benefits upon reaching age 55 and the City has no further obligation to provide any benefits under this Section to the employee and/or his spouse or domestic partner.

25.8 City Funding of Retiree Health Benefit

City contributions to the retiree medical benefit will begin on July 1, 1998. Funding of this benefit will be set aside in a trust to be established by the City.

The retiree medical benefit will be funded by a charge of 0.25% of payroll in each year of this Agreement, so that contributions are at 1% of the payroll in the fourth year of the Agreement. The City will fund the benefit at approximately 1% of the payroll for every year thereafter with the intent of achieving a funding level of 70% after 30 years. The funding will be ongoing to maintain a 70% funding level thereafter.

IBEW Local 1245

Effective June 28, 2009, a total charge of twenty six thousand, six hundred and forty (\$26,640) of employee payroll will be charged in the final three years of the Agreement so that contributions are at 1.764% in the final year of the Agreement. The purpose of the 0.764% increase in payroll contributions is to fund Pre Age 65 Retiree Health Insurance.

25.9 Actuarial Study

The Union understands and acknowledges that the City conducted an actuarial study to determine the percentage of payroll it needed to set aside each year and the rate of return of 7% it must achieve to fund the retiree health benefit provided in this Section. The City will conduct an actuarial study by an outside actuary of the retiree medical plan prior to June 30, 2002. After that time, the City will conduct an actuarial study by the outside actuary of the retiree medical plan every two to three years to review the funding status of the program. The outside actuary will be selected by mutual agreement of the parties. The Union and City agree that if the Actuary concludes that the City's funding of this benefit by contribution of 1% of the payroll for all miscellaneous employees is insufficient to fully fund the retiree medical benefits, the City shall not be required to increase its funding for this benefit to more than 1% of the payroll for miscellaneous employees. In the event that there are insufficient funds in the trust to cover all retirees' monthly health premiums, the City and the Union agree to meet and confer regarding the City's distribution of its 1% contribution.

ARTICLE 5 – TERMS AND CONDITIONS OF EMPLOYMENT

SECTION 26: PROBATIONARY PERIOD

26.1 Duration and Effect of Military Leave on Probationary Period

Original and promotional appointments from employment lists shall be tentative and subject to a probationary period of: one (1) year (full time equivalent exclusive of all leave and light duty completed within 18 months) upon original appointment to the unit; employees promoted within the unit are subject to a probationary period of six (6) months (half time equivalent exclusive of all leave and light duty completed within one year); in addition, employees in trainee programs with specified training, evaluation or probation periods shall be governed by the applicable provisions of such program. Probationary employees who are granted military leaves of absence shall complete the balance of their probationary period within a period of six (6) months following their return to City service. No provision of this Section shall be interpreted to preclude the City from establishing new classifications, which may require a probationary period of more than six (6) months.

26.2 Provisional Appointments to a Higher Classification

If, before completing the required probationary period, an employee is provisionally appointed to a higher class in the same or a related series of classes, the time served in such higher class shall be counted toward completion of the probationary period in the lower class.

26.3 Report Requirement

If the service of the probationary employee has been satisfactory to the department head, the department head shall file with the Director of Human Resources a statement in writing to such effect and stating that the retention of such probationer in the service is desired. If such service has been unsatisfactory, the department head shall file with the Director of Human Resources such a statement in writing with the recommendation to the City Manager that the employee be rejected.

26.4 Rejection during Probationary Period

During the probationary period, an employee may be rejected at any time without right of appeal or hearing in any manner. An employee rejected from a position to which he has been promoted shall be reinstated to the position from which he was promoted unless charges are filed and he is discharged as provided in Section 31 (Discharge). Career City employees who are rejected during their probationary period in a promotional position or fail to complete the training requirements of a training program (academics or OJT) shall revert to their prior classification.

IBEW Local 1245

26.5 Rejections Subject to Disciplinary Appeal

A promoted employee may not grieve his or her rejection from probationary period/training. However, an employee disciplined for cause while on promotional probationary period shall have access to the disciplinary appeal procedure.

26.6 Probationary or Temporary Employee Assignments

No probationary or temporary employee will be assigned as the only ground worker with an Electrician until such employee has completed one (1) month of familiarization with aerial truck equipment.

SECTION 27: TRANSFER

A transfer may be made at any time by the City Manager. Transfer shall not be used to effect a promotion, demotion, advancement, or reduction, each of which may be accomplished only as provided elsewhere in this Memorandum Agreement. No person shall be transferred to a position for which he or she does not possess the minimum qualifications. An employee with permanent status who is transferred from one class to another class shall assume permanent status in the class to which the employee is transferred.

SECTION 28: PROMOTION

Insofar as practicable and consistent with the best interests of the service, all vacancies in the competitive service shall be filled by promotion from within the competitive service after a promotional examination has been given and a promotional list established.

If, in the opinion of the City Manager, the best interests of the service can be served by an open, competitive examination instead of a closed, promotional examination, and if there is not already a promotional list for the higher position, which list has not been abolished and from which the vacancy could be filled, then the City Manager may instruct the Director of Human Resources to call for applications for the vacancy and arrange for an open, competitive examination and for the preparation and publication of an eligibility list.

Interview of Employees: A City employee who is on a closed promotional or open competitive list shall have the option to interview for the vacancy. A City employee who is unsuccessful and who so requests shall be advised of steps he or she may take to increase his or her competitive standing for future promotional opportunities.

IBEW Local 1245

Employees who have qualified for promotional lists shall be considered for promotion based on the following factors: previous work performance, previous training and experience, merit, ability, and seniority.

SECTION 29: DEMOTION

The City Manager may demote an employee who so requests it, or whose ability to perform his or her required duties falls below standard, or for disciplinary purposes. No employee shall be demoted to a class for which he does not possess the minimum qualifications as determined by the Director of Human Resources.

Notice of the demotion shall be given the employee not later than four (4) weeks prior to the effective date of demotion and a copy of said notice filed with the Director of Human Resources. Any employee who has been demoted shall be entitled to receive a written statement of the reasons for such action.

An employee with permanent status who is demoted shall assume permanent status in the class to which he or she is demoted.

Upon request of the employee, demotion may be made to a vacant position as a substitution for layoff. In such cases, the employee shall be restored to his or her former position without further examination whenever such position is again to be filled.

SECTION 30: SUSPENSION

The City Manager may suspend an employee from his position at any time for the good of the service, for a disciplinary purpose, or for other just cause. Suspension without pay shall not exceed twenty (20) working days, nor shall any employee be penalized by suspension for more than twenty (20) days in any fiscal year. Any employee who has been suspended shall be entitled to receive a written statement of the reasons for such action.

For the good of the service, a department head may suspend an employee for not more than three (3) working days for any one offense. Such suspension shall be reported immediately to the City Manager.

SECTION 31: DISCHARGE

An employee may be discharged at any time by the City Manager, but if the probationary period has been completed then such discharge must be for just cause. Any employee

IBEW Local 1245

who has been discharged shall be entitled to receive a written statement of the reasons for such action.

SECTION 32: RESIGNATION

An employee wishing to leave the competitive service in good standing shall file with the department head, at least two (2) weeks before leaving the service, a written resignation stating the effective date and reasons for leaving. The resignation shall be forwarded to the Director of Human Resources with a statement by the department head as to the resigned employee's service performance and other pertinent information concerning the cause for resignation. Failure of the employee to give the notice required shall be entered on the service record of the employee and may be cause for denying future employment by the City. The resignation of an employee who fails to give notice shall be reported by the department head immediately.

SECTION 33: REINSTATEMENT

A permanent or probationary employee who has resigned with a good record may be reinstated within two (2) years to his or her former position, if vacant, or to a vacant position in the same or comparable class without further competitive examination. This Section shall not be interpreted as a guarantee of reinstatement to an employee who has resigned with a good record and requests reinstatement within two (2) years.

SECTION 34: NOTICE OF TERMINATION

Department head shall notify the Director of Human Resources of all terminations of employment before or within two (2) days after the termination has taken place. Such notice shall be on forms provided by the Director of Human Resources and shall indicate the effective date and reasons for termination and a statement certifying whether or not the employee's services have been satisfactory.

ARTICLE 6 - GRIEVANCE AND DISCIPLINARY APPEAL PROCEDURE

SECTION 35: GRIEVANCES

35.1 Definition

A grievance is any dispute which involves the interpretation or application of those rules, regulations, and resolutions which have been or may hereafter be adopted by the

City Council to govern personnel practices and working conditions, including such rules, regulations, and resolutions as may be adopted by the City Council to effect memorandum agreements which result from the meeting and conferring process.

- A grievant may be any member of the bargaining unit covered by the terms of this Agreement, or the grievant may designate the Union to act on his or her behalf, or the Union itself may file a grievance on behalf of a member or group of members. Grievances must be filed in a timely manner. Grievances that are filed by the Union on behalf of a member or group of members, or when a grievant designates the Union to act on his or her behalf, are subject to settlement at any step of the grievance procedure at the Union's sole discretion.
- A member of the bargaining unit that files a grievance where the Union is not representing or acting on the grievant's behalf, may only utilize this grievance procedure through step 3. The Union retains the sole discretion to refer which, if any, grievances to move to arbitration.
- The parties agree that disclosure to the other party of all relevant information available to them is critical to the successful resolution of grievances at the lowest possible step of the grievance procedure. The parties therefore agree to disclose to each other the relevant information in their possession related to the grievance. Further, the parties agree to comply with reasonable grievance related information requests from the other party in a timely manner.

35.2 Procedure

Grievances shall be processed in the following manner:

- 35.2.1 **Informal Step:** An employee who believes that he or she has a grievance (and / or the employee's steward or Union representative) may discuss the complaint with the Deputy Director of Public Works, or with such subordinate management official as the Deputy Director may designate. If the issue is not resolved informally, the formal procedures hereinafter specified may be invoked.
- 35.2.2 **First Formal Step:** Any informal grievance which has not been resolved by the Deputy Director of Public Works or his or her designee, must be filed in writing to the Deputy Director of Public Works within thirty (30) days from the date the informal discussion was initiated. The written grievance must state specific issues involved, the decision rendered at the informal step, and the remedy sought. The Deputy Director or his or her designee shall have five (5) days to respond in writing to the formal, written grievance.

- 35.2.3 **Second Formal Step:** Any formal grievance which has not been resolved by the Deputy Director of Public Works or his or her designee, may be referred to the Director of Public Works or his or her designee within thirty (30) days from the date of the Deputy Director's formal written response, or if more than five (5) days have elapsed since the grievance was presented at the Deputy Director's level.
 - 35.2.3.1 **Referral to Department Director:** The grievance may be referred to the Director's level and shall include a copy of the original grievance, the decision rendered at the Deputy Director's level, and a clear, concise statement of the reasons for the referral.
 - Director of Public Works may designate a personnel representative to investigate the merits of the complaint, to meet with the complaining employee (and/or the employee's steward or Union representative); if the grievant is not the Union itself, to meet with the officials of the Union; and to settle such grievance or to make recommendations thereon to the City Manager in his or her capacity as an employee relations officer. The Director of Public Works shall issue a formal written decision within five (5) days of receipt of the grievance or if a meeting is held with the employee and his or her Union, the meeting must be held within ten (10) days of receiving the referral. The Director of Public Works or his or her designee shall issue a written decision within five (5) days from his or her meeting with the complaining employee and/or Union representative. If the issue is not resolved the grievance may be referred to the Third Formal Step within thirty (30) days of receipt of the response from the Director of Public Works.
- 35.2.4 **Third Formal Step:** Any grievance which has not been resolved by the procedures hereinabove set forth may be referred to the City Manager by the grievant, the Union, or the Director of Public Works. Any such referral shall be in writing including a copy of the original grievance, the decision rendered at the Director's level, and a clear, concise statement of the reasons for the referral. The City Manager shall designate a Human Resources representative to investigate the merits of the complaint, and upon request from the employee or the union, that representative to meet with the complaining employee (and/or the employee's steward or Union representative), if the grievant is not the Union itself, to meet also with the officials of the Union, and to settle such

grievance or to make recommendations thereon to the City Manager in his or her capacity as an employee relations officer.

The City Manager shall issue a decision within ten (10) days of receipt of the grievance referral or within ten (10) days of the meeting, or if a meeting is held with the employee and/or the Union, the meeting must be held within ten (10) days of the referral. Grievances filed by a bargaining member where the Union is not representing or acting on the grievant's behalf, shall end at this step and no further right of appeal exists. The City Manager's decision shall be final.

Union controlled grievances: If the issue is not resolved, the grievance may be referred to Arbitration by the Union pursuant to Section 35.2.5 (Arbitration: The Union retains the sole discretion to move a grievance to arbitration) below within thirty (30) days of receipt of the City Manager's response. Provided further that the Union shall forward to the City the Union's portion of the State Mediation and Conciliation Services (SMCS) fee within sixty (60) days of receipt of the City Manager's response. Failure by the Union to meet this sixty (60) days deadline for both referral to Arbitration and payment of the SMCS fee shall be deemed as a full and complete waiver by the Union to appeal the City Manager decision to Arbitration and the City Manager decision shall be final and binding on all parties.

35.2.5 **Arbitration**: **The Union retains the sole discretion to move a grievance to arbitration**. Any grievance which has not been resolved by the procedures hereinabove set forth may be referred by the Union, or the City to an impartial arbitrator who shall be designated by mutual agreement between the Union and the City, using the State Mediation and Conciliation Services (SMCS) to provide a list of five (5) arbitrators. The City and the Union will alternately strike a name until one remains. The remaining name will be the arbitrator. The fees and expenses of the arbitrator, the State Mediation and Conciliation Services (SMCS) and of the court reporter shall be shared equally by the Union and the City. Each party, however, shall bear the cost of its own presentation, including preparation and post-hearing briefs, if any. Arbitrator decisions on matters properly before them shall be final and binding on both parties.

The arbitrator may hear testimony, receive written briefs, interview witnesses, and conduct any investigation she or he deems appropriate, and shall render a final and binding decision to the parties which will end the formal grievance process.

- 35.2.5.1 **Arbitrator Jurisdiction:** No arbitrator shall entertain, hear, decide, or make recommendations on any dispute involving a position over which a formally recognized employee organization has jurisdiction unless such dispute falls within the definition of a grievance as hereinabove set forth in Section 35 (Grievances).
- 35.2.5.2 **No Modifications:** Proposals to add to or change this Memorandum Agreement or written agreements or addenda supplementary hereto shall not be arbitrable, and no proposal to modify, amend, or terminate this Memorandum Agreement, nor any matter arising out of or in connection with such proposal, may be referred for arbitration under this Section; and no arbitrator shall have the power to amend or modify or recommend amendment or modification of this Memorandum Agreement or any written agreements or addenda supplementary hereto or to establish or recommend establishment of any new terms and conditions of employment.
- 35.2.5.3 **Timeliness of Award:** No arbitrator will be selected hereunder who does not agree to render an award not later than thirty (30) calendar days after the close of the hearing. By mutual agreement between the City and the Union, the arbitrator may render an award immediately upon the conclusion of the presentation of evidence.

35.3 EEO Complaints

Any grievance which in any way affects the implementation of the City's Equal Employment Opportunity Policy shall not be subject to arbitration. The decision as to whether or not implementation of the Equal Employment Opportunity Policy is in any way involved shall be made by the City Manager in his or her sole discretion. If, in his or her judgment, any grievance which involves the Equal Employment Opportunity Policy, the Equal Employment Opportunity and Diversity Officer shall notify the Union to that effect in writing within seven (7) days of the date upon which the grievance is received in the Human Resources Department and, in such notification, refer to that Section of the Equal Employment Opportunity Complaint Investigation & Resolution Procedure which is involved; provided, however, that such notice may come at any time prior to arbitration if additional factors come to the attention of the Equal Employment Opportunity and Diversity Officer on the basis of which he or she considers it appropriate to change his or her original determination.

35.4 Compensation Complaints

All complaints involving or concerning the payment of compensation shall be initially filed in writing with the Director of Human Resources. In such cases, no adjustment shall be retroactive for more than thirty (30) days from the date upon which the complaint was filed or thirty (30) days from the date when an employee may reasonably be expected to have learned of said claimed violation. Only grievants which allege that employees are not being compensated in accordance with the rules, regulations, and resolutions of the City Council or in accordance with the understanding contained in any Memorandum Agreement which has resulted from the meeting and conferring process shall be considered as grievances. Any other matters of compensation are to be resolved in the meeting and conferring process, shall be deemed withdrawn until the meeting and conferring process is next open for such discussion.

35.4.1 **Response Time Limits:** The Director of Human Resources or his or her designee shall issue a decision in writing within twenty (20) days from the time he or she received the compensation grievance in writing. The grievant or the Union may refer the decision to the Third Formal Step (Section 35.2.4 - Third Formal Step) of this Memorandum Agreement) within twenty (20) days of receipt of the written decision.

35.5 Changes or Interpretations

No changes in this Memorandum Agreement or interpretation thereof (except interpretations resulting from arbitration proceedings hereunder) will be recognized unless agreed to by the City Manager and the Union.

35.6 Extension of Timelines

Timelines noted in this Section may be extended for cause upon mutual agreement between the City and the Union.

35.7 Days

All references in this Section 35 (Grievances) to "days" shall mean normal working days except as otherwise noted.

35.8 Grievances Challenging Disciplinary Action

- 35.8.1 Any grievance involving any disciplinary action up to as three (3) day suspension will be filed at the Director level.
- 35.8.2 All grievances involving demotion, discharge, or suspension of greater than three (3) days will be filed at the City Manager's level of the grievance

Page 68 of 183 2020-2022 2022-2025 Memorandum Agreement

City of Berkeley

IBEW Local 1245

- procedure. If the issue is not resolved within fifteen (15) days of referral to this step, the procedures hereinafter specified may be invoked.
- 35.8.3 No grievance involving the suspension or discharge of an employee will be entertained unless it is filed in writing with the City Manager within five (5) working days of the time at which the affected employee was notified of such action in writing. If the City Manager, in pursuance of the procedures outlined in Section 35.2.4 (Third Formal Step) above, resolves a grievance which involves suspension or discharge, he or she may agree to payment for lost time or to reinstatement with or without payment for lost time.
- 35.8.4 Arbitrator decisions on matters properly before them which pertain to the suspension or discharge of an employee shall be final and binding on both parties, to the extent permitted by the Charter of the City.

ARTICLE 7 – GENERAL PROVISIONS

SECTION 36: GENERAL PROVISIONS

36.1 Personal Conduct

- 36.1.1 No employee shall accept appointment to the deputyship or assistantship of any county or state office or position, or otherwise incur an obligation of civil public service outside his or her regular municipal employment without first obtaining the recommendation of the head of his or her department and of the City Manager.
- 36.1.2 No employee shall be disciplined for off-the-job activities which do not affect his or her job performance.
- 36.1.3 Employees shall so arrange their personal financial affairs so that the demands of creditors and collection agencies shall not impose a recurring burden upon the offices of the City Manager, the department head, or the Director of Human Resources for the purpose of making collections.
- 36.1.4 Full-time City employees may not carry on concurrently with their public service any private business or undertaking, attention to which affects the time or quality of their work or which casts discredit upon or creates embarrassment for the City government.
- 36.1.5 No official or employee who wears a badge or other official insignia as evidence of his or her authority and identity shall permit such badge or insignia to be used or worn by any other person of the same or another department or otherwise to leave his possession without approval by the head of this department. The department head shall not grant such approval except as to persons regularly and formally appointed by the City Manager to the position designated by the badge or insignia.

36.2 Rain Gear

The City will provide rain gear for employees in Representation Unit C assigned to work in inclement weather.

36.3 Tools

Each employee shall continue to be responsible for providing tools of the trade or other equipment, but shall receive a flat annual tool replacement allowance each July 1.

New employees hired before or after July 1 will be entitled to a prorated tool replacement allowance payable the following July 1 after employment.

36.3.1 **Allowance:** Effective July 1, 2015, the tool replacement is five hundred dollars (\$500). Tool allowances shall be paid at the end of the fiscal year in which they are earned.

36.4 Shoes

Effective June 18, 2017, all classifications covered by this MA shall be eligible for reimbursement of up to two hundred dollars (\$200.00) per fiscal year for the cost of safety shoes. Employees receiving this reimbursement must wear safety shoes on duty.

36.4.1 **Reimbursement**: Employees shall be required to submit a reimbursement request, along with receipts verifying the purchase of safety shoes within sixty (60) days of purchase. Reimbursements will be paid within forty-five (45) days of complete reimbursement requests.

36.5 Training

- 36.5.1 **Responsibility:** Responsibility for developing training programs for employees shall be assumed by the City Manager. Such training programs may include lecture courses, demonstrations, assignment of reading matter, or such other devices as may be available for the purpose of improving the efficiency and broadening the knowledge of municipal employees in the performance of their duties. The City shall endeavor to provide training for all new technology purchased by the City and maintained by members of this Agreement.
- 36.5.2 Advancement/Promotional Consideration for Special Training Courses: Participation in and successful completion of special training courses may be considered in making advancements and promotions. Evidence of such activity shall be filed with the Director of Human Resources by the department head.
- 36.5.3 **Mileage and Tuition Reimbursement:** The City shall reimburse all employees for mileage and tuition expenses related for attendance at job-related courses. Provided that the employee must have prior authorization from his or her department head and the course has been approved by the City.
- 36.5.4 **Education Leave:** The City shall allow up to forty (40) hours off with pay per year to employees:

- 36.5.4.1 Who are required to obtain a license (excluding Class C driver's licenses) issued by the State of California and, in order to do so, must take courses which were not offered as a part of their basic curriculum or.
 - 36.5.4.2 Who are required to obtain continuing education as a condition of license renewal; provided the license is required by the City for the employment or continued employment of the employee and that no more hours than are required by the State shall be granted. Employees seeking time off to take courses for an initial license must provide verification that the course was not offered as a part of their basic curriculum.
 - Employees who take classes during non-scheduled work hours in order to retain a job required license or to meet continuing education requirements shall be allowed time off from work on an hour-for-hour basis without loss of compensation or other benefits.
- 36.5.4.3 To obtain education and training related to job skills to enhance performance, or to qualify for promotion, at the discretion of the department head.
- 36.5.5 **Management/Union Meetings:** The City and Union agree that it is beneficial to both parties to discuss and develop training for all classifications. The Director of Public Works and the Union shall meet on a quarterly basis to discuss and review progress made in addressing initial employee orientation, new equipment training, and vendor training.
- 36.5.6 **Commercial Driver's License**: The City shall provide the necessary training for employees to obtain a California Class B driver's license when such license is a requirement of their job classification and a condition of employment. Employees who fail to obtain the required California Class B driver's license after training will be subject to rejection from the probationary period pursuant to Section 26.4 (Rejection during Probationary Period) of this Memorandum Agreement.

36.6 Use of Automobiles

The City Manager shall govern the use of City-owned automotive equipment and privately-owned automotive equipment by such rules and regulations as he or she may establish. Compensation shall be given in the form of a cash allowance for the use of private-owned automobiles on City business if such use has been

authorized in advance by the City Manager. The cash allowance will be equal to the amount established by the Internal Revenue Service, and will change as necessary to comply with IRS Standard Mileage Rate.

36.7 Safety

- 36.7.1 **Safety Committee:** The City and the Union will make every effort to maintain excellent health and safety standards. No employee shall be required to perform work with unsafe equipment or in situations which are injurious to his or her health or safety. To further these purposes, the City shall maintain an ongoing safety program which shall include committees comprised of representatives from the Union and appropriate supervisory personnel.
- 36.7.2 **Rainy Weather Assignments:** The City shall endeavor to avoid assigning non-emergency, routine work in rainy weather, which would constitute a safety hazard.

36.8 YMCA Group Membership

- The City shall offer employees a low or no-cost group membership in the Berkeley Central YMCA. As of July 1, 1999, the City will pay 75% of the membership fee. If the monthly fee is increased to more than \$60, the employee share will be capped at \$30 per month; the City will pay the balance. The amount the City contributes toward the employee's monthly membership fee is subject to federal and state income tax withholding.
- Use of the YMCA membership by a City of Berkeley employee, as provided for in this Agreement, is non-compensable, is not a part of the employee's work-related duties, is not required for employment and is not condoned as part of a physical fitness program, or required to remain top physical conditioning for the employee's job performance.
- The City of Berkeley or its Claims Administrator may not be liable for any injury that arises out of a City of Berkeley employee's participation in and use of a YMCA membership.

36.9 Annual Performance Evaluation

The City may implement a program of annual performance evaluation. Such evaluations shall be conducted by the employee's immediate supervisor and reviewed by additional levels of supervision. Each employee may make written comments on the evaluation, which shall be made a part of the employee's personnel records.

ARTICLE 8 - PUBLIC EMPLOYEES RETIREMENT SYSTEM

SECTION 37: PAYMENT OF EMPLOYEES' PERS CONTRIBUTIONS

37.1 Miscellaneous Designation

The City shall continue participation under the Miscellaneous Employees Plan of the California Public Employees' Retirement System, (CalPERS).

37.2 "Classic Employees" Definition

Classic Employees are defined as current employees and future employees who do not qualify as "New Members" under the California Public Employees' Pension Reform Act of 2013 (PEPRA).

37.3 "New Member" Definition

"New Members are as defined in the Public Employees' Pension Reform Act of 2013 (PEPRA), Government Code Section 7522.04(f).

37.4 CalPERS Retirement Formula for "New Members" as Defined Under the Public Employees' Pension Reform Act of 2013 (PEPRA)

"New Members" as defined by PEPRA who are hired by the City on or after January 1, 2013 shall be entitled to the retirement formula set forth in PEPRA.

37.5 CalPERS Retirement Formula and Employer Paid Member Contribution for Classic Employees (i.e. current employees and employees who do not qualify as "New Members" under PEPRA)

Current employees and other employees who do not qualify as "New Members" under PEPRA shall continue to be entitled to the 2.7% at age 55 retirement formula benefit, and the City shall continue the contribution of eight percent (8%) to CalPERS on behalf of the employee.

37.6 CalPERS New Members Normal Share of Cost

New Members as defined by PEPRA who are hired on or after January 1, 2013 shall pay 50% of the normal share of cost required by PEPRA. New Members shall receive any other additional optional CalPERS benefits that the City provides to Classic Employees as allowed by PEPRA.

37.7 Reporting of Contributions

Effective July 3, 1994, contributions made pursuant to Section 37.5 above have been reported to CalPERS as "special compensation" as provided in Government Code Section 20636(c)(4) pursuant to Section 20691. Said contributions shall not apply in the case of temporary or provisional employees.

- The member contributions made by the City shall not be considered as a part of an employee's salary for the purpose of computing straight time earnings, compensation for overtime worked, or education incentive pay; nor shall such contribution be taken into account in determining the level of any other benefit which is a function of or percentage of salary. Pursuant to California Government Code Section 20691, the City shall not increase, reduce or eliminate payments of the normal contributions on behalf of the employee without engaging in the meet and confer process with the Union.
- The City will not treat these contributions as compensation subject to income tax withholding unless the Internal Revenue Service or Franchise Tax Board indicates that such contributions are taxable income subject to withholding. Each employee shall be solely and personally responsible for any federal, state or local tax liability of the employee that may arise out of the implementation of this Section or any penalty that may be imposed therefore.

37.8 City Contracts with CalPERS

The City's contract with CalPERS includes the following optional benefits:

- 37.8.1 Classic Employees One Year of Final Compensation: Classic Employees as defined in Section 37.2 ("Classic Employees" Definition) above and as provided in Government Code Section 20042 (July 9, 1978).
- 37.8.2 **New Members Three Years Final Compensation**: Provided further that New Members as defined by PEPRA hired on or after January 1, 2013 shall be eligible to receive retirement allowance based on three (3) highest consecutive years of compensation under the plan as provided under the California Public Employees' Pension Reform Act of 2013, or as subsequently amended.
- 37.8.3 Post Retirement Survivor Allowance as provided in Sections 21624, 21626 and 21628 (December 16, 1973).
- 37.8.4 Post Retirement Survivor Allowance to Continue after Remarriage as provided in Section 21635 (July 18, 1986).
- 37.8.5 Credit for Unused Sick Leave as provided in Section 20965 (June 26, 1988).
- 37.8.6 1959 Survivor Benefits to Surviving Spouse at Age 60 as provided in Section 21580 (December 16, 1973).

- 37.8.7 Third Level of 1959 Survivor Benefits as provided in Section 21573 (November 28, 1996).
- 37.8.8 2% @ 55 for Local Miscellaneous Members as provided in Section 21354 (June 30, 1992).
- 37.8.9 Military Service Credit as Public Service as provided in Section 21024 (April 9, 1999).
- 37.8.10 Public Service Credit for Peace Corps or AmeriCorps/VISTA Service as Provided in Section 21023.5 (April 14, 2000).

37.9 Index Level 1959 Survivor Benefit

The City intends to amend its contract with CalPERS to provide the Indexed Level of the 1959 Survivor Benefit as provided in Section 21574.5 when administratively feasible.

37.10 Classic Members' Pension Contribution through 20516 a Contract Amendment

37.10.1 **July 3, 2016:** Effective July 3, 2016, employees will contribute one and one-quarter percent (1.25%) toward the City's CalPERS employer share of pension through a 20516 CalPERS amendment that allows such contributions via automatic payroll deduction on a pre-tax basis, in exchange for the City granting the salary increase set forth in Section 11.1.5 (July 3, 2016 - Salary/PERS Exchange) of this MA. Such employee deductions by the City shall be used towards the City's CalPERS required contributions.

37.11 New Members' Pension Contributions through 20516 a Contract Amendment

- 37.11.1 **July 3, 2016**: Effective July 3, 2016, in addition to the contributions in Section 37.4 (CalPERS Retirement Formula for "New Members" as Defined Under the Public Employees' Pension Reform Act of 2013 (PEPRA)), New Members will contribute one and one-quarter percent (1.25%) towards the City's CalPERS employer share of pension through a 20516 CalPERS amendment that allows such contributions as pre-tax via automatic payroll deduction, in exchange for the City granting the salary increase set forth in Section 11.1.5 (July 3, 2016 Salary/PERS Exchange) of this MA. Such employee deductions by the City shall be used towards the City's CalPERS required contributions.
- 37.11.2 The parties recognize that the CalPERS 20516 amendment that allows employee contributions toward the employer rate is in addition to the

City of Berkeley

IBEW Local 1245

required 50% of the normal cost of "New Members" benefits and made in consideration of additional salary increases in Section 11.1.5 (July 3, 2016 - Salary/PERS Exchange) above (a total of 1% salary increases in exchange for employees paying a total of 1.25% towards CalPERS pension cost).

As soon as administratively possible, following council approval of this agreement, the additional contribution of 1.25%, as indicated in Section 37.11.1, shall be eliminated through a contract amendment with CalPERS. The start of the first full pay period following completion of the contract amendment shall be the effective date of the reduction.

SECTION 38: SUPPLEMENTAL RETIREMENT/DISABILITY INSURANCE PLAN

Effective January 1, 1983, the majority of miscellaneous employees under the City's contract with the State of California Public Employees Retirement System who were covered by the integrated Social Security Program voted to withdraw from participation in the Federal Social Security Program. In lieu of Social Security payments, the City has agreed to pay an amount equal to that percent of individual pay (6.7% payable on the first \$32,400 of salary paid in the calendar year) which had been paid by the City to Social Security as of December 31, 1982 to a Supplemental Retirement and Income and Long Term Disability Insurance Plan for those employees previously covered under the integrated PERS/Social Security Plan. Provisions of this plan are described in Berkeley Municipal Code Chapter 04.36.101 (Supplemental Retirement and Income Plan I) as amended and Berkeley Municipal Code Chapter 04.38.101 (Supplemental Retirement and Income Plan II) as amended. All employees hired after July 22, 1988 are in SRIP II. All employees hired prior to July 22, 1988 are in SRIP I unless they chose to enroll in SRIP II prior to December 17, 1988.

ARTICLE 9 - LAYOFF PROCEDURE

SECTION 39: LAYOFF

The layoff policy for the City of Berkeley is intended to provide the maximum employment protection of the City staff should a layoff become necessary. The policy also aims to minimize the impact such a layoff might have on the City's affirmative action accomplishments.

39.1 Announcement of Layoff

- 39.1.1 **Notification:** The City Council, City Manager, and department head shall make every reasonable effort to manage and budget the City's resources effectively and to plan for the delivery of City services in a manner which will avoid the necessity of laying off career City employees. If a reduction in the work force for more than thirty (30) calendar days is necessitated by, but not limited to, the following: a material change in duties and organization, adverse working conditions, return of employee from leave of absence, or shortage of work or funds, the City Manager shall notify the Director of Human Resources of the intended action and the reason for the layoff.
- 39.1.2 Freezing of Vacancies: Immediately following a decision which may involve the potential layoff of career City employees, the City Manager shall freeze all current City vacancies in the competitive service in similar and related classifications to those likely to be targeted for layoff, as well as all related full-time, benefited, temporary positions which are expected to last six (6) months or more, and notify all department heads that such current and anticipated vacancies will be frozen until further notice in order to implement the provisions of Section 39.6 (Flexible Placement Program).

39.2 Seniority Service Date

39.2.1 All service in the employ of the City shall be counted toward the establishment of an employee's seniority service date, including, for example, permanent, probationary, provisional, temporary (full-time and intermittent), seasonal, and exempt employment, as well as leaves of absence for obligatory military service while an employee of the City. Less than full-time service will be consolidated in equivalences of full-time service for the purpose of establishing the seniority service date. Time off as a result of formal disciplinary action will be subtracted from the seniority service date.

39.2.2 The auditor's office will maintain up-to-date and current seniority dates for all City employees holding probationary and permanent appointments.

39.3 Establishment of Seniority Lists

- 39.3.1 Whenever a layoff of one or more career employees becomes necessary, as defined above, such layoffs shall be made according to City-wide classification seniority lists. Upon receiving notification that the City Manager must proceed with a possible reduction in the work force, and following receipt of information concerning the specific positions, programs, and departments involved, the Human Resources Department will immediately establish separate probationary and permanent seniority lists for each classification targeted for layoff.
- 39.3.2 The names of all City employees holding permanent and probationary appointments in a given classification will be listed on the appropriate list in descending order by seniority service date. Employees on both lists shall be laid off on the basis of their seniority service date, i.e., employees with the least amount of total service shall be laid off first. All emergency, temporary, and provisional employees working in classifications similar to those identified for layoff must be terminated prior to the layoff of probationary or permanent employees. Employees on the probationary seniority list for a specific classification will be laid off prior to employees on the permanent seniority list for that classification.
- 39.3.3 Probationary or permanent employees temporarily acting out of classification and holding a provisional appointment in another classification will only be listed on a seniority list of the classification in which they hold permanent or probationary status targeted for layoff.
- 39.3.4 If two (2) or more employees on a seniority list have an identical seniority service date, the tie shall be broken in the following order:
 - 39.3.4.1 Time in classification the employee having least time in the classification shall be released first;
 - 39.3.4.2 By lot.

39.4 Employee Retreat Rights

- 39.4.1 Before an employee with permanent or probationary status may be released from employment with the City of Berkeley, the Human Resources Department must consider the employee's right to retreat to lower level classifications through which he or she was originally promoted or any subsequently created intermediate level career classification which provides normal progression through the classification series. Retreat rights shall also extend to employees who have not previously been promoted through a classification but for whom the classification is a natural progression or beginning in the classification series.
- 39.4.2 In the process of retreating, the same rules concerning the length of service, classification seniority lists, etc., apply as in the first stage of the layoff process. In order to retreat, the targeted employee must be higher on the seniority list for the classification into which he or she is retreating than at least one of the incumbents on the probationary or permanent seniority list for that classification.
- 39.4.3 If an employee is qualified for retreat into more than one classification with comparable salary ranges, or if a vacancy exists in a classification to which an employee is entitled to retreat, the options shall be discussed with the employee and due consideration shall be given to the employee's preferences. However, it is the prerogative of the City Manager to determine the final placement offer to the employee.
- 39.4.4 The retreating employee has a right to be retained in the highest salary range possible which is equal to or less than his or her present salary range. An employee involved in layoff does not have a right of mandatory placement to positions with a higher salary range, i.e., promotion.

39.5 Employee Notification

- 39.5.1 Emergency, temporary, intermittent, seasonal, etc., employees shall be notified individually, in writing, of pending layoff as soon as possible, but no definite time period is required. However, at least two (2) weeks notification is desirable if possible.
- 39.5.2 Provisional employees shall be notified individually, in writing, of pending layoff as soon as possible, with no less than fifteen (15) calendar days notification if targeted for release or reassignment.
- 39.5.3 Permanent, probationary, and career-exempt employees should be notified individually in writing of pending layoffs as soon as possible, with no less

than thirty (30) calendar days notification if targeted for release or reassignment.

- If an employee fails to accept a bona fide offer of reassignment within ten (10) calendar days after the offer has been made, he or she forfeits further right to employment retention. Acceptance of a reassignment does not remove the right of appeal under Section 39.9 (Appeal Procedure).
- 35.5.4 An employee who is transferred in lieu of layoff when his or her position has been eliminated shall have automatic return rights to the previous position if it is restored within one (1) year of the date of the transfer. If an employee with a full time position is offered a reduction in hours in that position or in a lower classification, the employee may elect to be targeted for layoff for purpose of consideration under Section 39.6 (Flexible Placement Program.) If there is no flexible placement available for the employee, the employee may accept the reduction in hours, in lieu of layoff.

39.6 Flexible Placement Program

- 39.6.1 In order to minimize the negative impact of layoff, the City Manager will, as previously stated in Section 39.1.2 (Freezing of Vacancies), impose a citywide freeze on all appropriate vacancies as soon as it has been determined that a layoff of career City employees may be necessary.
- 39.6.2 Following the release of all emergency, temporary, and provisional employees in classes similar to those targeted for layoff, and as soon as employees targeted for layoff have been identified and the provisions under Sections 39.4 (Employee Retreat Rights) have been carried out, the Human Resources Department will review and identify the frozen vacant classifications into which employees ultimately targeted for layoff may be placed on the basis of total experience and education. In making this decision, a waiver of minimum qualification standards, and/or the substitution of related experience and education may be made, with an understanding on the part of management and supervisory personnel that adequate on-the-job training, which can be completed within no more than six (6) months, will be provided to facilitate job adjustment and to compensate for the waiver of qualification standards if that has occurred.

This shall be called the qualification period. The employee shall be advised of their progress after two months and four and six months in the new classification. If at the end of the qualification period the department head notifies the Human Resources Department that the employee is

unable to adequately perform the assignment or fails a test for the position administered by Human Resources then the employee shall be subject to the layoff process.

- 39.6.3 Assignments under the Flexible Placement Program shall be limited to positions in the same or lesser salary range as the classification from which the employee is to be laid off, except that the City Manager may authorize the offer of a flexible placement to position with a maximum salary of no more than five (5) percent above the salary range as the classification from which layoff is targeted, when it is in the best interest of the City service to do so. Whenever flexible placement is made to a classification with a greater salary range, the appointment shall be probationary, in accordance with the terms of that classification.
- 39.6.4 Offers to positions under the Flexible Placement Program shall be made according to seniority service date and in accordance with the probationary and permanent seniority list certification process outlined in Section 39.3 (Establishment of Seniority Lists). All offers and placements made under this provision of the layoff policy shall be documented in detail, with records available for audit and review at all times.
- 39.6.5 If an employee fails to accept a bona fide written offer of an alternative job within ten (10) calendar days after the offer has been made, he or she forfeits further rights to employment retention. Acceptance of an alternative job under the Flexible Placement Program in no way jeopardizes an employee's standing on the reemployment priority lists on which his or her name has been placed in accordance with Section 39.7 (Reemployment Lists).

39.7 Reemployment Lists

- 39.7.1 The names of all probationary and permanent employees released from positions in the competitive service as a result of layoff must be placed on reemployment priority lists for those classifications from which they were separated, as well as all other classifications to which they have retreat rights in accordance with Section 39.4 (Employee Retreat Rights).
- 39.7.2 A reemployment priority list shall remain in effect for three (3) years.
- 39.7.3 Departments with vacancies in any classification for which there is an active reemployment priority list must use the reemployment priority list to fill their positions and may not use any other recruitment or appointment

method to fill a vacancy until appropriate reemployment lists have been exhausted.

- 39.7.4 When a vacancy occurs in a class for which there is a reemployment priority list, the name of the employee on the appropriate reemployment priority list with the highest seniority date shall be certified to the selecting official. Employees so certified from the reemployment priority list must be appointed to the existing vacancy.
- 39.7.5 If a former employee fails to accept a bona fide written offer of reemployment within fifteen (15) calendar days, his or her name will be removed permanently from the reemployment priority list from which the offer was made. Failure to accept an offer of reemployment to the classification with the highest salary range for which the employee is eligible for reemployment will result in automatic removal from all reemployment priority lists. However, the employee may decline (or accept) reemployment to lower salary range classifications without jeopardizing his or her standing on the reemployment priority list for the classification from which he or she was originally terminated.
- 39.7.6 Upon reappointment to the classification from which the employee was originally separated or demoted, the employee has the right to be placed at the step of the salary range which the employee held at the time of layoff or demotion.

39.8 Career Exempt Employees

Only those employees holding full-time, benefited, exempt positions, who in the past have achieved permanent status and have been continuously employed without a break in service between their career and exempt appointment, have the right to retreat to previously held career classifications, placement on the reemployment priority lists, and all other provisions governing layoff procedures. For the purpose of layoff, such employees shall be referred to as "career-exempt."

39.9 Appeal Procedures

Any permanent, probationary, or career-exempt employee who is laid off, demoted, or reassigned as a result of layoff who believes that the layoff procedure has been improperly administered as it pertains to the employee's case may appeal the action under Section 35 (Grievances). In addition, employees may, at all times before, during, and subsequent to layoff, review all records, including seniority lists, reemployment priority lists, documentation pertaining to appointments under the Flexible Placement Program, etc., which pertain to their classification and their rights under the provisions of the layoff policy.

Page 83 of 183 2020-2022 2022-2025 Memorandum Agreement

City of Berkeley

IBEW Local 1245

39.10 Reemployment Priority Rights

If it is determined that a vacancy has been filled by a non-reemployment priority list eligible in a classification for which a reemployment priority list existed and which included available applicants at the time, the former employee with reemployment rights shall be hired and given retroactive pay from the date that the vacancy occurred. The employee who was originally hired to fill the vacancy shall continue to be retained in City employment, provided he or she has completed the probationary period.

SIGNATURE PAGE

representatives whose signatures appear be	
FOR INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS, LOCAL 1245	FOR THE CITY OF BERKELEY
Bob Dean, Business Manager	Dee Williams-Ridley, City Manager
Charley Souders, Business Representative (Chief Negotiator)	LaTanya Bellow Donald Ellison, Interim Director of Human Resources
Greg Marwick, Steward	Andrew Brozyna Dawud Brewer, Employee Relations Manager (Chief Negotiator) Deputy Director of Public Works
Tom HartmanRyan Schaffler, Negotiation Committee	Alicia PlattJoy Brown, Operations Manager Public WorksSenior Human Resources Analyst
	Amani Manning, Assistant Human Resources Analyst

-EXHIBIT A - Hourly Salary Schedule

As of July August 257, 20224 (43.0% Salary Increase)

Job	Rep	Classification Title	FLSA	Step	Step 2	Step 3	Step 4	Step 5
Code	Unit			1		3334		333,
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	46.8878 <u>48.2945</u>	49.1430 50.6173	51.3853 <u>52.9270</u>
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	34.2840 <u>35.3125</u>	36.4983 <u>37.5934</u>	38.4124 <u>39.5648</u>	39.3832 40.5647
51030	С	ELECTRICIAN	N	0	0	46.8878 <u>48.2945</u>	4 9.1430 50.6173	51.3853 <u>52.9270</u>
51100	С	LEAD COMMUNICATION TECHNICIAN	Z	0	0	50.1275 <u>51.6314</u>	52.3965 <u>53.9685</u>	54.912 4 <u>56.5598</u>
51080	С	LEAD ELECTRICIAN	N	0	0	50.1275 <u>51.6314</u>	52.3965 <u>53.9685</u>	54.912 4 <u>56.5598</u>

As of July 7, 2023 (3.0% Salary Increase)

<u>Job</u> <u>Code</u>	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	<u>C</u>	COMMUNICATIONS TECHNICIAN	<u>N</u>	<u>0</u>	<u>0</u>	49.74.33	<u>52.1358</u>	54.5148
<u>52180</u>	<u>C</u>	ELECTRICAL PARTS TECHNICIAN	<u>N</u>	<u>0</u>	36.3719	38.7212	40.7517	41.7816
<u>51030</u>	<u>C</u>	ELECTRICIAN	<u>N</u>	<u>0</u>	<u>0</u>	49.7433	<u>52.1358</u>	54.5148
51100	<u>C</u>	LEAD COMMUNICATION TECHNICIAN	<u>N</u>	<u>0</u>	<u>0</u>	53.1803	<u>55.5876</u>	<u>58.2566</u>
51080	<u>C</u>	LEAD ELECTRICIAN	<u>N</u>	<u>0</u>	<u>0</u>	53.1803	<u>55.5876</u>	58.2566

As of July 9, 2024 (1.0% Salary Increase)

Job Code	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	<u>C</u>	COMMUNICATIONS TECHNICIAN	<u>N</u>	<u>0</u>	<u>0</u>	50.2407	52.6572	<u>55.0599</u>
<u>52180</u>	<u>C</u>	ELECTRICAL PARTS TECHNICIAN	<u>N</u>	<u>0</u>	36.7356	39.1084	41.1592	42.1994
<u>51030</u>	<u>C</u>	ELECTRICIAN	<u>N</u>	<u>0</u>	<u>0</u>	50.2407	52.6572	55.0599
51100	<u>C</u>	LEAD COMMUNICATION TECHNICIAN	<u>N</u>	<u>0</u>	<u>0</u>	53.7121	56.1435	58.8392
51080	<u>C</u>	LEAD ELECTRICIAN	<u>N</u>	<u>0</u>	<u>0</u>	53.7121	<u>56.1435</u>	58.8392

Note: Employee's monthly salary will depend upon the number of hours worked. Employees hired before January 9, 2000, have an option (discussed in detail in Exhibit E (Hours and Days of Work) to work either a 37.5-hour work week or a 40-hour work week. Employees hired after January 9, 2000 work a 40-hour per week schedule.

EXHIBIT C

CITY OF BERKELEY MUNICIPAL CODE CHAPTER 4.04.120

4.04.120 Exemption from Career Service (At-Will Employees).

The provisions of this chapter shall apply to all positions in the service of the city except:

- A. The city manager, assistant city manager, deputy city manager, assistant to the city manager, police review commission officer, police review commission investigator, and assistant to the mayor;
- B. Assistant, associate and senior management analyst when appointed to the city manager's department or to the Budget Unit of the Management and Administrative Services Agency; and secretary to the mayor, administrative secretary and secretary when appointed to the mayor's office;
- C. All department heads, health officer, and supervising psychiatrist;
- D. Persons employed seasonally in the summer camps;
- E. Persons employed as reserve or emergency employees during the period of national emergency as provided in the ordinance creating such employment;
- F. All persons who are paid at an hourly rate with the exception of library aides. Persons appointed as hourly Library Aides shall be part of the career service with the exception of those hired to fill temporary positions of six months or less.

Persons appointed to these positions which are exempt from the career service shall serve at the pleasure of the appointing authority and may be demoted, suspended, or otherwise rejected at any time without cause and without right of appeal or hearing in any manner. Except that, any employee in the career service promoted or transferred to a position not included in the career service and made exempt from the career service shall be reinstated to his or her career appointment from which he or she was promoted or transferred if within six months of appointment to a position not included in the career service, action is taken to dismiss him or her, unless charges are filed and the employee is discharged in accordance with this chapter and the rules established hereunder for positions in

Page 88 of 183 2020-2022 2022-2025 Memorandum Agreement

City of Berkeley

IBEW Local 1245

the career service. All employments designated in this Section shall be entitled to only those benefits provided for at-will employees in the personnel rules and regulations or in applicable memoranda agreements.

EXHIBIT D – Service Vacation Leave

Years During Which Service Rendered	Number of Years of Service	AUTHORIZED ANNUAL VACATION LEAVE (IN CALENDAR WEEKS)
All years prior to 01/01/1950		2
01/01/1950 through 12/31/1956	First 10 years of service Years of service in excess of 10	2 3
01/01/1957 through	First 5 years of service	2
12/31/1961	6 through 25 years of service Years in excess of 25	3 4
01/01/1962 through 12/31/1965	First 5 years of service 6 through 20 years of service	2 3
01/01/1966 through	Years of service in excess of 20 First 5 years of service 6 through 20 years of service	4 2 3
06/30/1970	21 through 25 years of service Years of service in excess of 25	4 5
Subsequent to 06/30/1970	First 5 years of service 6 through 15 years of service 16 through 25 years of service	2 3 4
	Years of service in excess of 25	5
Effective 01/01/1981	First 4 years of service 5 through 12 years of service 13 through 20 years of service	2 3 4
	First 3 years of service 4 through 11 years of service	2 3
Effective 07/01/1984	12 through 17 years of service 18 and subsequent years of service	4 5
	First 3 years of service 4 through 11 years of service	2 3
Effective 12/08/1987	12 through 17 years of service 18 through 24 years of service 25 and subsequent years of service	4 5 6

EXHIBIT E - HOURS AND DAYS OF WORK

The hours and days of work applicable to employees in Representation Unit C, as set by the City Manager, are presently as follows:

1. Hours per Day

Field Operations: Working arrangements may vary as to daily schedules under the

reduced workweek.

Flexible Hours: On a job-by-job basis, with prior approval of the Senior Electrical

Supervisor, a crew may schedule a workday, or workdays, other than

the normal workday schedule of 7:00 a.m. to 3:30 p.m.

2. Hours per Week (Basic)

Field Operations: 40 hours. However, employees in Representation Unit C as of January 9, 2000 have the option of retaining a 37.5-hour per week work schedule. An employee will have an option to change his work schedule to 40 hour per week at the beginning of a payroll period. An employee will have a one-time option to revert to a 37.5 hour per week work schedule at the beginning of a payroll period. After exercising this one time option, if the employee chooses to change his work schedule to 40 hours per week, the employee must maintain this work schedule for the balance of his career in this Representation Unit. All persons hired on or after January 9, 2000 will work a 40-hour per week schedule and will not be permitted to change his or her work schedule to 37.5 hours per week.

The parties may agree in writing to alternative work schedules other than those set forth in this Agreement, such as a 9-80 work schedule which produces an 80-hour pay period with the hours worked in nine work days. The City will take into consideration its operational needs, including the impact of potential overtime, in deciding whether to approve an alternate work schedule. However, such alternate work schedule will not be unreasonably denied.

3. Work Week

The work week will begin at 12:01 a.m., Sunday and end at Saturday midnight. Regular days off will be considered to be Saturday and Sunday.

City of Berkeley

IBEW Local 1245

EXHIBIT F - IMPLEMENTING THE REDUCED WORK SCHEDULE

NOTE: This Exhibit F applies only to employees hired before January 9, 2000 who are permitted to work a 37.5 hour per week work schedule subject to the conditions set forth in Exhibit E (Hours and Days of Work).

The City and the Union agree that the implementation of the reduced work schedule which is to be effective June 29, 1986 will be as follows:

That one-fourth of the Electrical Division employees will be off every fourth Monday on a staggered basis. This time would be treated as a scheduled day off and could not be adjusted or moved. A yearly schedule would be set up at the beginning of the year so that each employee would know what days he would have off. Since the agreed upon reduced work schedule does not calculate evenly into one day off every four weeks, it is proposed that an additional day off (Friday) be given once every 16 weeks. This agreement very closely approximates the reduced work schedule. Any discrepancies would be credited to the employee's vacation at the end of the contract year.

For Fair Labor Standards Act purposes, the workweek is defined as 40 hours.

Sick leave, vacation, and other compensable leaves of absences shall be based on the 40 hour work week.

When a scheduled day off falls on a holiday, the next regularly scheduled work day shall be observed as the scheduled day off.



EXHIBIT G

DIRECTIVE PROHIBITING UNAUTHORIZED CONSTRUCTION

Office of the City Manager

July 31, 2001

To:

Department Directors

From:

Weldon Rucker City Manager

Subject:

DIRECTIVE PROHIBITING UNAUTHORIZED CONSTRUCTION OR FACILITY MAINTENANCE WORK IN CITY FACILITIES

It has come to my attention that some construction, building maintenance and electrical work is being improperly performed at various facilities that are either owned or leased by the City. Because this work can affect the health and safety of all City employees and can result in a significant potential liability for the City, it is absolutely essential that any such work be done by competent and experienced individuals. Even if licensed contractors are brought in to do the work, they may not be fully aware of how their work will impact other City projects or facilities. Furthermore, already scarce City funds are wasted when Public Works has to correct improper or hazardous work performed by unauthorized individuals or contractors. Accordingly, such construction and maintenance work must only be performed by properly trained and supervised Public Works maintenance staff or contractors selected by, and under the direction of, the Public Works Department.

Therefore, I am reminding and directing all department directors to ensure that absolutely no construction, building maintenance or electrical work be performed in any City owned or leased facilities by contractors or anyone else not under the direction of Public Works. All construction and facility maintenance work is to be referred to the Public Works Department.

For routine maintenance, minor repairs, or minor physical modifications, contact the Public Works Customer Service representative at 644-6620. For larger construction projects, contact Public Works Capital Improvement Projects at 981-6330. Public Works can then make appropriate decisions as to whether construction or maintenance work should be performed by City employees or by outside contractors, utilities, or agencies.

Thank you, in advance, for your full compliance with this direction.

cc: City Manager's Office

2180 Milvia Street, Berkeley, CA 94704 Tel: 510.981.7000 TDD: 510.981.6903 Fax: 510.981.7099 E-mail: manager@ci.berkeley.ca.us





MEMORANDUM AGREEMENT

Between the

CITY OF BERKELEY

and the

INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 1245 - A.F.L. - C.I.O.

July 1, 2022 to June 30, 2025

Page 94 of 183

Table of Contents (click section titles to jump to page)

ARTICLE 1 - AD	MINISTRATION	1
SECTION 1:	RECITALS	1
SECTION 2:	PARTIES TO THE AGREEMENT	2
2.1 Exclusive F	Representation	2
2.2 Manageme	nt Rights	2
SECTION 3:	NO DISCRIMINATION	2
SECTION 4:	UNION SECURITY	3
4.1 Union Secu	ırity	3
4.3 New Emplo	oyee Orientation	3
SECTION 5:	DEDUCTION OF UNION DUES	5
5.1 Deductions		5
5.2 Revocation		5
5.3 Authorizati	on Forms	5
5.4 Notice		5
SECTION 6:	UNION REPRESENTATIVES	5
6.1 Number of	Representatives	6
6.2 Maximum	Representatives	6
6.3 Notification	1	6
6.4 Union/Man	agement Meetings	6
6.5 Union Train	ning	6
SECTION 7:	SEPARABILITY OF PROVISIONS	7
SECTION 8:	EXISTING MEMORANDUM AGREEMENT	7
SECTION 9:	FINALITY OF RECOMMENDATIONS	7
SECTION 10:	DURATION	7
ARTICLE 2 - SA	LARIES, HOURS OF WORK AND COMPENSATION ISSUES	7
	SALARIES	
11.1 Salary Rates		8
11.2 Step Placeme	ent	8
•	alary Rate	
11.4 Effect of Lea	ves	9
11.6 Y-Rate		9
	cy	
• •	ification	
	appointment	
*	ses for Provisional Employees	
	r Pay Period	
	nderpayment	
	ies	
	se	
	ate of Salary Adjustment	
	tion	

Page 95 of 183

11.17 Premium Pay Differential - Working at Heights of 60 Feet or More	.12
11.18 Commercial Driver's License Premium	.12
11.19 Longevity Pay	.12
SECTION 12: HOURS AND DAYS OF WORK	.13
12.1 Hours and Days of Work	.13
12.2 Sunday and Graveyard Shifts	.13
12.3 Shift Assignment	
12.4 Classes during Shift Assignments	.14
12.5 Rest Period	.14
12.6 Daylight Saving Time	.14
SECTION 13: OVERTIME	
13.1 Rate	.15
13.2 Department Head Discretion	.15
13.3 Payoff of Overtime	.15
13.4 FLSA Workweek	.15
13.5 Emergency Overtime	.15
13.6 Meals	.16
13.7 Standby Pay	.16
13.8 Emergency Overtime Assignments	.17
13.9 Overtime Offers	.17
SECTION 14: SHIFT DIFFERENTIAL	.18
14.1 Swing Shift	.18
14.2 Night Shift	.18
SECTION 15: NATURAL DISASTER/DECLARED EMERGENCY	.18
15.1 Call Outside Normal Working Hours	.19
15.2 Shift Differential	.19
15.3 Reduction in Hours	.19
15.4 Change in Assignments	.19
15.5 Hours Worked in Excess of 12-hours	.19
ARTICLE 3 - LEAVES	.19
SECTION 16: VACATION	.19
16.2 Vacation Approval	.19
16.3 Use of Incremental Vacation	.20
16.4 Accrual Rate as of December 8, 1987	.20
16.5 Accrual, Use, and Limitation for Employees with Less than 6-Months of Service	.20
16.6 Vacation Eligibility for Part-time, Intermittent, Leave Without Pay or	.21
Reemployed Employees	.21
16.7 Maximum Vacation Accumulation	.21
16.8 Prorated Vacation due to Extended Absence, Reemployment or Reinstatement	.22
16.9 Payment of Vacation upon Termination or Extended Leave after Six Months of Service	.22
16.10 Calculation and Payment or Reimbursement of Vacation Credits upon Termination	or
Extended Leave	.22
16.11 Lump Sum Payment of Vacation	.23
16.12 Calculation of Vacation Based on Actual Paid Hours	.23

Page 96 of 183

16.13 Vacation Accounting Procedures.	23
16.14 Voluntary Time Off	23
SECTION 17: HOLIDAYS	23
17.1 Holidays	23
17.2 Floating Holidays	24
17.3 Floating Holiday Accrual upon Employment	24
17.4 Limitations	25
17.5 Holidays for Employees whose Work Week is Other than Monday through Friday.	25
17.6 Work during a Holiday	25
17.7 Holidays for Part-Time Employees	25
SECTION 18: SICK LEAVE	26
18.1 Eligibility	26
18.2 Sick Leave Accrual	26
18.3 Pro-rated Sick Leave for Part-Time Employees	26
18.4 Sick Leave for Intermittent Employees	26
18.5 Maximum Accumulation	26
18.6 Payment upon Retirement/Termination	26
18.7 Annual Payment for Hours in Excess of Maximum for Employees Working Half	-Time or
More for Employees Hired on or Before June 30, 2013	27
18.8 Restoration of Sick Leave if Reemployed within Two Years	28
18.9 Sick Leave Use/Family Illness	28
18.10 Prohibition for Use with Outside Employment	28
18.11 Notification Requirement	
18.12 Cessation of Accrual	29
18.13 Workers' Compensation	29
18.14 Light Duty	30
18.15 Control Program for Sick Leave Use	31
18.16 Bonus Time for Unused Sick Leave	31
18.17 State Disability Insurance	32
SECTION 19: FUNERAL LEAVE	33
SECTION 20: MILITARY AND MARITIME LEAVE	33
SECTION 21: PARENTAL LEAVE	34
21.1 Eligibility:	34
SECTION 22: LEAVE OF ABSENCE WITHOUT PAY	35
22.1 Department Head Authority	35
22.2 Union Sponsored Training	35
22.3 Exhaustion of Accrued Time	35
SECTION 23: JURY DUTY LEAVE	35
23.1 Court Time	36
ARTICLE 4 - HEALTH AND WELFARE BENEFITS	37
SECTION 24: HOSPITAL-MEDICAL, DENTAL, AND VISION COVERAGE	37
24.1 Medical Coverage	37
24.2 Maximum Medical Premium Payments	37
24.3 Dental Coverage	38

Page 97 of 183

24.4 Vision Coverage	39
24.5 Change in Insurance Carrier	39
24.6 Domestic Partner Coverage	40
24.7 Part-Time Employees and Prorated Benefits	40
24.8 Commencement of Medical, Dental, and Vision Benefits	40
24.9 Health Insurance In-Lieu Payments	40
24.10 Group Life Insurance	40
24.11 Pre-Tax Status	40
SECTION 25: RETIREE MEDICAL COVERAGE	41
25.1 Amendment of Retiree Health Premium Assistance Plan III, effective January 23,	1998,
Restated and Amended effective March 22, 2011	41
25.2 Eligibility	42
25.3 Pre Age 65 Retiree Health Insurance	42
25.4 Retiree Benefits for Employees Age 65 and Over	44
25.5 Termination by City of Retiree Medical Benefit	45
25.6 Retiree Medical Benefit for Employees Retiring Between the Ages of 50 and 55	45
25.7 Employees Retiring with a CalPERS Approved Disability Retirement	45
25.8 City Funding of Retiree Health Benefit	45
25.9 Actuarial Study	46
ARTICLE 5 – TERMS AND CONDITIONS OF EMPLOYMENT	47
SECTION 26: PROBATIONARY PERIOD	47
26.1 Duration and Effect of Military Leave on Probationary Period	
26.2 Provisional Appointments to a Higher Classification	47
26.3 Report Requirement	47
26.4 Rejection during Probationary Period	47
26.5 Rejections Subject to Disciplinary Appeal	48
26.6 Probationary or Temporary Employee Assignments	48
SECTION 27: TRANSFER	48
SECTION 28: PROMOTION	48
SECTION 29: DEMOTION	49
SECTION 30: SUSPENSION	49
SECTION 31: DISCHARGE	49
SECTION 32: RESIGNATION	50
SECTION 33: REINSTATEMENT	50
SECTION 34: NOTICE OF TERMINATION	50
ARTICLE 6 - GRIEVANCE AND DISCIPLINARY APPEAL PROCEDURE	50
SECTION 35: GRIEVANCES	50
35.1 Definition	50
35.2 Procedure	51
35.3 EEO Complaints	54
35.4 Compensation Complaints	55
35.5 Changes or Interpretations	55
35.6 Extension of Timelines	55
35.7 Days	55

Page 98 of 183

35.8 Grievances Challenging Disciplinary Action	55
ARTICLE 7 – GENERAL PROVISIONS	57
SECTION 36: GENERAL PROVISIONS	57
36.1 Personal Conduct	57
36.2 Rain Gear	57
36.3 Tools	57
36.5 Training	58
36.6 Use of Automobiles	59
36.7 Safety	60
36.8 YMCA Group Membership	60
36.9 Annual Performance Evaluation	60
ARTICLE 8 - PUBLIC EMPLOYEES RETIREMENT SYSTEM	61
SECTION 37: PAYMENT OF EMPLOYEES' PERS CONTRIBUTIONS	61
37.1 Miscellaneous Designation	61
37.2 "Classic Employees" Definition	61
37.3 "New Member" Definition	
37.4 CalPERS Retirement Formula for "New Members" as Defined Under the	61
Public Employees' Pension Reform Act of 2013 (PEPRA)	61
37.5 CalPERS Retirement Formula and Employer Paid Member Contribution for Cla	ssic
Employees (i.e. current employees and employees who do not qualify as "New Members" ur	
PEPRA)	
37.6 CalPERS New Members Normal Share of Cost	61
37.7 Reporting of Contributions	61
37.8 City Contracts with CalPERS	62
37.9 Index Level 1959 Survivor Benefit	
37.10 Classic Members' Pension Contribution through 20516 a Contract Amendment	63
37.11 New Members' Pension Contributions through 20516 a Contract Amendment	63
SECTION 38: SUPPLEMENTAL RETIREMENT/DISABILITY INSURANCE PLAN	
ARTICLE 9 - LAYOFF PROCEDURE	65
SECTION 39: LAYOFF	65
39.1 Announcement of Layoff	65
39.2 Seniority Service Date	
39.3 Establishment of Seniority Lists	
39.4 Employee Retreat Rights	
39.5 Employee Notification	67
39.6 Flexible Placement Program	
39.7 Reemployment Lists	
39.8 Career Exempt Employees.	
39.9 Appeal Procedures	
39.10 Reemployment Priority Rights	
SIGNATURE PAGE	
EXHIBIT A - Hourly Salary Schedule	
EXHIBIT C	
EXHIBIT D – Service Vacation Leave	

Page 99 of 183

EXHIBIT G	76
EXHIBIT F - IMPLEMENTING THE REDUCED WORK SCHEDULE	79
EXHIBIT E - HOURS AND DAYS OF WORK	78

MEMORANDUM AGREEMENT Between CITY OF BERKELEY and the INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 1245

ARTICLE 1 - ADMINISTRATION

SECTION 1: RECITALS

- 1.1 This Memorandum Agreement is entered into pursuant to the Meyers-Milias Brown Act (Government Code, Sections 3500-3511), as amended, and has been jointly prepared by the parties.
- 1.2 The City Manager is the representative of the City of Berkeley (hereinafter referred to as "the City") in employer-employee relations as provided in Resolution No. 43,397-N.S., adopted by the City Council on October 14, 1969.
- 1.3 International Brotherhood of Electrical Workers, Local 1245, is the recognized employee organization for Representation Unit C (Electrical Occupations), which organization has been certified as such pursuant to said Resolution No. 43,397N.S. The employee positions in such Representation Unit are hereinafter set forth in Exhibit A attached hereto and made a part hereof, and International Brotherhood of Electrical Workers, Local 1245 (hereinafter referred to as "the Union") is recognized as the sole representative of employees assigned to such positions.
- 1.4 The parties have met and conferred in good faith regarding wages, hours, and other terms and conditions of employment of the employees in said Representation Unit C, have exchanged freely information, opinions, and proposals, and have endeavored to reach agreement on all matters relating to the employment conditions and employer-employee relations of such employees.
- 1.5 This Memorandum Agreement shall be presented to the City Council as the joint recommendation of the undersigned.
- 1.6 Section titles in this Memorandum Agreement are for identification purposes only and are not to be used for the purpose of interpreting either the intent or the meaning of the language of any section.

SECTION 2: PARTIES TO THE AGREEMENT

2.1 Exclusive Representation

The Union is the exclusive representative of all employees within Representation Unit C (Electrical Occupations) and shall continue to be recognized as such unless, in accordance with the provisions of Resolution No. 43,397-N.S. or said Resolution may be amended, the Union is no longer certified as the Recognized Employee Organization for employees in Representation Unit C.

2.2 Management Rights

The Union recognizes that the management of the City and its business and the direction of its working forces are vested exclusively in the City, and this includes, but it not limited to, the following: to direct and supervise the work of its employees, to hire, promote, demote, transfer, suspend, and discipline or discharge employees for just cause; to plan, direct, and control operations; to lay off employees because of lack of work or for other legitimate reasons; to introduce new or improved methods or facilities, provided, however, that all of the foregoing shall be subject to the provisions of this Agreement, arbitration decisions, or letters of agreement, or memorandums of understanding clarifying or interpreting this Agreement.

The City shall not by reason of the execution of this Agreement (a) abrogate or reduce the scope of any present plan or rule beneficial to employees, such as its vacation and sick leave policies or its retirement plan, or (b) reduce the wage rate of any employee covered hereby, or change the conditions of employment of any such employee to his disadvantage. The foregoing limitation shall not limit City in making a change in a condition of employment if such change has been negotiated and agreed to by City and Union.

SECTION 3: NO DISCRIMINATION

The City and Union agree that they will not discriminate against employees based on race, creed, color, ethnicity, ancestry, religion, political affiliation, gender, sexual orientation, age, national origin, marital or domestic partner status, gender identity or gender expression, parental status, pregnancy, disability or medical condition, Acquired Immune Deficiency (AIDS/HIV) or AIDS related condition, or any other status protected by applicable state or federal law, or protected Union activity. Furthermore, the City and Union agree to comply with all applicable federal, state and local laws pertaining to nondiscrimination and equal employment opportunity.

The Union agrees that it will support programs for making members of minority groups and women aware of employment opportunities within the City's jurisdiction, and that it will work with the City to increase recruitment efforts of such minorities and women into City of Berkeley IBEW Local 1245

City service. The Union recognizes and supports the City of Berkeley's commitment to equal employment opportunity.

SECTION 4: UNION SECURITY

4.1 Union Security

- 4.1.1 An employee included in Unit C, may become and remain a member in good standing of the Union; or pay to the Union an initial fee equal to the regular initiation fee and, thereafter, a monthly fair share service fee equal to the regular monthly union dues and general assessments.
- 4.1.2 **Contracting Out:** The City will not hire any contractors, firm or individual which will result in the layoff of any employees covered under this Agreement.
- 4.1.3 Subcontracting: If the City intends to subcontract any electrical or communication work, the City will notify the Union in advance of any bid solicitation whose value is estimated to be \$25,000 or more. The City will notify the Union in advance by mailing a copy of the City Council Bid Notifications Report to the Union Stewards. Information will be sent to the Union at the same time it is forwarded to the City Council and will include nature of project, the name of department requesting the work, and the budget code. Upon written request from the Union, the City will provide a copy of the actual bid specifications.

For electrical or communication work estimated to cost less than \$25,000, the City will provide the Union Stewards a copy of the Council Awarded Contracts Report on a monthly basis.

- 4.2.3 Indemnification: The Union shall indemnify and save harmless the City, its officers and employees, from and against any and all loss, damages, costs, expenses, claims, attorneys' fees, demands, actions, suits, judgments and other proceedings arising out of any claims made by Unit C employees for deductions made in reliance on information provided by the Union.
- 4.2.4 Union Reports: The City shall furnish the Union, on a monthly basis, the name, date of hire, salary, classification and work location of all newlyhired employees subject to this Agreement.

4.3 New Employee Orientation

This provision shall apply to new hire employees appointed to classifications covered by this Agreement.

4.3.1 Onboarding: The parties acknowledge that the City provides a new employee orientation (onboarding) to each new employee hired by the City. As such, the Union will be provided with not less than 10 calendar days' advanced notice of the time, date, and location of the onboarding of any new employee represented by the Union. The Union will be given 30-minutes at the start of the new employee onboarding in a room designated by the City for no more than one (1) representative to present Union membership information. The City representative will excuse him or herself during the Union portion of the onboarding. The Union agrees in its portion of the onboarding not to engage in speech that could cause disruption or material interference with City activities.

The City will provide 30 minutes of Union Release Time to the Union representative presenting the Union membership information during the scheduled onboarding. The Union shall provide the Union representative's immediate supervisor with the Union representative's name at least five (5) days prior to the onboarding. The Union representative shall be released for this purpose unless unusual operation needs interfere with such release in which case the Union representative's immediate supervisor will provide a written explanation of why release could not be approved. If the Union representative is not released due to department operational needs, the Union representative may arrange an alternative date and time to meet with the newly hired employee within the first two (2) weeks of employment, subject to the 30-minutes onboarding and Union Release Time requirements as stipulated above.

4.3.2 Information Provided: On a quarterly basis (March, June, September, and December), the City will provide the Union with a digital file via email to the email address designated by the Union. The Union acknowledges and understands that the City is working diligently and in good faith to update its contact information database functionality to incorporate all the fields of contact information listed below. As a result, the City may not initially be able to provide the employee's work telephone number, personal phone number, and personal email address until the completion and implementation of the City's new Enterprise Resource Management Application (ERMA) system.

The City will provide the Union with the following information on file, to the extent the City has it on file:

- Name.
- Job Title.
- Department.
- Work Location.

City of Berkeley

IBEW Local 1245

- · Home telephone number.
- Home address.
- Personal cellular telephone number (new hires hired on or after October 1, 2017).
- Work telephone number (upon implementing ERMA).
- Personal email addresses on file with the City (upon implementing ERMA).

Notwithstanding the foregoing, limited to the express purpose of the requirements of Government Code section 3558 only, an employee may opt out via written request to the City (copy to the Union) to direct the City to withhold disclosure of the employee's: Home address; home telephone number; personal cellular telephone number; personal email address; and date of birth.

SECTION 5: DEDUCTION OF UNION DUES

5.1 Deductions

The City shall deduct, once monthly, the amount of Union regular and periodic dues, service fees, or insurance premiums as may be specified by the Union under the authority of written notification by the Union certifying that the employee has provided written affirmed consent for deduction of union membership dues or fair share service fees.

Such deductions, together with a written statement of the names and amounts deducted, shall be forwarded promptly to the Union office.

5.2 Revocation

Upon receipt of a written notice from the Union to the Human Resources Department, the City shall cease deducting membership dues or fair share service fees from the designated employee.

5.3 Authorization Forms

The City may request verification of employee's authorization form giving affirmed consent to deduct dues or fees only if a legitimate dispute arises about the existence or term of the authorization.

5.4 Notice

Union agrees to give reasonable notice of dues change.

SECTION 6: UNION REPRESENTATIVES

City of Berkeley IBEW Local 1245

6.1 Number of Representatives

The City shall allow two (2) representatives of the Union reasonable time off from work, without loss of compensation or other benefits, to represent its members in disputes which involve the interpretation or application of those rules, regulations, and resolutions which have been or may hereafter be adopted by the City Council to govern personnel practices and working conditions, including such rules, regulations, and resolutions as may be adopted by the City Council to effect memoranda of understanding which may result from the meeting and conferring process, and to represent its members in meeting and conferring in good faith for amendments to this Memorandum Agreement in the future, subject to the conditions set forth in Sections 6.2 (Maximum Representatives) and 6.3 (Notification).

All release time shall be recorded on time sheets and time cards with appropriate codes.

6.2 Maximum Representatives

Two (2) Union representatives shall be the maximum number who will be allowed concurrent time off.

6.3 Notification

The representatives shall notify the first non-bargaining supervisor a minimum of one workday in advance before leaving their work assignments except for emergency situations which require the immediate attention of said representatives (an emergency is defined as a safety problem or hazardous condition), but in no case shall an employee leave his or her job without notification of the first non-bargaining supervisor. Such request shall include the location, and area of activity, the approximate time needed and the general nature of the union business involved. Such request shall not be unreasonably denied. Failure to comply with notification and time recording rules may be grounds for denial of leave.

6.4 Union/Management Meetings

Quarterly meetings shall be held between the union and management. Release time shall be granted for up to two union representatives. Agendas for such meetings shall be set one week in advance between the Business Representative of the Union and the City Manager or their designees. Meetings within department may be held at the discretion of the department head.

6.5 Union Training

Conditional upon prior approval of the course content and upon receipt of certification of completion, if the employee is an elected official or steward of the Union, the City shall reimburse the employee for up to one-half of his or her regular work time spent in such training at the employee's permanent rate of pay, not to exceed twenty (20) hours of paid leave in a calendar year.

SECTION 7: SEPARABILITY OF PROVISIONS

In the event that any provisions of this Memorandum Agreement are declared by a court of competent jurisdiction to be illegal or unenforceable, that provision of this Memorandum Agreement shall be null and void, but such nullification shall not affect any other provisions of this Memorandum Agreement, all of which other provisions shall remain in full force and effect.

SECTION 8: EXISTING MEMORANDUM AGREEMENT

Pursuant to Section A of the Memorandum of Understanding dated August 9, 1972 by and between the City and the Union, this Memorandum Agreement shall supersede all existing Memoranda of Understanding between the City and the Union. There is no guarantee that working conditions and practices will be continued if they are not included in this Memorandum Agreement or have not been or are not hereafter specifically authorized by ordinance or by resolution or the City Council.

SECTION 9: FINALITY OF RECOMMENDATIONS

The recommendations set forth herein are final. No changes or modifications shall be offered, urged, or otherwise presented by the Union, or the City Manager prior to October 20, 2018; provided, however, that nothing herein shall prevent the parties to this Memorandum Agreement from meeting and conferring and making modifications herein by mutual consent.

SECTION 10: DURATION

This Memorandum Agreement shall be effective upon ratification by the City Council except for those provisions which have been assigned other effective dates as herein set forth, and shall remain in full force and effect up to and including June 30, 2025. This Memorandum Agreement and all its rights, obligations, terms and provisions shall expire and otherwise be fully terminated at midnight June 30, 2025.

ARTICLE 2 - SALARIES, HOURS OF WORK AND COMPENSATION ISSUES

SECTION 11: SALARIES

11.1 Salary Rates

Salary rates for the period of 0July 1, 2022 through June 30, 2025, for all classes of positions in Representation Unit C shall be set according to the classifications and salary ranges assigned to those classifications listed in Exhibit "A" and attached hereto and made part hereof.

11.1.1 **Salary Increase upon Council Approval:** The salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase as follows: Effective the first full pay period following council approval, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of three percent (3.0%).

Effective July 9, 2023, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of three percent (3.0%).

Effective July 7, 2024, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of one percent (1.0%).

11.1.2 **One-Time Payment**: A onetime non-pensionable lump sum payment of \$1,000.00 for each career employee in this unit shall be paid the first full pay period after council approval. Eligible employees must be current City employees in paid status for the full pay period this lump sum is conferred and shall not apply to unit members who are in unpaid status. Such payment shall be subject to required payroll tax deductions. This lump sum provision does not create or bind the City to any precedent or past practice.

If during any fiscal year during the term of this agreement, the City reaches agreement with the largest bargaining unit or extends the unrepresented employees an across the board Cost of Living Adjustment (COLA) increase, wage increase, lump sum payment, or other benefit increase the City agrees to apply these same increases to the salary ranges and/or benefits for those classifications covered by this MA as listed in Exhibit B.

City of Berkeley IBEW Local 1245

11.2 Step Placement

Employees occupying a position in the competitive service shall be paid a salary or wage within the range established for that position's class as set forth in Exhibit "B". The minimum rate for the class shall apply to employees upon original appointment to the position except as may be determined by the City Manager within the negotiated wage ranges. Employees who have been laid-off and are rehired to the same classification shall be placed on the same wage step they were in when laid-off. Transfers shall not affect an employee's salary rate. Employees appointed to any of the positions set forth in Exhibit "B" and employed or working on a part-time basis shall be paid in proportion to the time worked and described in their appointment.

11.3 Maximum Salary Rate

No salary advancement shall be made so as to exceed the maximum rate established for the class to which the advanced employee's position is allocated. Advancement shall be in accordance with the compensation plan of the City and shall depend upon increased service value of an employee to the City as exemplified by recommendations of the department head, performance record, special training, length of service, and other pertinent evidence.

11.4 Effect of Leaves

An employee's pay increase shall not be affected by any leave of absence without pay if the employee is off the payroll for less than one hundred sixty (160) consecutive hours. If the employee is off the payroll for more than one hundred sixty (160) consecutive hours, the total amount of time off shall be made up before the employee shall be entitled to such pay increase.

11.6 Y-Rate

Any employee occupying a position which is reallocated to a class, the maximum salary for which is less than the incumbent's present salary, or occupying a position in a class, the salary rate or range for which is reduced, shall continue to receive his present salary. Such salary shall be designated as a Y rate. When an employee on a Y rate vacates his or her position, subsequent appointments to that position shall be made in accordance with Section 11.1 (Salary Rates).

11.7 Pay Frequency

Payment of salaries shall be made bi-weekly. Each pay period shall begin at 12:01 a.m. Sunday up to and including 12:00 midnight Saturday two weeks following. Each payment shall be made not later than the Friday following the end of each payroll period and shall include payment for all earnings during the previous payroll period.

- 11.7.1 **Full-Time:** Bi-weekly payment to full-time employees shall be made on the basis of the hourly rate, as set forth in Exhibit B, multiplied by the number of hours worked during the pay period.
- 11.7.1.1 **40 Hour Week**: For employees on a forty (40) hour week schedule, the annual salary shall be the product of the hourly salary times 2,080 hours.
- 11.7.1.2 **37.5 Hour Week**: For employees on a thirty-seven and one-half (37.5) hour work week schedule, the annual salary shall be the product of the hourly salary times 1,950 hours.
- 11.7.1.3 **Monthly Salary**: The monthly salary shall be the quotient of the annual salary divided by 12.
- 11.7.2 **Part-Time:** Bi-weekly payment to part-time employees in a class for which a monthly salary rate has been herein established shall be made on the basis of the hourly rate multiplied by the number of hours worked during the pay period. The hourly rate for such part-time employees shall be computed in the same manner as for full-time employees.
- 11.7.3 Intermittent: Bi-weekly payment to intermittent employees in a class for which a monthly salary rate has been established shall be made on the basis of the hourly rate multiplied by the number of hours worked during the pay period. The hourly rate for such intermittent employees shall be computed in the same manner as for full-time employees.
- 11.7.4 **Bi-Weekly Daily Rate:** Bi-weekly payment to employees in a class for which a daily rate rather than a monthly rate has been herein established shall be made on the basis of the actual days worked during the biweekly pay period multiplied by the daily rate.
- 11.7.5 **Bi-Weekly Hourly Rate:** Bi-weekly payment to employees in a class for which an hourly rate rather than a monthly rate has been herein established shall be made on the basis of the hourly rate multiplied by the number of hours worked during the pay period.

11.8 Higher Classification

The Department Head will work all employees within their career classifications. The departments may specifically assign an employee to work in a higher class. Such assignments shall be in writing and shall indicate the reasons, length and duties of the assignment. Assignments over one (1) week shall be approved in advance by the City Manager or his or her designee. Blanket authorizations for department will be issued on a yearly basis for applicable positions. To be eligible

for higher-class pay, the employee must work a minimum of four (4) hours, meet the minimum qualifications, and perform the duties of the higher class. Employees meeting these requirements will be compensated at the lowest step of the higher classification which provides at least a five percent (5%) differential. The journey level employees will rotate the assignment on a monthly basis and perform the necessary duties. In the event the offer of higher class is declined by the journey level employee assigned for that month, volunteers will be asked according to the rotation list. If there are no volunteers, the journey-level employee assigned for that month will be required to perform the necessary duties.

11.9 Temporary Appointment

When a temporary vacancy is to be filled in a classification for which there is an existing eligibility list, the City shall attempt to make the temporary appointment from that list.

11.10 Step Increases for Provisional Employees

An employee who holds a provisional appointment in a classification shall receive step increases in such classification as if the employee held a permanent appointment thereto.

11.11 End of Year Pay Period

For all salary and benefit purposes, the parties agree that the last day of the last pay period ending in the calendar year shall be the end of the year. For excess leave only, the end of the year shall be treated as the last day of the pay period nearest March 31st.

11.12 Overtime Underpayment

The City will include any payment of overtime, which was inadvertently omitted or miscalculated, in the following pay period's check.

11.13 Equity Studies

A list of comparison jurisdictions is established for the purpose of salary equity studies: Alameda County, Contra Costa County, Oakland, Palo Alto, Richmond, San Francisco, San Jose, San Leandro, Santa Clara, and Vallejo. Other jurisdictions may be added as required by agreement between the parties. Job classifications which fall below the median for these jurisdictions may be reviewed. It is the policy of the City that within available funding limits, equity adjustments which are in the interests of the service will be considered.

11.14 Step Increase

Step increases shall be effective for payroll purposes only on the first day of the pay period nearest to the actual anniversary date. Personnel records will maintain actual dates and will be used to resolve any discrepancies or questions that may arise.

11.15 Effective Date of Salary Adjustment

Any general future salary adjustments will be set to become effective on the first day of the pay period.

11.16 Reclassification

In the event the City reclassifies a position based on an employee initiated review from a lower level classification to a higher level classification, an incumbent occupying such position shall be reclassified without competitive examination provided he or she has performed the duties of the new classification for one (1) year and has not received an unsatisfactory evaluation during that period. All other employees shall be required to pass an examination for the higher classification and shall serve the normal probationary period. If a position is reclassified, the employee shall receive salary compensation retroactive to the date the position description questionnaire is submitted and verified as having been received by the department head. Position description questionnaires will be provided by the City. Employees may request one (1) job audit one (1) year after an initial job audit has been completed on their position. Thereafter one (1) job audit will be permitted every two (2) years.

Reclassification or reallocation of positions shall not be used as a mechanism, the sole purpose of which is to improperly circumvent the provisions of this Agreement, including provisions relating to layoff, transfer, demotion or promotion.

11.17 Premium Pay Differential - Working at Heights of 60 Feet or More

Assignments for working at heights of 60 feet or more will be on a voluntary basis. However, if all qualified employees decline the assignment, the City will be free to contract the work out to an outside service provider.

Employees who work at heights of 60 feet or more, including but not limited to performing maintenance of lights at San Pablo Park, shall receive Height Premium Pay Differential of one and one-half times (1½) times the normal hourly rate for all time performing such work. The Height Premium Pay Differential shall be reported to CalPERS as Height Premium Special Assignment Pay. However, any hours worked on overtime are excluded from CalPERS reported "compensation earnable" in California Government Code Section 20635.

11.18 Commercial Driver's License Premium

Employees required to possess and maintain a valid California Class A or Class B Commercial Driver's License as a condition of employment shall receive three percent (3%) differential to hourly rate. This differential shall be reported to CalPERS as Special Class Driver's License Pay. However, any hours worked on overtime are excluded from CalPERS reported "Compensation earnable" in California Government Code Section 20636.

11.19 Longevity Pay

Effective the first full pay period after Council approval on its Regular Agenda, employees completing nineteen (19) years of service shall receive a three percent (3.0%) differential beginning with the anniversary date of beginning of the twentieth (20th) year of service and shall apply to all hours in a paid status. This Longevity Pay shall be reported to CalPERS as Longevity Pay Incentive Pay.

11.20 State Certification Premium

An employee who possesses and maintains a "General Electrician" state certification from the Department of Industrial Relations shall receive an additional three percent (3%) to their base rate of pay. The City shall retain the right to add the general electrician state certification as a minimum qualification for the Electrician and Lead Electrician classifications. If the certification is made a minimum qualification, active employees, employed in classifications within this unit at the time of implementation, shall be exempt from this qualification for the remainder of their employment.

SECTION 12: HOURS AND DAYS OF WORK

12.1 Hours and Days of Work

Hours and days of work shall be governed by rules established by the City Manager. (At the present time, the hours and days of work are as shown in Exhibit E (Hours and Days of Work)

12.2 Sunday and Graveyard Shifts

Regularly scheduled Sunday shifts and "graveyard" shifts, as defined in Section 14 (Shift Differential), shall be for eight (8) consecutive hours, including up to one-half (½) hour for lunch.

12.3 Shift Assignment

Within a given classification, shift assignments shall first be offered to employees by classification seniority on a voluntary basis. In the event shift assignments are not filled voluntarily, such assignments will be made on the basis of inverse classification seniority.

12.3.1 Types of Shifts:

- a. **Swing shift** means authorized work schedules regularly assigned in which at least four (4) hours worked are between the hours of 5:00 p.m. and 12:00 midnight of each workday.
- Night shift means authorized work schedules regularly assigned in which at least four (4) hours worked are between the hours of 12:00 midnight and 7:00 a.m. of each workday.
- c. **Day shift** means any authorized work schedules between the hours of 7:00 a.m. and 5:00 p.m.

12.4 Classes during Shift Assignments

If an employee assigned to the swing or graveyard shift wishes to attend classes during the day, the employee shall notify the first non-bargaining supervisor of his or her intention to enroll in classes during the day and the completion date of the classes. The City will review the work program and advise the employee within 14 days whether or not the day classes will interfere with the shift assignments. If no conflict exists, the City will not change the shift assignment until after the semester ends provided that the employee continues to be actively enrolled in the class.

12.5 Rest Period

Each employee shall be entitled to a rest period of fifteen (15) minutes during each one-half ($\frac{1}{2}$) shift worked.

12.6 Daylight Saving Time

- 12.6.1 **Spring**: In the Spring when transitioning to Daylight Saving Time (DST), employees working during the one (1) hour transition from Standard Time to DST will be paid only for actual hours worked. Employees working on a shift which includes the one (1) hour transition may be granted an option by the Department Head or his or her designee, to work an additional hour or use compensatory time, floating holiday, or vacation to make up the lost work hour.
- 12.6.2 **Fall**: In the Fall when transitioning from DST, employees working during the one (1) hour transition will be paid for all hours worked including overtime at one and one-half (1½) times the straight-time rate of pay for hours worked in excess of the regular workweek as set forth in Section 13 (Overtime) of this Agreement.

IBEW Local 1245

SECTION 13: OVERTIME

Employees required to work in excess of their basic work week or in excess of 8 hours in one day (excepting voluntary regular schedules of 9 or 10 hours) during any one week shall be compensated for such overtime services as follows:

13.1 Rate

The overtime rate shall be one and one-half (1½) times the regular hourly salary for the first four (4) hours of such excess and at two (2) times the regular hourly salary for the balance of such excess.

13.2 Department Head Discretion

Whether an employee shall be compensated for overtime by compensatory time off or by payment shall be at the sole discretion of the employee's department head.

13.2.1 **Compensatory Time Off:** Compensatory time off may be earned in lieu of overtime pay at the rate of one and one-half (1½) hours for each overtime hour worked up to a maximum of sixty (60) hours of such compensatory time. Accumulation of compensatory time off in excess of sixty (60) hours may be allowed at the discretion of the department head. Utilization of compensatory time shall be at the discretion of the employee's department head. The times during the calendar year at which an employee shall take his or her compensatory time off shall be determined by the department head with due regard for the wishes of the employee and particular regard for the needs of the service. As used herein, sixty (60) hours is equivalent to ninety (90) hours of time off work.

13.3 Payoff of Overtime

In the event that an employee resigns or is terminated, he or she shall be entitled to compensation for his or her accumulated overtime.

13.4 FLSA Workweek

For the purpose of computing overtime, the work week shall be defined as beginning at 12:01 a.m. Sunday and ending at 12:00 midnight Saturday. Any applied benefited leave shall be included in the calculation of overtime.

13.5 Emergency Overtime

Employees who are called from their living quarters for emergency work or duty on days other than normal workdays, or on normal workdays outside of their regular work hours, shall be paid emergency overtime compensation for actual time worked;

provided, however, that in any case of emergency overtime as herein provided the minimum time for which such overtime compensation shall be paid shall be four (4) hours; and provided further that if such overtime work is performed prior to the beginning of the regularly scheduled work period and such overtime continues into such regularly scheduled work period without a break in service, compensation shall be paid only for the actual time worked.

13.5.1 Overtime Earned Rest Period

If an employee has worked four (4) hours or more of overtime between the hours of 10:30 p.m. and 6:30 a.m. preceding the beginning of his/her regular work hours on a workday, he/she shall be entitled, except in an emergency, to a rest period to include two (2) hours of his/her workday without loss of his/her regular straight-time pay. If the overtime work period extends for six (6) hours or more of overtime between the hours of 10:30 p.m. and 6:30 a.m. preceding the beginning of his/her regular work hours on a workday, he/she shall be entitled, except in an emergency, to a rest period to include five (5) hours of his/her workday without loss of his/her regular straight-time pay. In the event that an employee, due to operational need, is required to work during an earned rest period during regular work hours, the employee will be paid for the earned rest period in addition to wages earned for hours worked.

13.6 Meals

The City will provide meals to employees during emergency assignments and when the employees work continuously for two (2) or more hours immediately before or after their regular shift of eight (8) or nine (9) hours for a total of ten (10) or eleven (11) hours. The employee shall earn an additional meal for every four (4) hours of continuous work after the first meal is earned or taken.

The City will reimburse the employees for the costs of the meal up to twenty five dollars (\$25) if no meals are provided.

13.7 Standby Pay

An employee assigned to standby shall receive ten (10) hours of overtime compensation in addition to any overtime worked while on standby.

13.7.1 **Standby Week:** For the purpose of this Section 13.7.1 (Standby Week), a week shall mean the seven (7) consecutive calendar days following assignment to standby service. In the case of emergencies, the Senior Supervisor or Superintendent may assign another employee to complete the standby week for the originally assigned employee. In that case, the ten (10) hours standby compensation will be pro-rated between the employees.

13.7.2 **Missed Assignment while on Standby:** If an employee misses an overtime assignment while on standby duty, he or she will not receive the standby pay for that day. If he or she misses a second call during the same standby period, he or she will forfeit all the standby pay for that period.

- 13.7.3 **Consequences:** If an employee develops a pattern of failing to perform the service when called, they will be subject to disciplinary action and the Union reserves the right to appeal any action.
- 13.7.4 **Standby Vehicle:** The City will provide the employee on standby duty with a vehicle. Said employee shall be responsible for taking all reasonable steps to insure the safety of the tools and equipment on that vehicle.

13.8 Emergency Overtime Assignments

- All emergency overtime assignments including calling an employee other than the employee on scheduled standby status will be paid the minimum overtime compensation as defined in Section 13.5 (Emergency Overtime). In the event that the employee on scheduled standby is not called, and another employee has been called to perform the emergency overtime assignment, both the employee who performed the assignment and the employee on scheduled standby status will receive the minimum overtime compensation.
- 13.8.1 Scheduled Overtime during Standby: The employee who is on scheduled standby status may perform scheduled overtime tasks on a voluntary basis. The standby premium shall not be pro-rated. If an emergency call occurs during the time that the employee is performing scheduled overtime tasks that call shall be paid at the time and one-half rate and not at the call back rate set forth in Section 13.5 (Emergency Overtime). If other City electrical employees are concurrently performing scheduled overtime tasks, the Senior Electrical Supervisor may, at his or her discretion, assign the work to electrical employees present within the City performing the scheduled overtime tasks at the regular overtime rate of pay. The employee who is on scheduled standby status who chooses to perform scheduled overtime shall defer to other employees who choose schedule overtime assignments pursuant to Section 13.9 (Overtime Offers).

13.9 Overtime Offers

The Department will offer Scheduled Overtime on a voluntary basis to employees with the least amount of overtime worked in accordance with the following provisions:

13.9.1 **Posting of Overtime:** The Department will post a Scheduled Overtime Sign-up List on the Division's bulletin board once every two weeks.

Employees who wish to be considered for scheduled overtime during that time period must sign the list.

- 13.9.2 **Scheduled Overtime Definition:** Scheduled Overtime shall be defined as an overtime assignment scheduled at least 24 hours in advance.
- 13.9.3 **Overtime Report:** The Department will prepare and post a Report of Overtime Worked for employees on a calendar year basis beginning with a zero balance on January 1. The report will be updated after each pay period.
- 13.9.4 **Overtime Assignments:** As scheduled overtime assignments occur, the Department will offer the assignment to individuals on the list for the applicable time period beginning with the person on the sign-up list with the least amount of overtime worked during the calendar year. Notification of scheduled overtime assignments shall be made during work hours at the work place, and employees must be present at the work place to receive the overtime work assignment.
- 13.9.5 **Overtime Assignment Tie Breaker:** If employees have equal amounts of overtime, the scheduled overtime assignment shall be offered to employees in the same order as their names appear on the sign-up list.

SECTION 14: SHIFT DIFFERENTIAL

14.1 Swing Shift

Employees who regularly work a full shift of seven and a half (7½) hours or more on swing shift as defined in Section 12.3.1 (Types of Shifts), which includes more than four (4) hours between the hours of 5:00 p.m. and 12:00 midnight, shall be paid their regular salary plus seven and a half percent (7½%) of their regular monthly salary per month.

14.2 Night Shift

Employees who regularly work a full shift of seven and a half (7½) hours or more on night shift as defined in Section 12.3.1 (Types of Shifts), which includes more than four (4) hours between the hours of 12:00 midnight and 7:00 a.m. (night shift), shall be paid their regular salary plus ten percent (10%) of their regular monthly salary per month; provided, however, that in the case of any such employee who is regularly assigned to such night-shift work for less than an entire work week, the additional payment shall be made only for the portion of the work week worked on the night-shift assignment.

IBEW Local 1245

SECTION 15: NATURAL DISASTER/DECLARED EMERGENCY

If an emergency is formally declared by the City, county, state or national authority:

15.1 Call Outside Normal Working Hours

If an employee is called outside of normal working hours, the employee gets time and one-half (1½) the normal rate of pay for the first whole shift regardless of the number of hours worked. If the employee is not called from home the regular rules apply (i.e., overtime for hours worked above eight in a day).

15.2 Shift Differential

Thereafter, the first eight hours at regular rate and 7½% for hours worked between 5:00 p.m. and midnight; 10% for hours worked from midnight to 7:00 a.m. For hours greater than eight in a shift, the employee gets time and one-half (1½) the normal rate of pay but no shift differential on those hours above eight.

15.3 Reduction in Hours

There will be no reduction in the number of hours in the regular work week schedule.

15.4 Change in Assignments

If an employee is at work during the day when an emergency is declared and is assigned to work a night shift and is sent home, the employee will be compensated as follows:

- 15.4.1 If an employee has been at work for up to but not more than two (2) hours the employee will be paid for two (2) hours at the straight time rate.
- 15.4.2 If an employee has been at work for more than two (2) hours but not more than four (4) hours the employee will be paid for four (4) hours at the straight time rate.
- 15.4.3 If an employee has been at work for more than four (4) hours the employee will be paid for eight (8) hours at the straight time rate.

15.5 Hours Worked in Excess of 12-hours

An employee working more than twelve (12) hours in one continuous shift will be compensated at two (2) times the normal hourly rate.

ARTICLE 3 - LEAVES

SECTION 16: VACATION

16.2 Vacation Approval

The times during the calendar year at which an employee shall take his or her vacation shall be determined by the department head with due regard for the wishes of the employee and particular regard for the needs of the service. Wherever practical, employees working in the same classifications within a division shall be given preference of vacation time by seniority. If the requirements of the service are such that a department head cannot permit an employee within his or her department to take an annual vacation leave or any part of such leave within a particular calendar year, the City Manager may permit such employee to take the deferred vacation during the following year.

16.3 Use of Incremental Vacation

Employees may, with advance supervisory approval, use vacation leave in increments of one hour.

16.4 Accrual Rate as of December 8, 1987

Effective December 8, 1987, the vacation accrual rate shall be modified as follows:

Vacation Accrual Schedule	Authorized Annual Workweeks of Vacation	Vacation Leave Credits in Work Days per Month of Service	Vacation Leave Credits in Hours Earned per Month of Service
Through the first three (3) calendar years of service (except as provided in Section 16.4 (Accrual Rates as of December 8, 1987) and 16.5	2 work weeks	0.833	6.667
Fourth (4 th) through eleventh (11 th) calendar years of service (except as provided in Section 16.5 below)	3 work weeks	1.25	10.00
Twelfth (12 th) through seventeenth (17 th) calendar years (except as provided in Section 16.5	4 work weeks	1.667	13.33
Eighteenth (18th) through twenty-fourth (24th) calendar years (except as provided in Section 16.5	5 work weeks	2.083	16.667
Twenty-fifth (25 th) and subsequent calendar years (except as provided in Section 16.5	6 work weeks	2.5	20.00

The authorized annual vacation leave for prior years is set forth in Exhibit D attached hereto and made a part hereof.

16.5 Accrual, Use, and Limitation for Employees with Less than 6-Months of Service Each employee, during that portion of the calendar year in which he or she is originally appointed and during the next succeeding year, shall be entitled to vacation leave credits at the rate of 0.833 work days for each calendar month of service. Each such employee shall be entitled to take during these two (2) calendar years only such annual vacation leave as he or she earns.

16.6 Vacation Eligibility for Part-time, Intermittent, Leave Without Pay or Reemployees

For an employee who has worked on a part-time or intermittent basis or who has been on leave of absence without pay for a total of six (6) months or more or who has been terminated and subsequently reemployed, the actual years of service with the City shall be used for the purpose of computing length of service in determining eligibility for vacation at the three (3), four (4), five (5) and six (6) weeks' rate.

- 16.6.1 **Prorated Vacation for Intermittent or Part-Time Employees:** Employees working on an intermittent or part-time basis who have worked half-time or more in the preceding calendar year without termination of employment shall be entitled to a prorated vacation leave based upon the actual years of service with the City and upon the actual amount of time worked in the preceding calendar year.
- 16.6.2 **Effect of Military Leave on Vacation Eligibility:** For the purpose of computing length of service in determining eligibility for vacation at the three (3), four (4), five (5) or six (6) weeks' rate, time spent on extended military leave shall be counted as time spent in the service of the City.

16.7 Maximum Vacation Accumulation

Employees may accrue vacation earned up to a maximum cumulative total of 320 hours. Once an employee accrues 320 hours vacation, the employee shall not accrue any additional vacation leave hours until his or her vacation leave balance is below the maximum 320 hours. Under no circumstances will an employee be allowed to accrue more than 320 hours vacation leave. As provided below, the City will advise employees and their supervisors when the employee has attained an accumulation of 280 hours. Supervisors should be flexible in granting vacation requests from employees who reach 280 hours. Provided further that with regard to employees who have reached 280 hours, no vacation request by such employee shall be unreasonably denied. No employee shall be denied vacation leave such that it causes him or her to reach the 320 hour limit.

Notification: To assist employees to remain below the 320 hours vacation leave maximum accumulation, the parties agree that no later than October 1 of each

year of this Agreement, the City will provide the Union and Department Heads with a report identifying accumulated vacation hours of all employees. The City also agrees to notify all employees who have accrued 280 hours of vacation leave, as of this date, that they must take vacation leave to reduce their vacation leave to avoid reaching the 320 hour maximum accrual.

16.8 Prorated Vacation due to Extended Absence, Reemployment or Reinstatement

An employee who has returned from extended military leave or any other extended leave of absence without pay or who has been reemployed or reinstated shall be entitled, during the calendar year in which he or she returns to the City service, to a prorated vacation based upon the total years of service with the City and upon the total number of months of actual service with the City during the said calendar year. For succeeding calendar years, his or her vacation leave shall be as provided elsewhere in this Section 16 (Vacation).

An employee who is granted a leave of absence without pay and who is off the payroll for less than one hundred sixty (160) consecutive hours shall be entitled to a full vacation. If such an employee is off the payroll for one hundred sixty (160) consecutive hours, his or her vacation shall be as provided in the preceding paragraph of this Section 16.8 (Prorated Vacation due to Extended Absence, Reemployment or Reinstatement) or Section 16.9 (Payment of Vacation upon Termination or Extended Leave after Six Months of Service).

16.9 Payment of Vacation upon Termination or Extended Leave after Six Months of Service

If an employee, is terminated, or is granted an extended military leave or other extended leave of absence without pay, such employee, or his or her estate, shall be paid for vacation credits in excess of the actual amount of vacation leave taken or such employee, or his or her estate, shall reimburse the City for the actual amount of vacation leave taken in excess of vacation leave credits, as the case may be.

16.10 Calculation and Payment or Reimbursement of Vacation Credits upon Termination or Extended Leave

Upon termination, extended military leave, or other extended leave of absence without pay, vacation leave credits shall be totaled, and the actual amount of vacation leave taken, including any that may have been taken during the year in which the termination, extended military leave, or other extended leave of absence without pay occurs, shall be deducted from the total credits. If the credits exceed the actual amount of vacation leave taken, such employee, or his or her estate, shall be paid for the excess of credits on the basis hereinafter set forth. If the actual amount of vacation leave taken exceeds the credits, such employee, or his or her estate, shall reimburse the City on the same basis.

The basis for such payment by the City or for such reimbursement to the City shall be as follows:

The employee's hourly pay rate at date of termination, extended military leave, or other extended leave of absence without pay, multiplied by the excess of credits over vacation leave actually taken or excess of vacation leave actually taken over credits, as the case may be.

16.11 Lump Sum Payment of Vacation

Upon termination, extended military leave or other extended leave of absence without pay, payment for excess of vacation leave credits shall be made in one lump sum at time of termination, extended military leave, or other extended leave of absence without pay, or as soon thereafter as possible; provided, however, that an employee may elect to use excess vacation leave credits prior to termination, extended military leave, or other extended leave of absence without pay, to the extent permitted by this Section 16 (Vacation), and receive a lump sum payment for the balance of leave credits, if any. An employee, or his or her estate, shall not be paid for vacation leave credits in excess of eight (8) calendar weeks. Notwithstanding the foregoing, accumulated but unearned vacation credit at the time of retirement shall be paid off in a lump sum.

16.12 Calculation of Vacation Based on Actual Paid Hours

All vacation benefits shall be calculated upon actual paid hours of work. This provision shall go into effect immediately upon implementation of necessary data processing and programming changes.

16.13 Vacation Accounting Procedures

The City may revise vacation accumulation provisions in order to standardize accounting procedures without effect on the amount of employees' vacation, subject to review and approval by the union.

16.14 Voluntary Time Off

The City will implement a voluntary time off (VTO) program, which may include a yearend holiday closure. It is understood by the parties that employee participation is completely voluntary and that employees who wish to work during such periods will be permitted to do so.

SECTION 17: HOLIDAYS

17.1 Holidays

Recognized holidays for employees in Representation Unit C shall be:

Page 123 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

17.1.1	New rear a bay
17.1.2	Martin Luther King's Birthday (3rd Monday of January)
17.1.3	Lincoln's Birthday
17.1.4	Washington's Birthday (observed on the third Monday in February)
17.1.5	Malcolm X's Birthday (observed on Monday or Friday nearest May 19th)
17.1.6	Memorial Day (observed on the last Monday in May)
17.1.7	Juneteenth National Independence Day – observed on the Monday or
	Friday nearest June 19
17.1.8	Independence Day
17.1.9	Labor Day (observed on the first Monday in September)
17.1.10	Indigenous Peoples' Day (observed on the second Monday in October)
17.1.11	Veteran's Day
17.1.12	Thanksgiving Day
17.1.13	Day after Thanksgiving Day
17.1.14	Christmas Day
17.1.15	Effective January 9, 2000, each employee shall be granted three (3)

17.2 Floating Holidays

17 1 1

New Year's Day

Any employee shall be granted three (3) floating holidays each calendar year. The days selected shall be by mutual agreement between the employee and the department head (or his or her designee). In the event mutual agreement cannot be reached on the selection of floating holidays, the employee shall have the three (3) days added to his or her accrued vacation time. Employees may take Floating Holidays in one (1) hour increments.

floating holidays each calendar year.

17.2.1 Additional Floating Holidays

For employees who were required to remain in the workplace from March 17, 2020 - June 1, 2020, the City will provide 8 hours of floating holidays for every 40 hours of regularly scheduled hours worked in the workplace up to a maximum of 32 hours of floating holiday. The City will credit these floating holiday hours in the first full pay period after adoption of the MOU.

City will use a specific pay code for the additional floating holiday, hours will be available June 30, 2021. These additional 32 hours of floating holiday shall have no cash value and may not roll over to vacation if not used by June 30, 2021.

17.3 Floating Holiday Accrual upon Employment

In the first calendar year of employment, an employee hired before May 1 shall be entitled to three (3) floating holidays; and an employee hired on or after May 1, but before September 1, shall be entitled to two (2) floating holidays; and an employee hired

on or after September 1, shall be entitled to one (1) floating holiday in that first calendar year.

17.4 Limitations

No monetary award shall be authorized for unused accumulated Floating Holidays for employees who terminate employment prior to the completion of six (6) months of continuous City service.

17.5 Holidays for Employees whose Work Week is Other than Monday through Friday

Employees whose work week is Monday through Friday shall be allowed all holidays with pay which fall within such work week. Those employees whose work week is other than Monday through Friday shall be entitled to the same number of holidays, with pay, during each calendar year as are allowed to employees whose work week is Monday through Friday. The procedure for allowing holidays for employees whose workweek is other than Monday through Friday shall be established by the City Manager. The provisions of this Section 17.5 (Holidays for Employees whose Work Week is Other than Monday through Friday) are not applicable to intermittent employees.

17.6 Work during a Holiday

An employee required to work on any day which is a holiday for employees whose work week is Monday through Friday shall be paid for the number of hours worked during such day at the rate of one and one-half (1½) times the straight-time rate, based upon his or her regular monthly salary, or shall be granted compensatory time off in an amount equal to one and one-half (1½) times the number of hours worked on such holiday. Any employee who works on Christmas day or Thanksgiving Day shall be paid double time for that day. The hours worked on such holiday and paid at the rate herein provided shall not be credited in computing the hours worked in the week for overtime purposes.

The holiday pay provided for shall be in addition to an employee's regular salary. In the event that the time worked on such a holiday is also overtime, as provided in Section 13 (Overtime) of this Memorandum Agreement, payment will be made for the hours worked either as overtime under said Section 13 (Overtime), or as holiday pay under this Section 17 (Holidays), but will not be made under both Sections.

17.7 Holidays for Part-Time Employees

Regularly scheduled part-time employees 1) who are assigned to a regular schedule of twenty (20) or more hours per week; 2) who have worked for the City as regular or part-time employees for one (1) or more years; and 3) who during such period of time have worked a minimum of 1,000 hours shall be entitled to the holiday pay for the number of hours which such employee would have worked had the holiday not

occurred, provided such holiday occurs on a day which such employee is regularly scheduled to work.

SECTION 18: SICK LEAVE

18.1 Eligibility

Any employee shall be entitled to take sick leave with full pay in case of sickness, disability, or serious illness within the immediate family of the employee in accordance with the provisions of Sections 18.2 (Sick Leave Accrual) to 18.12 (Cessation of Accrual).

18.2 Sick Leave Accrual

Each employee shall be credited with one (1) working day of sick leave with full pay for each month of service.

For the purposes of this Section 18 (Sick Leave), a month of service shall mean thirty (30) consecutive calendar days in the case of employees working on a fulltime or part-time basis, and shall mean 163 hours of work in the case of employees working on an intermittent basis.

18.3 Pro-rated Sick Leave for Part-Time Employees

An employee working on a part-time basis shall be entitled to use earned sick leave only on a pro rata basis; for example if an employee works half-time, he or she shall be paid for time off on sick leave on a half-time basis.

18.4 Sick Leave for Intermittent Employees

An employee who works on an intermittent basis shall be entitled to use earned sick leave only for those days on which he or she would have worked if he or she had not been sick; provided, however, that an employee working on an intermittent basis who works only when called shall be entitled to use earned sick leave only when he or she becomes sick after reporting to work in response to such call.

18.5 Maximum Accumulation

Such sick leave as provided in Section 18.2 (Sick Leave Accrual), when not used, shall be cumulative; but the accumulated unused period of sick leave shall not exceed two hundred (200) working days, regardless of the length of service. When the maximum of two hundred (200) working days has been reached, and thereafter part of said maximum has been used, the used part of said maximum may subsequently be replenished at the applicable rate provided in Section 18.2 (Sick Leave Accrual).

18.6 Payment upon Retirement/Termination

All accumulated sick leave shall be canceled when an employee terminates or is terminated, except as provided below for employees hired on or before June 30, 2013.

- 18.6.1 Employees Hired on or Before June 30, 2013: For employees hired on or before June 30, 2013 who retire or voluntarily terminate with a vested pension, and with between twenty (20) years and twenty-eight (28) years of benefited service shall be entitled to receive payment in an amount equal to thirty eight percent (38%) of their accrued sick leave days up to a maximum of the two hundred (200) day maximum accumulation. Provided further that any employee retiring on permanent disability arising out of and incurred in the course and scope of his employment with the City shall be entitled to receive payment at retirement for thirty eight percent (38%) of accumulated unused sick leave days but not, in any event, more than thirty eight percent (38%) of the two hundred (200) day maximum accumulation.
- 18.6.2 Payment After 28 Years of Service for Employees Hired on or Before June 30, 2013: Employees who were hired on or before June 30, 2013, who terminate with at least twenty-eight (28) years of benefited City of Berkeley service or employees retiring on permanent disability arising out of and incurred in the course and scope of their employment with the City with at least twenty-eight (28) years of benefited service shall be entitled to receive payment in an amount equal to fifty percent (50%) of their accrued sick leave days up to a maximum of two hundred (200) unused sick leave days.

18.6.3 Sick Leave Payout to 401(a) Plan for Employees Hired on or Before June 30, 2013: The City and the Union have met and conferred on an Internal Revenue Code Section 401(a) plan and trust agreement to address the liquidation of sick leave at time of retirement. This plan and trust agreement was originally negotiated with the Berkeley Fire Fighters Association/I.A.F.F. Local 1227 and has been submitted to the Internal Revenue Service for a Determination Letter and a Private Letter Ruling which are pending. If the City receives a positive response from the Internal Revenue Service, the plan and trust agreement will be extended to the employees in the bargaining unit. This will provide the employees with an irrevocable option to defer accrued but unused sick leave at time of retirement into a 401(a) plan or be paid out the balance of the accrued but unused sick leave less withholding of applicable federal and state taxes.

18.7 Annual Payment for Hours in Excess of Maximum for Employees Working Half-Time or More for Employees Hired on or Before June 30, 2013

Employees who regularly work one-half (½) time or more and who have attained the one hundred fifty (150) day maximum sick leave accumulation shall be entitled to receive payment for one-third (1/3) of the first twelve (12) days of sick leave for which they become eligible but do not use and would otherwise forfeit because of the one hundred fifty (150) day maximum limitation. Determination of eligibility for such payment shall be made on an annual calendar year basis, and payment for such sick leave for any calendar year shall be made not later than January 22 of the following year. Such payment shall be made at the employee's salary rate in effect on the preceding December 31 and shall be made only in units of whole days and not for any fraction of a day.

18.8 Restoration of Sick Leave if Reemployed within Two Years

Accumulated sick leave, which has been canceled by reason of an employee's termination, shall be credited back to such employee if the employee returns to City employment within two (2) years of such termination.

18.9 Sick Leave Use/Family Illness

Sick leave shall not be considered as a privilege which an employee may use at his or her discretion but shall be allowed only in case of his or her sickness or disability or in the case of serious illness within the immediate family of the employee. Not more than fifteen (15) working days in any calendar year may be taken as sick leave because of the illness of a member of the employee's immediate family, except for serious medical conditions covered under the provisions of Administrative Regulation 2.4 (Family Care Leave). The immediate family of an employee, for the purpose of this Section, shall be defined as: spouse, domestic partner, son, daughter, parent or dependent.

18.10 Prohibition for Use with Outside Employment

No sick leave shall be allowed for time off for an injury incurred while working for another employer, provided that such injury is covered by the Workers' Compensation laws of the State of California, or other provision for payment for time off because of such injury is made by such other employer. In the event such injury is not covered by the Workers' Compensation laws of the State of California and no other provision for payment for time off because of injury is made by such other employer, sick leave in accordance with the provisions of this Section shall be allowed only if such outside employment has been approved by the City.

18.11 Notification Requirement

In order to receive compensation while absent on sick leave, the employee shall notify his department head prior to or within four (4) hours after the time set for beginning his or her daily duties, or as may be approved by the head of his or her department. The Union and the City recognize it is advantageous to both parties that calls should normally be made prior to the beginning of the work shift. Leave for non-

emergency doctors' appointments shall be requested in advance. In specific instances the City may by written notice require an employee (s) to call in prior to the beginning of their shift in order to be eligible for sick leave.

18.12 Cessation of Accrual

An employee who is granted a leave of absence without pay and who is off the payroll for less than one hundred-sixty consecutive hours shall receive his or her earned sick leave credit. If the employee is off the payroll for one hundred-sixty (160) consecutive hours or more, he or she shall not earn sick leave credit for each two successive pay periods that he or she is off the payroll.

18.13 Workers' Compensation

- All probationary and permanent employees of the City and provisional employees who have worked a total of four (4) months or more for the City, who have suffered any disability arising out of and in the course of their employment as defined by the Worker's Compensation Insurance and Safety Act of the State of California, and who are receiving or shall receive compensation from the insurance carrier for such disability, and during the first seven (7) days after such disability when compensation is not paid by the insurance carrier shall be entitled to remain absent from duty with pay until such time as they are able to return to duty or some other final disposition is made of their case; provided, however, that provisional employees who have worked a total of four (4) months or more for the City shall be entitled to remain absent from duty with pay for the period they would have been permitted to remain provisional employees under their employment at the time of injury.
- All employees, other than probationary and permanent employees and provisional employees who have worked a total of four (4) months or more, shall be entitled to such compensation as may be allowed them by the Workers' Compensation Insurance and Safety Act of the State of California.
 - 18.13.1 Workers' Compensation Payments: Payments from the insurance carrier for disability arising out of and in the course of employment shall be paid to the employee. The amount of such payment or payments shall be deducted from the monies, which the employee received from the City. Payments from the insurance carrier, plus the monies paid to the employee by the City, shall be equivalent to the employee's regular full pay.
 - 18.13.2 Workers' Compensation Leave and Salary Continuation: Payments under the Workers' Compensation law for temporary disability, or a recurrence thereof, arising out of and in the course of employment shall be paid for a period not to exceed 365 days at a maximum payment of the employee's pre-disability net pay but shall not exclude any salary

adjustments to which the employee is entitled. Thereafter, the employee will continue to receive only the temporary disability payments provided under state law, and the City will cease to pay the difference. However, salary continuation payments above the statutorily required temporary disability payments shall not be reported by the City to CalPERS as compensation. No time worked shall be part of the 365 calendar days of salary supplement paid by the City. This change shall not affect employees who are currently off the job with a work-related injury which occurred prior to July 1, 1979.

- 18.13.3 **Calculation:** The City shall continue to calculate salary continuation at pre-disability gross pay. The City may calculate salary continuation payments at pre-disability net pay at such time when they develop the capacity to administer it equitably. Any change in calculation shall not reduce employees' combination of disability payments and salary continuation payments below employees' pre-disability net pay.
 - The change in calculation shall not affect employees who are off the job with a work-related injury prior to the new calculation method being implemented.
- 18.13.4 Workers' Compensation Related Absence of Less than Four (4) Hours: An employee who is absent from work for a medical appointment or physical therapy for less than four (4) hours will have ½ day charged against his or her 365 day maximum period for receipt of salary continuation pre-disability net pay.
- 18.13.5 Workers' Compensation Related Absence of Four (4) Hours or More:
 An employee who is absent from work for a medical appointment or physical therapy for 4 hours or more will have 1 day charged against his or her 365 day maximum period for receipt of salary continuation predisability net pay.
- 18.13.6 **Calculation Based on Actual Paid Hours:** All sick leave benefits shall be calculated upon actual paid hours.

18.14 Light Duty

Assignments for temporarily disabled employees.

An employee who is absent by reason of industrial disability may be returned to work and given temporary light duties within his or her ability to perform. The duration of any such period of temporary work should be determined by the City. Employees should be compensated at the rate of pay of their regular classification while

engaged in such temporary duties, and such work assignments are to incorporate the following provisions:

- 18.14.1 **Modified Duty Accommodation:** The City shall accommodate, when feasible, employees covered by this memorandum under the provisions of workers' compensation, and such work assignments are to incorporate the following provision:
 - 18.14.1.1 The assignment shall be consistent with medical limitations as determined by the physician of record.
 - 18.14.1.2 The Assignment shall be within the City of Berkeley and may include hours and days of work other than the employee's regular assignment.
- 18.14.2 Modified Duty Accommodation for Non-Industrial Disabilities: The City may accommodate an employee disabled with a non-industrial disability by providing a modified work assignment in that employee's classification. To be eligible for such a modified assignment, the employee must provide the Human Resources Department with a medical statement from his or her treating physician that clearly states the medical limitations and abilities of the employee. If modification of that position does not serve the best interests of the City, other classifications may be considered, subject to the approval of the Human Resources Director. Compensation will be provided at the level of the classification in which the temporarily disabled employee works during the disability. The employee must meet standards of satisfactory performance for the duration of the work assignment.
- 18.14.3 Modified Duty for Pregnancy-Related Disability: In the case of a medically certified, pregnancy-related disability, in which the normal duties clearly threaten the health and safety of the employee or the unborn child, the Human Resources Department will endeavor to place the employee in a position which best serves the interest of the City with no loss of pay, but in no event will such placement exceed 5 months in duration.
- 18.14.4 **Light Duty Assignments**: Availability of light duty job assignments shall be discussed upon request at the monthly Labor-Management Meetings.

18.15 Control Program for Sick Leave Use

The City may establish a reasonable program for the control of abuse of sick leave and absenteeism, subject to Union review and comment.

IBEW Local 1245

18.16 Bonus Time for Unused Sick Leave

For every six (6) months of perfect sick leave attendance the employee will receive eight (8) hours of bonus time. A leave of absence from work pursuant to workers' compensation is counted as an absence from work in the same manner as sick leave for the purpose of this bonus. This bonus time will be prorated for part-time employees. Such bonus time can be used for any leave purpose covered by this Memorandum Agreement. Such bonus time shall be counted as vacation leave credits for purposes of determining eligibility for carry-over and cash payment.

18.17 State Disability Insurance

Except as provided in 18.17.1.3 below, any employee who is absent due to personal illness for more than 7 calendar days (or for any period of time if hospitalized) may apply for State Disability Insurance Benefits.

- 18.17.1 **State Disability Integration:** After such employee has been absent from work due to personal illness for six (6) consecutive work days, if the employee applies and is eligible for State Disability Insurance, the City shall integrate the employee's pay with the employee's State Disability benefits in the following way:
 - 18.17.1.1 The City will determine the weekly SDI benefit amount based on the amount of wages earned with the City of Berkeley in the SDI base period.
 - 18.17.1.2 The weekly SDI benefit will be subtracted from the employee's normal weekly wages and the amount necessary to bring the total of State Disability plus wages to 100% will be deducted from any accumulated sick leave, vacation leave and compensatory time available to the employee. The integration with vacation leave and compensatory time is optional but will be automatically implemented after sick leave has expired unless written notification is received from the employee, as discussed below.
 - 18.17.1.3 The employee must notify the payroll clerk, in writing, to stop integration of State Disability Insurance payments with vacation leave or compensatory time. Upon receipt of notification, the payroll clerk will cease integration of any future leave for that incident of illness.
 - 18.17.1.4 The employee must show the State of California form
 (Disability Insurance Notice of Computation) to his or her payroll clerk to verify dates covered by SDI and the amount to

City of Berkeley

IBEW Local 1245

be paid. The employee must inform their payroll clerk of all SDI payments. Any employee entitled to State Disability Insurance shall receive in addition thereto such portion of his or her accumulated leave as will meet, but not exceed, the standard earnings of the employee for his or her normal workweek, up to a maximum of five (5) days.

SECTION 19: FUNERAL LEAVE

In the case of death within the immediate family of an employee (who is not excluded from such benefit pursuant to Chapter 4.04.120 of the Berkeley Municipal Code (Personnel Ordinance) which is attached hereto as Exhibit C (City of Berkeley Municipal Code Chapter 4.04.120) and made a part hereof, such employee shall be entitled to remain absent from duty with pay in order to attend the funeral or memorial service for a period not exceeding three (3) working days or, in the case of a funeral or memorial service conducted out of the State of California, for a period not exceeding five (5) working days. For the purpose of this Section, the immediate family of an employee shall be defined as: husband, wife, domestic partner, mother, father, sister, brother, child, grandmother, grandfather, grandchildren, mother-in-law, father-in-law, brother-in-law, sister-in-law, son-inlaw, and daughter-in-law, step-parent, step-sibling, step-child, aunts and uncles, nieces and nephews.

Leave of absence with pay because of death in an employee's immediate family is allowed for the purpose of attending the funeral or memorial services, and such leave shall not be charged against vacation or sick leave that an employee may be entitled to but shall be in addition thereto. Employees may request, and the City will make reasonable efforts to accommodate requests, for employees to supplement bereavement leave by using accrued vacation, compensatory time, or floating holiday. All accrued leave (and/or sick leave, if applicable) shall be utilized prior to taking a leave of absence without pay.

Funeral Leave for Persons Other than Immediate Family: In special cases, with the approval of the department head, the City Manager may grant a death leave to allow an employee to attend funeral or memorial services because of a death of a person not included within the definition of the immediate family.

SECTION 20: MILITARY AND MARITIME LEAVE

Page 133 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

Military and Maritime Leave shall be governed by the provisions of the Federal Uniformed Services Employment and Reemployment Rights Act (USERRA) and any regulations promulgated to implement the Act and the California Military and Veteran's Code.

If an employee voluntarily extends his or her military leave in excess of two (2) weeks, the amount of pay received by the employee for his or her military duty shall be deducted from his or her regular pay for such period.

SECTION 21: PARENTAL LEAVE

21.1 Eligibility:

Employees with one (1) or more years of employment with the City (or equivalent in the case of part-time employees) shall be entitled to parental leave as follows:

- A continuous parental leave of up to one year will be granted to any employee with one year (full-time) or more years of employment with the City (or equivalent in the case of part-time employees) upon the birth of a child or the legal adoption of a child who is five years or younger, provided that:
 - 21.1.1 **Commencement of Parental Leave:** The one year parental leave must commence no later than 13 months from the date of birth or adoption and must expire no later than 25 months from the date of birth or adoption, and.
 - 21.1.2 **Notice Requirement:** Employees exercising their rights under this provision must provide the City at least 45 calendar days notice prior to the anticipated commencement date of the parental leave, unless a shorter notice is approved for good cause.
 - 21.1.3 **Use of Sick Leave:** The employee, at his or her option, may request that all or any portion of sick leave (up to a maximum of two hundred (200) days) or vacation leave that he or she has accumulated be paid in the same manner as it would if he or she had been absent due to illness or on vacation during the leave. In the event both parents are employed by the City, nothing in the Personnel Rules and Regulations shall prohibit both employees from taking simultaneous parental leave.
 - 21.1.4 **Verification Requirement:** The foregoing leave shall be granted upon medical certification of pregnancy or the presentation of legal evidence of adoption.
 - 21.1.5 Life and Health Insurance Coverage Upon Exhaustion of Leave:

City of Berkeley

IBEW Local 1245

During approved parental leave, after all earned leaves are exhausted, (except sick leave) the City agrees to maintain life and health insurance coverage for duration of the approved leave subject to any regular participation requirement of the employee. Thereafter the City agrees to continue coverage for the employee at the employee's expense.

SECTION 22: LEAVE OF ABSENCE WITHOUT PAY

22.1 Department Head Authority

Upon the request of the employee, a department head may grant a leave of absence to an employee within his or her department without pay for a period not to exceed fifteen (15) working days. No leave without pay shall be granted for more than fifteen (15) working days except upon the written request of an employee and approval of the City Manager. Failure on the part of an employee on leave to report promptly at its expiration shall be cause for discharge.

22.2 Union Sponsored Training

A leave of absence without pay shall be granted at the request of an employee and the Union for the purpose of the employee's attending a training course sponsored by the Union. The maximum duration of such leave shall not exceed two (2) consecutive payroll periods in a calendar year.

22.3 Exhaustion of Accrued Time

Leaves of absence without pay shall be granted only after the employee has utilized all accrued vacation and any other time owed to the employee, except sick leave.

SECTION 23: JURY DUTY LEAVE

An employee who is called or required to serve as a trial juror shall be entitled to be absent from work with pay during the period of jury service or while required to be present in court as a result of a call to jury duty. An employee is required to be present at work when not serving as a trial juror or as a member of a jury selection panel. An employee will notify his or her supervisor of any unusual constraints (e.g., time to call in, time to report for jury service) made by the court that affect the employee's ability to simultaneously fulfill his or her jury duty service and employment obligations; and the supervisor will attempt to accommodate the employee based on the operational needs of the department. Absence from work to perform jury duty service shall apply to employees who work swing and graveyard shifts for those days on jury duty. An employee who serves jury duty on his or her day off shall be granted an equivalent number of days off during his or her normal workweek. Employees are required to submit a written proof of jury duty service issued

Page 135 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

by the court in order to receive payment for Jury Duty Leave. The employee will keep any payment received for jury service including mileage reimbursement.

23.1 Court Time

The City will guarantee a minimum of three (3) hours pay for every court appearance required by an employee in the conduct of official City of Berkeley job duties on the employee's scheduled day off and three (3) hours minimum if on a workday but outside scheduled working hours. In addition, employees assigned to court phone standby in the conduct of official City of Berkeley job duties will be compensated by earning compensatory time as follows: Duty day, outside of scheduled working hours, one hour minimum compensatory time and hour for hour thereafter. Day off, two-hour minimum compensatory time and hour for hour thereafter.

IBEW Local 1245

ARTICLE 4 - HEALTH AND WELFARE BENEFITS

SECTION 24: HOSPITAL-MEDICAL, DENTAL, AND VISION COVERAGE

24.1 Medical Coverage

The City shall pay the cost of health insurance coverage for employees who are not excluded from such benefit pursuant to Chapter 4.04.120 of the Berkeley Municipal Code (Personnel Ordinance) which is attached hereto as Exhibit C (City of Berkeley Municipal Code Chapter 4.04.120) and made a part hereof, and who have such coverage under any group health insurance plan authorized by the City Council. The present level of benefits under the Kaiser Plan shall be maintained at City expense for the duration of this Memorandum Agreement. Medical benefits are extended to full-time employees, spouse of the employee or domestic partner, and IRS dependent up to age 26.

The City is committed to providing at least one fully paid employer provided health insurance option for employees and eligible dependents. The Union acknowledges the City's policies as stated above.

24.2 Maximum Medical Premium Payments

- 24.2.1 Effective January 1, 2003, the City will transfer employees who chose to maintain Kaiser medical plan coverage from the Kaiser V-5 Plan into the Kaiser S-1 Plan. Effective January 1, 2003, the City will transfer employees who chose to maintain Health Net HMO coverage from the Health Net W-2 Plan to the Health net C9A Plan.
- 24.2.2 For employees hired on or after January 1, 2003, the maximum amount the City shall be required to pay for medical insurance premiums shall be the applicable Kaiser rate (i.e., single party, two-party, or family) regardless of the City sponsored health plan selected by the employee.
- 24.2.3 For employees in a probationary or career benefited status as of January 1, 2003, the City will continue to pay 100% of the health care premium costs (employee and any dependents) for the health plan the employee is enrolled in as of this date as long as the employee remains employed in the bargaining unit. After January 1, 2003, if the employee transfers health coverage to a different health plan, the employee will assume responsibility for paying the difference, if any, between the Kaiser monthly premium rate (i.e., single-party, two-party, or family) and the plan chosen by the employee from that date forward.

- 24.2.4 Effective January 1, 2009, the amount the City contributes each calendar year toward the payment of health insurance premiums, in accordance with Section 24.2 (Maximum Medical Premium Payment), will increase by the lesser of twenty percent (20%) (single, two-party, family) or the amount of the Kaiser HMO premium amount (single, twoparty, family) in effect on that date.
- 24.2.5 Effective January 1, 2014, for those employees who are enrolled on October 31, 2012 in the Health Net Point of Service (POS) health plan (Payroll Benefit Code Description HJ, HK, and HL) the City shall pay medical insurance premiums at the applicable rate for the Health Net HMO plan (i.e., single; 2-party, or family). If the employee chooses to remain enrolled in the Health Net POS Plan on or after January 1, 2014, the employee will assume responsibility for paying the difference between the Health Net POS and the Health Net HMO monthly premium rate (single; 2-party, or family). Provided further that if at any time after January 1, 2014 the employee transfers health coverage from Health Net POS to the Kaiser health plan, the maximum amount the City shall be required to pay for medical insurance premiums shall be the applicable Kaiser rate (i.e., single 2-party; or family) even if the employee subsequently enrolls in a different City sponsored health plan.
- 24.2.6 Meet & Confer: The Parties agree to meet and confer with the City, commencing no sooner than January 1, 2017. The negotiations shall be on developing comparable and less expensive health plan options as a means of reducing or ensuring that the City shall not be required to pay any penalties associated with the Excise Tax. This meet and confer process will be subject to normal rules of collective bargaining, including applicable impasse, strike or lock-out procedures.
- Any changes resulting from a review of employee health insurance or other health related benefits will be subject to meet and confer between the Union and the City, except as otherwise provided in Section 24.5 (Change in Insurance Carrier).

24.3 Dental Coverage

The City shall provide a dental care program for employees who are not excluded from such benefit by Chapter 4.04.120 of the Berkeley Municipal Code (Personnel Ordinance) which is attached hereto as Exhibit C (City of Berkeley Municipal Code Chapter 4.04.120) and made a part hereof. The present level of benefits under the Dental Program shall be maintained at City expense for the duration of this Memorandum. Dental benefits are extended to full-time employees, spouse of the employee or domestic partner, and IRS dependents up to age 26.

- 24.3.1 Effective January 1, 2000, the dental plan is improved from 90% coverage to 100% coverage.
- 24.3.2 Effective January 1, 2003, the annual maximum dental benefit will be increased to \$4000 per year, the lifetime maximum orthodontia benefit will be increased to \$4000, and the annual cleanings will be increased from two (2) to four (4). Effective January 1, 2016, the number of annual cleanings will be decreased from four (4) to three (3).

24.4 Vision Coverage

Effective January 1, 2003, the City shall provide a Vision Care Program for employees covered by this Agreement. The annual maximums for this benefit are as follows:

Benefit	Benefit Frequency	
Exam	12-Months	
Lenses	12-Months	
Frames	24-Months	
Contact Lenses**	12-Months	
**Note: Benefits for Contact Lenses are in lieu of benefits for lenses and		
frames.		

- The maximum amount the City shall be required to pay for the Vision Care Program shall be the applicable Vision Services Plan (\$25 Plan B) rate (i.e., employee only, employee plus spouse, employee plus one (1) child; employee plus family).
- If during the term of this Agreement the premiums for such Vision Care Program are increased, the amount the City contributes shall increase no more than five percent (5%) above the previous calendar year's contribution amount towards the payment of the monthly premium.
- Recognizing that the "vision exam" portion of the Vision Care Program duplicates existing benefit provided under Medical Coverage Section 24.1 (Medical Coverage) and 24.2 (Maximum Medical Premium Payments), the Union and the City agree to revisit the Vision Care Program during the term of this Agreement with the expectation that it may be possible to identify a plan that provides a "Hardware Only" benefit (Lenses every 12-months; Frames every 24-months). Such "Hardware Only" plan shall be at a lower City and employee cost than the present Vision Services Plan (\$25 Plan B).

24.5 Change in Insurance Carrier

Before the City acts to change an insurance carrier during the term of this Agreement, the City shall give the Union thirty (30) days notice of its intention to change carriers and shall, upon written request, meet with the Union to discuss the reasons for

such change. The final determination of insurance carriers shall be at the sole discretion of the City.

24.6 Domestic Partner Coverage

If an employee chooses to complete and submit an Affidavit of Domestic partnership and sign up for medical and/or dental benefits for his or her domestic partner, the employee shall be subject to federal and state income tax withholding.

24.7 Part-Time Employees and Prorated Benefits

All career and grant-funded, benefited employees working less than full time shall receive prorated rather than full fringe benefits and shall pay, by payroll deduction, a prorate portion of the health and dental insurance premiums.

24.8 Commencement of Medical, Dental, and Vision Benefits

Medical, dental, and vision benefits shall begin the first day of the calendar month following the date of hire, and end the last day of the month an employee is in pay status.

24.9 Health Insurance In-Lieu Payments

Effective January 1, 2019, for employees who show proof of alternate medical coverage, the City will compensate the employee \$576.00 per month, prorated for less than full-time employees, as provided in Section 24.7 (Part-Time Employees and Prorated Benefits). In order to include in-lieu payments in the regular rate of pay for the correct overtime calculation, the City must provide in-lieu payments on the bi-weekly payroll cycle. This benefit shall be frozen at this amount for the term of this agreement.

24.10 Group Life Insurance

Effective January 1, 2016, the City shall provide group life insurance, by a carrier of the City's choice, in the amount of \$100,000 which shall include a standard accidental death and dismemberment provision of a like amount. In addition, the employee may purchase additional life insurance up to a maximum of \$300,000 at a rate offered by the City's insurance carrier and subject to any medical exam as required by the insurance carrier. Life insurance shall become effective the first day of the calendar month following the date of hire, and shall continue until the last day of the calendar month in a pay status.

24.11 Pre-Tax Status

When employees are required to contribute to the cost for medical, dental, or vision insurance, those contributions will continue to qualify as pre-tax expenses under the provisions of IRS Section 125.

IBEW Local 1245

SECTION 25: RETIREE MEDICAL COVERAGE

The City and Union have agreed that the City will make available retiree health insurance coverage under certain terms and conditions described below. The retiree medical benefit described below is the plan tentatively agreed to during multi-union bargaining during the summer of 1998. The terms and conditions of this benefit shall be set forth in a separate document which shall contain a full plan description and shall control the administration of the retiree medical plan.

The City will begin to provide the retiree medical coverage set forth in this Section on July 1, 1998. An employee's entitlement to any and all benefits provided by the City under this retiree medical cover plan are subject to the funding limitations set forth in sub-Section 25.8 (City Funding of Retiree Health Benefit).

25.1 Amendment of Retiree Health Premium Assistance Plan III, effective January 23, 1998, Restated and Amended effective March 22, 2011

Employees who retire on or after June 21, 2015, shall be permitted, at their discretion, to enroll in non-City sponsored health plans. After Council approval of the successor Memorandum of Understanding, the City shall amend the Retiree Health Premium Assistance Plan III as soon as practicable to allow enrollment in non-City sponsored health plans. In the event a retiree elects to enroll in a nonCity sponsored health plan, the City shall make medical insurance premium payments directly to the health insurance provider in an amount equal to what the City would contribute to the City sponsored health plan. Retiree shall be solely responsible for all aspects of the requirements to enroll in a non-City sponsored health plan and maintain eligibility for such a plan; the City's sole obligation is to pay the medical insurance premium contribution required under this Section, as directed by the retiree to a non-City sponsored health plan. The City shall not be responsible for any excess cost differentials associated with the direct payment of premiums to non-City sponsored plans. The City will only make payments through its third party administrator to provide medical insurance premium payments for an individual plan and will not make payments for a group plan. The retiree and/or surviving spouse or domestic partner that enroll in non-City sponsored health plans shall be solely responsible for paying the administrative set up fee, the monthly administrative fee, and/or any other fees established by the third party administrator, and said fees will be deducted directly from the retiree's monthly contribution. No cash payments will be paid directly to the retiree and/or the retiree's spouse/domestic partner. There shall be no cash in lieu payments made under this benefit.

The City and the Union agree that the City will also amend the Retiree Premium Assistance Plan III to allow eligible retirees who retired prior to June 21, 2015 to enroll in a non-City sponsored health plan.

25.2 Eligibility

An employee is eligible for the retiree health insurance coverage set forth in subSection 26.2 (Provisional Appointments to a Higher Classification) below if he or she meets all the following criteria:

- 25.2.1 retires on or after July 1, 1998,
- 25.2.2 is vested with CalPERS,
- 25.2.3 has at least eight (8) years of CalPERS qualifying service with the City,
- 25.2.4 is at least age 55.

25.3 Pre Age 65 Retiree Health Insurance

25.3.1 **Beginning July 1, 1998**: The City shall make available health insurance coverage to the employee and his or her spouse or domestic partner. The City will pay on the employee's behalf no more than \$166.26 per month for an employee electing single party health coverage and no more than \$332.52 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the following chart:

Years of CalPERS Qualifying Service	Percent of City Contribution
8	30%
9	40%
10	50%
11	58%
12	66%
11	58%
12	66%
13	74%
14	82%
15	90%
16	92%
17	94%
18	96%
19	98%
20	100%

The employee will pay the difference between the City's monthly contribution and the actual monthly insurance premium charged by the health plan he or she has elected for retiree medical coverage. If during the term of this Agreement, the premiums for such health insurance are increased, the

IBEW Local 1245

amount the City contributes shall increase no more than 4.5% above the previous year's contribution. No increases in the amount the City contributes shall occur before July 1, 1999. Thereafter, any increase in the amount contributed by the City will occur on July 1 each year thereafter.

- 25.3.2 **Effective June 28, 2009**: For employees who retire on or after June 28, 2009, the City will pay on the employee's behalf no more than \$358.19 per month for an employee electing single party health coverage and no more than \$716.38 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.3 **Effective June 27, 2010**: For employees who retire on or after June 27, 2010, the City will pay on the employee's behalf no more than \$424.31 per month for an employee electing single party health coverage and no more than \$848.61 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.4 **Effective June 26, 2011**: For employees who retire on or after June 26, 2011, the City will pay on the employee's behalf no more than \$468.40 per month for an employee electing single party health coverage and no more than \$936.80 per month for an employee electing two party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.5 **Effective June 24, 2012**: For employees who retire on or after June 24, 2012, the City will pay on the employee's behalf no more than \$589.48 per month for an employee electing single party health coverage and no more than \$1,178.96 per month for an employee electing two-party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.
- 25.3.6 **Effective June 23, 2013**: For employees who retire on or after June 23, 2013, the City will pay on the employee's behalf no more than \$666.00 per month for an employee electing single party health coverage and no more than \$1,332.01 per month for an employee electing two-party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.

25.3.7 **Effective June 22, 2014**: For employees who retire on or after June 22, 2014, the City will pay on the employee's behalf no more than \$720.97 per month for an employee electing single party health coverage and no more than \$1441.94 per month for an employee electing two-party coverage. The actual monthly amount of money the City will contribute on the employee's behalf will be based on the employee's total years of CalPERS service as provided in the above-referenced chart.

25.3.7.1 Retiree Health Premium Assistance Plan Benefit as of June 30, 2017: As of June 30, 2017, the maximum amount the City will pay the health care service provider, which were increased in accordance with Section 25.3 (Pre Age 65 Retiree Health Insurance), total \$1,517.40 for two party coverage for the retiree and spouse or domestic partner or \$758.70 for single party coverage.

25.4 Retiree Benefits for Employees Age 65 and Over

Once an employee or retiree reaches age 65, he or she is eligible for Medicare. As a result his or her eligibility for the retiree medical benefits set forth in subsection 25.3 (Pre Age 65 Retiree Health Insurance) ceases. On reaching age 65, the City will make available health insurance coverage in addition to Medicare. When an employee or retiree reaches age 65, the City will contribute no more than \$50.00 per month on the employee's behalf for single party health insurance coverage and no more than \$100.00 per month for two party health coverage. If during the term of this Agreement, the premiums for such health insurance are increased, the amount the City contributes shall increase no more than 4.5% above the previous year's contribution.

The City will take such actions under the provisions of Section 218(g) of the Social Security Act to permit employees who are not currently paying employee portion of the Medicare Tax with a one-time opportunity to choose to be covered by the Medicare Tax. If the employee chooses to be covered by the Medicare Tax the choice cannot be revoked at a later date.

The City shall include in its next actuarial request, the impact on the City if the City were to amend its Retiree Health Premium Assistance Plan III to provide the following benefit levels:

Single Party Health: \$200.00 Two Party Health: \$400.00

25.4.1 Retiree Health Premium Assistance Plan Benefit as of June 30, 2017: As of June 30, 2017, the maximum amount the City will pay the health care service provider, which were increased in accordance with Section 25.4

(Retiree Benefits for Employees Age 65 and Over), total \$62.59 for two party coverage for the retiree and spouse or domestic partner or \$31.30 for single party coverage.

25.5 Termination by City of Retiree Medical Benefit

Failure of the retiree or surviving spouse to pay their monthly share of the health insurance premium will result in termination of the retiree medical benefit and relieve the City of any further obligation to provide any further benefits under this Section.

25.6 Retiree Medical Benefit for Employees Retiring Between the Ages of 50 and 55

An employee who is at least 50 years of age, but less than 55, and has at least eight years of CalPERS qualifying employment with the City will retain eligibility for the retiree medical benefits provided in sub-Section 25.3 (Pre Age 65 Retiree Health Insurance) when the employee reaches age 55 if the employee is enrolled in a group health plan coverage from the date of his or her termination from City employment until the employee's 55th birthday. If for any reason the employee has a lapse in group health care coverage the employee forfeits his or her eligibility for the retiree health plan benefits upon reaching age 55 and the City has no further obligation to provide any benefits under this Section to the employee and/or his spouse or domestic partner.

25.7 Employees Retiring with a CalPERS Approved Disability Retirement

If an employee retires from the City before age 55 with a CalPERS approved disability retirement, the employee will retain eligibility for the retiree medical benefits provided in sub-Section 25.3 (Pre Age 65 Retiree Health Insurance) when the employee reaches age 55 if the employee is enrolled in a group health plan coverage from the date of his or her termination from City employment until the employee's 55th birthday. If for any reason the employee has a lapse in health care coverage the employee forfeits his or her eligibility for the retiree health plan benefits upon reaching age 55 and the City has no further obligation to provide any benefits under this Section to the employee and/or his spouse or domestic partner.

25.8 City Funding of Retiree Health Benefit

City contributions to the retiree medical benefit will begin on July 1, 1998. Funding of this benefit will be set aside in a trust to be established by the City.

The retiree medical benefit will be funded by a charge of 0.25% of payroll in each year of this Agreement, so that contributions are at 1% of the payroll in the fourth year of the Agreement. The City will fund the benefit at approximately 1% of the payroll for every year thereafter with the intent of achieving a funding level of 70% after 30 years. The funding will be ongoing to maintain a 70% funding level thereafter.

Page 145 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

Effective June 28, 2009, a total charge of twenty six thousand, six hundred and forty (\$26,640) of employee payroll will be charged in the final three years of the Agreement so that contributions are at 1.764% in the final year of the Agreement. The purpose of the 0.764% increase in payroll contributions is to fund Pre Age 65 Retiree Health Insurance.

25.9 Actuarial Study

The Union understands and acknowledges that the City conducted an actuarial study to determine the percentage of payroll it needed to set aside each year and the rate of return of 7% it must achieve to fund the retiree health benefit provided in this Section. The City will conduct an actuarial study by an outside actuary of the retiree medical plan prior to June 30, 2002. After that time, the City will conduct an actuarial study by the outside actuary of the retiree medical plan every two to three years to review the funding status of the program. The outside actuary will be selected by mutual agreement of the parties. The Union and City agree that if the Actuary concludes that the City's funding of this benefit by contribution of 1% of the payroll for all miscellaneous employees is insufficient to fully fund the retiree medical benefits, the City shall not be required to increase its funding for this benefit to more than 1% of the payroll for miscellaneous employees. In the event that there are insufficient funds in the trust to cover all retirees' monthly health premiums, the City and the Union agree to meet and confer regarding the City's distribution of its 1% contribution.

ARTICLE 5 – TERMS AND CONDITIONS OF EMPLOYMENT

SECTION 26: PROBATIONARY PERIOD

26.1 Duration and Effect of Military Leave on Probationary Period

Original and promotional appointments from employment lists shall be tentative and subject to a probationary period of: one (1) year (full time equivalent exclusive of all leave and light duty completed within 18 months) upon original appointment to the unit; employees promoted within the unit are subject to a probationary period of six (6) months (half time equivalent exclusive of all leave and light duty completed within one year); in addition, employees in trainee programs with specified training, evaluation or probation periods shall be governed by the applicable provisions of such program. Probationary employees who are granted military leaves of absence shall complete the balance of their probationary period within a period of six (6) months following their return to City service. No provision of this Section shall be interpreted to preclude the City from establishing new classifications, which may require a probationary period of more than six (6) months.

26.2 Provisional Appointments to a Higher Classification

If, before completing the required probationary period, an employee is provisionally appointed to a higher class in the same or a related series of classes, the time served in such higher class shall be counted toward completion of the probationary period in the lower class.

26.3 Report Requirement

If the service of the probationary employee has been satisfactory to the department head, the department head shall file with the Director of Human Resources a statement in writing to such effect and stating that the retention of such probationer in the service is desired. If such service has been unsatisfactory, the department head shall file with the Director of Human Resources such a statement in writing with the recommendation to the City Manager that the employee be rejected.

26.4 Rejection during Probationary Period

During the probationary period, an employee may be rejected at any time without right of appeal or hearing in any manner. An employee rejected from a position to which he has been promoted shall be reinstated to the position from which he was promoted unless charges are filed and he is discharged as provided in Section 31 (Discharge). Career City employees who are rejected during their probationary period in a promotional position or fail to complete the training requirements of a training program (academics or OJT) shall revert to their prior classification.

26.5 Rejections Subject to Disciplinary Appeal

A promoted employee may not grieve his or her rejection from probationary period/training. However, an employee disciplined for cause while on promotional probationary period shall have access to the disciplinary appeal procedure.

26.6 Probationary or Temporary Employee Assignments

No probationary or temporary employee will be assigned as the only ground worker with an Electrician until such employee has completed one (1) month of familiarization with aerial truck equipment.

SECTION 27: TRANSFER

A transfer may be made at any time by the City Manager. Transfer shall not be used to effect a promotion, demotion, advancement, or reduction, each of which may be accomplished only as provided elsewhere in this Memorandum Agreement. No person shall be transferred to a position for which he or she does not possess the minimum qualifications. An employee with permanent status who is transferred from one class to another class shall assume permanent status in the class to which the employee is transferred.

SECTION 28: PROMOTION

Insofar as practicable and consistent with the best interests of the service, all vacancies in the competitive service shall be filled by promotion from within the competitive service after a promotional examination has been given and a promotional list established.

If, in the opinion of the City Manager, the best interests of the service can be served by an open, competitive examination instead of a closed, promotional examination, and if there is not already a promotional list for the higher position, which list has not been abolished and from which the vacancy could be filled, then the City Manager may instruct the Director of Human Resources to call for applications for the vacancy and arrange for an open, competitive examination and for the preparation and publication of an eligibility list.

Interview of Employees: A City employee who is on a closed promotional or open competitive list shall have the option to interview for the vacancy. A City employee who is unsuccessful and who so requests shall be advised of steps he or she may take to increase his or her competitive standing for future promotional opportunities.

Employees who have qualified for promotional lists shall be considered for promotion based on the following factors: previous work performance, previous training and experience, merit, ability, and seniority.

SECTION 29: DEMOTION

The City Manager may demote an employee who so requests it, or whose ability to perform his or her required duties falls below standard, or for disciplinary purposes. No employee shall be demoted to a class for which he does not possess the minimum qualifications as determined by the Director of Human Resources.

Notice of the demotion shall be given the employee not later than four (4) weeks prior to the effective date of demotion and a copy of said notice filed with the Director of Human Resources. Any employee who has been demoted shall be entitled to receive a written statement of the reasons for such action.

An employee with permanent status who is demoted shall assume permanent status in the class to which he or she is demoted.

Upon request of the employee, demotion may be made to a vacant position as a substitution for layoff. In such cases, the employee shall be restored to his or her former position without further examination whenever such position is again to be filled.

SECTION 30: SUSPENSION

The City Manager may suspend an employee from his position at any time for the good of the service, for a disciplinary purpose, or for other just cause. Suspension without pay shall not exceed twenty (20) working days, nor shall any employee be penalized by suspension for more than twenty (20) days in any fiscal year. Any employee who has been suspended shall be entitled to receive a written statement of the reasons for such action.

For the good of the service, a department head may suspend an employee for not more than three (3) working days for any one offense. Such suspension shall be reported immediately to the City Manager.

SECTION 31: DISCHARGE

An employee may be discharged at any time by the City Manager, but if the probationary period has been completed then such discharge must be for just cause. Any employee

who has been discharged shall be entitled to receive a written statement of the reasons for such action.

SECTION 32: RESIGNATION

An employee wishing to leave the competitive service in good standing shall file with the department head, at least two (2) weeks before leaving the service, a written resignation stating the effective date and reasons for leaving. The resignation shall be forwarded to the Director of Human Resources with a statement by the department head as to the resigned employee's service performance and other pertinent information concerning the cause for resignation. Failure of the employee to give the notice required shall be entered on the service record of the employee and may be cause for denying future employment by the City. The resignation of an employee who fails to give notice shall be reported by the department head immediately.

SECTION 33: REINSTATEMENT

A permanent or probationary employee who has resigned with a good record may be reinstated within two (2) years to his or her former position, if vacant, or to a vacant position in the same or comparable class without further competitive examination. This Section shall not be interpreted as a guarantee of reinstatement to an employee who has resigned with a good record and requests reinstatement within two (2) years.

SECTION 34: NOTICE OF TERMINATION

Department head shall notify the Director of Human Resources of all terminations of employment before or within two (2) days after the termination has taken place. Such notice shall be on forms provided by the Director of Human Resources and shall indicate the effective date and reasons for termination and a statement certifying whether or not the employee's services have been satisfactory.

ARTICLE 6 - GRIEVANCE AND DISCIPLINARY APPEAL PROCEDURE

SECTION 35: GRIEVANCES

35.1 Definition

A grievance is any dispute which involves the interpretation or application of those rules, regulations, and resolutions which have been or may hereafter be adopted by the

City Council to govern personnel practices and working conditions, including such rules, regulations, and resolutions as may be adopted by the City Council to effect memorandum agreements which result from the meeting and conferring process.

- A grievant may be any member of the bargaining unit covered by the terms of this Agreement, or the grievant may designate the Union to act on his or her behalf, or the Union itself may file a grievance on behalf of a member or group of members. Grievances must be filed in a timely manner. Grievances that are filed by the Union on behalf of a member or group of members, or when a grievant designates the Union to act on his or her behalf, are subject to settlement at any step of the grievance procedure at the Union's sole discretion.
- A member of the bargaining unit that files a grievance where the Union is not representing or acting on the grievant's behalf, may only utilize this grievance procedure through step 3. The Union retains the sole discretion to refer which, if any, grievances to move to arbitration.
- The parties agree that disclosure to the other party of all relevant information available to them is critical to the successful resolution of grievances at the lowest possible step of the grievance procedure. The parties therefore agree to disclose to each other the relevant information in their possession related to the grievance. Further, the parties agree to comply with reasonable grievance related information requests from the other party in a timely manner.

35.2 Procedure

Grievances shall be processed in the following manner:

- 35.2.1 **Informal Step:** An employee who believes that he or she has a grievance (and / or the employee's steward or Union representative) may discuss the complaint with the Deputy Director of Public Works, or with such subordinate management official as the Deputy Director may designate. If the issue is not resolved informally, the formal procedures hereinafter specified may be invoked.
- 35.2.2 **First Formal Step:** Any informal grievance which has not been resolved by the Deputy Director of Public Works or his or her designee, must be filed in writing to the Deputy Director of Public Works within thirty (30) days from the date the informal discussion was initiated. The written grievance must state specific issues involved, the decision rendered at the informal step, and the remedy sought. The Deputy Director or his or her designee shall have five (5) days to respond in writing to the formal, written grievance.

35.2.3 **Second Formal Step:** Any formal grievance which has not been resolved by the Deputy Director of Public Works or his or her designee, may be referred to the Director of Public Works or his or her designee within thirty (30) days from the date of the Deputy Director's formal written response, or if more than five (5) days have elapsed since the grievance was presented at the Deputy Director's level.

- 35.2.3.1 **Referral to Department Director:** The grievance may be referred to the Director's level and shall include a copy of the original grievance, the decision rendered at the Deputy Director's level, and a clear, concise statement of the reasons for the referral.
 - Director of Public Works may designate a personnel representative to investigate the merits of the complaint, to meet with the complaining employee (and/or the employee's steward or Union representative); if the grievant is not the Union itself, to meet with the officials of the Union; and to settle such grievance or to make recommendations thereon to the City Manager in his or her capacity as an employee relations officer. The Director of Public Works shall issue a formal written decision within five (5) days of receipt of the grievance or if a meeting is held with the employee and his or her Union, the meeting must be held within ten (10) days of receiving the referral. The Director of Public Works or his or her designee shall issue a written decision within five (5) days from his or her meeting with the complaining employee and/or Union representative. If the issue is not resolved the grievance may be referred to the Third Formal Step within thirty (30) days of receipt of the response from the Director of Public Works.
- 35.2.4 **Third Formal Step:** Any grievance which has not been resolved by the procedures hereinabove set forth may be referred to the City Manager by the grievant, the Union, or the Director of Public Works. Any such referral shall be in writing including a copy of the original grievance, the decision rendered at the Director's level, and a clear, concise statement of the reasons for the referral. The City Manager shall designate a Human Resources representative to investigate the merits of the complaint, and upon request from the employee or the union, that representative to meet with the complaining employee (and/or the employee's steward or Union representative), if the grievant is not the Union itself, to meet also with the officials of the Union, and to settle such

grievance or to make recommendations thereon to the City Manager in his or her capacity as an employee relations officer.

The City Manager shall issue a decision within ten (10) days of receipt of the grievance referral or within ten (10) days of the meeting, or if a meeting is held with the employee and/or the Union, the meeting must be held within ten (10) days of the referral. Grievances filed by a bargaining member where the Union is not representing or acting on the grievant's behalf, shall end at this step and no further right of appeal exists. The City Manager's decision shall be final.

Union controlled grievances: If the issue is not resolved, the grievance may be referred to Arbitration by the Union pursuant to Section 35.2.5 (Arbitration: The Union retains the sole discretion to move a grievance to arbitration) below within thirty (30) days of receipt of the City Manager's response. Provided further that the Union shall forward to the City the Union's portion of the State Mediation and Conciliation Services (SMCS) fee within sixty (60) days of receipt of the City Manager's response. Failure by the Union to meet this sixty (60) days deadline for both referral to Arbitration and payment of the SMCS fee shall be deemed as a full and complete waiver by the Union to appeal the City Manager decision to Arbitration and the City Manager decision shall be final and binding on all parties.

35.2.5 **Arbitration**: The Union retains the sole discretion to move a grievance to arbitration. Any grievance which has not been resolved by the procedures hereinabove set forth may be referred by the Union, or the City to an impartial arbitrator who shall be designated by mutual agreement between the Union and the City, using the State Mediation and Conciliation Services (SMCS) to provide a list of five (5) arbitrators. The City and the Union will alternately strike a name until one remains. The remaining name will be the arbitrator. The fees and expenses of the arbitrator, the State Mediation and Conciliation Services (SMCS) and of the court reporter shall be shared equally by the Union and the City. Each party, however, shall bear the cost of its own presentation, including preparation and post-hearing briefs, if any. Arbitrator decisions on matters properly before them shall be final and binding on both parties.

The arbitrator may hear testimony, receive written briefs, interview witnesses, and conduct any investigation she or he deems appropriate, and shall render a final and binding decision to the parties which will end the formal grievance process.

- 35.2.5.1 **Arbitrator Jurisdiction:** No arbitrator shall entertain, hear, decide, or make recommendations on any dispute involving a position over which a formally recognized employee organization has jurisdiction unless such dispute falls within the definition of a grievance as hereinabove set forth in Section 35 (Grievances).
- 35.2.5.2 **No Modifications:** Proposals to add to or change this Memorandum Agreement or written agreements or addenda supplementary hereto shall not be arbitrable, and no proposal to modify, amend, or terminate this Memorandum Agreement, nor any matter arising out of or in connection with such proposal, may be referred for arbitration under this Section; and no arbitrator shall have the power to amend or modify or recommend amendment or modification of this Memorandum Agreement or any written agreements or addenda supplementary hereto or to establish or recommend establishment of any new terms and conditions of employment.
- 35.2.5.3 **Timeliness of Award:** No arbitrator will be selected hereunder who does not agree to render an award not later than thirty (30) calendar days after the close of the hearing. By mutual agreement between the City and the Union, the arbitrator may render an award immediately upon the conclusion of the presentation of evidence.

35.3 EEO Complaints

Any grievance which in any way affects the implementation of the City's Equal Employment Opportunity Policy shall not be subject to arbitration. The decision as to whether or not implementation of the Equal Employment Opportunity Policy is in any way involved shall be made by the City Manager in his or her sole discretion. If, in his or her judgment, any grievance which involves the Equal Employment Opportunity Policy, the Equal Employment Opportunity and Diversity Officer shall notify the Union to that effect in writing within seven (7) days of the date upon which the grievance is received in the Human Resources Department and, in such notification, refer to that Section of the Equal Employment Opportunity Complaint Investigation & Resolution Procedure which is involved; provided, however, that such notice may come at any time prior to arbitration if additional factors come to the attention of the Equal Employment Opportunity and Diversity Officer on the basis of which he or she considers it appropriate to change his or her original determination.

35.4 Compensation Complaints

All complaints involving or concerning the payment of compensation shall be initially filed in writing with the Director of Human Resources. In such cases, no adjustment shall be retroactive for more than thirty (30) days from the date upon which the complaint was filed or thirty (30) days from the date when an employee may reasonably be expected to have learned of said claimed violation. Only grievants which allege that employees are not being compensated in accordance with the rules, regulations, and resolutions of the City Council or in accordance with the understanding contained in any Memorandum Agreement which has resulted from the meeting and conferring process shall be considered as grievances. Any other matters of compensation are to be resolved in the meeting and conferring process, shall be deemed withdrawn until the meeting and conferring process is next open for such discussion.

35.4.1 **Response Time Limits:** The Director of Human Resources or his or her designee shall issue a decision in writing within twenty (20) days from the time he or she received the compensation grievance in writing. The grievant or the Union may refer the decision to the Third Formal Step (Section 35.2.4 - Third Formal Step) of this Memorandum Agreement) within twenty (20) days of receipt of the written decision.

35.5 Changes or Interpretations

No changes in this Memorandum Agreement or interpretation thereof (except interpretations resulting from arbitration proceedings hereunder) will be recognized unless agreed to by the City Manager and the Union.

35.6 Extension of Timelines

Timelines noted in this Section may be extended for cause upon mutual agreement between the City and the Union.

35.7 Days

All references in this Section 35 (Grievances) to "days" shall mean normal working days except as otherwise noted.

35.8 Grievances Challenging Disciplinary Action

- 35.8.1 Any grievance involving any disciplinary action up to as three (3) day suspension will be filed at the Director level.
- 35.8.2 All grievances involving demotion, discharge, or suspension of greater than three (3) days will be filed at the City Manager's level of the grievance

Page 155 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

- procedure. If the issue is not resolved within fifteen (15) days of referral to this step, the procedures hereinafter specified may be invoked.
- 35.8.3 No grievance involving the suspension or discharge of an employee will be entertained unless it is filed in writing with the City Manager within five (5) working days of the time at which the affected employee was notified of such action in writing. If the City Manager, in pursuance of the procedures outlined in Section 35.2.4 (Third Formal Step) above, resolves a grievance which involves suspension or discharge, he or she may agree to payment for lost time or to reinstatement with or without payment for lost time.
- 35.8.4 Arbitrator decisions on matters properly before them which pertain to the suspension or discharge of an employee shall be final and binding on both parties, to the extent permitted by the Charter of the City.

ARTICLE 7 – GENERAL PROVISIONS

SECTION 36: GENERAL PROVISIONS

36.1 Personal Conduct

- 36.1.1 No employee shall accept appointment to the deputyship or assistantship of any county or state office or position, or otherwise incur an obligation of civil public service outside his or her regular municipal employment without first obtaining the recommendation of the head of his or her department and of the City Manager.
- 36.1.2 No employee shall be disciplined for off-the-job activities which do not affect his or her job performance.
- 36.1.3 Employees shall so arrange their personal financial affairs so that the demands of creditors and collection agencies shall not impose a recurring burden upon the offices of the City Manager, the department head, or the Director of Human Resources for the purpose of making collections.
- 36.1.4 Full-time City employees may not carry on concurrently with their public service any private business or undertaking, attention to which affects the time or quality of their work or which casts discredit upon or creates embarrassment for the City government.
- 36.1.5 No official or employee who wears a badge or other official insignia as evidence of his or her authority and identity shall permit such badge or insignia to be used or worn by any other person of the same or another department or otherwise to leave his possession without approval by the head of this department. The department head shall not grant such approval except as to persons regularly and formally appointed by the City Manager to the position designated by the badge or insignia.

36.2 Rain Gear

The City will provide rain gear for employees in Representation Unit C assigned to work in inclement weather.

36.3 Tools

Each employee shall continue to be responsible for providing tools of the trade or other equipment, but shall receive a flat annual tool replacement allowance each July 1.

New employees hired before or after July 1 will be entitled to a prorated tool replacement allowance payable the following July 1 after employment.

36.3.1 **Allowance:** Effective July 1, 2015, the tool replacement is five hundred dollars (\$500). Tool allowances shall be paid at the end of the fiscal year in which they are earned.

36.4 Shoes

Effective June 18, 2017, all classifications covered by this MA shall be eligible for reimbursement of up to two hundred dollars (\$200.00) per fiscal year for the cost of safety shoes. Employees receiving this reimbursement must wear safety shoes on duty.

36.4.1 **Reimbursement**: Employees shall be required to submit a reimbursement request, along with receipts verifying the purchase of safety shoes within sixty (60) days of purchase. Reimbursements will be paid within forty-five (45) days of complete reimbursement requests.

36.5 Training

- 36.5.1 **Responsibility:** Responsibility for developing training programs for employees shall be assumed by the City Manager. Such training programs may include lecture courses, demonstrations, assignment of reading matter, or such other devices as may be available for the purpose of improving the efficiency and broadening the knowledge of municipal employees in the performance of their duties. The City shall endeavor to provide training for all new technology purchased by the City and maintained by members of this Agreement.
- 36.5.2 Advancement/Promotional Consideration for Special Training Courses: Participation in and successful completion of special training courses may be considered in making advancements and promotions. Evidence of such activity shall be filed with the Director of Human Resources by the department head.
- 36.5.3 **Mileage and Tuition Reimbursement:** The City shall reimburse all employees for mileage and tuition expenses related for attendance at job-related courses. Provided that the employee must have prior authorization from his or her department head and the course has been approved by the City.
- 36.5.4 **Education Leave:** The City shall allow up to forty (40) hours off with pay per year to employees:

- 36.5.4.1 Who are required to obtain a license (excluding Class C driver's licenses) issued by the State of California and, in order to do so, must take courses which were not offered as a part of their basic curriculum or.
 - 36.5.4.2 Who are required to obtain continuing education as a condition of license renewal; provided the license is required by the City for the employment or continued employment of the employee and that no more hours than are required by the State shall be granted. Employees seeking time off to take courses for an initial license must provide verification that the course was not offered as a part of their basic curriculum.
 - Employees who take classes during non-scheduled work hours in order to retain a job required license or to meet continuing education requirements shall be allowed time off from work on an hour-for-hour basis without loss of compensation or other benefits.
- 36.5.4.3 To obtain education and training related to job skills to enhance performance, or to qualify for promotion, at the discretion of the department head.
- 36.5.5 **Management/Union Meetings:** The City and Union agree that it is beneficial to both parties to discuss and develop training for all classifications. The Director of Public Works and the Union shall meet on a quarterly basis to discuss and review progress made in addressing initial employee orientation, new equipment training, and vendor training.
- 36.5.6 **Commercial Driver's License**: The City shall provide the necessary training for employees to obtain a California Class B driver's license when such license is a requirement of their job classification and a condition of employment. Employees who fail to obtain the required California Class B driver's license after training will be subject to rejection from the probationary period pursuant to Section 26.4 (Rejection during Probationary Period) of this Memorandum Agreement.

36.6 Use of Automobiles

The City Manager shall govern the use of City-owned automotive equipment and privately-owned automotive equipment by such rules and regulations as he or she may establish. Compensation shall be given in the form of a cash allowance for the use of private-owned automobiles on City business if such use has been

authorized in advance by the City Manager. The cash allowance will be equal to the amount established by the Internal Revenue Service, and will change as necessary to comply with IRS Standard Mileage Rate.

36.7 Safety

- 36.7.1 **Safety Committee:** The City and the Union will make every effort to maintain excellent health and safety standards. No employee shall be required to perform work with unsafe equipment or in situations which are injurious to his or her health or safety. To further these purposes, the City shall maintain an ongoing safety program which shall include committees comprised of representatives from the Union and appropriate supervisory personnel.
- 36.7.2 **Rainy Weather Assignments:** The City shall endeavor to avoid assigning non-emergency, routine work in rainy weather, which would constitute a safety hazard.

36.8 YMCA Group Membership

- The City shall offer employees a low or no-cost group membership in the Berkeley Central YMCA. As of July 1, 1999, the City will pay 75% of the membership fee. If the monthly fee is increased to more than \$60, the employee share will be capped at \$30 per month; the City will pay the balance. The amount the City contributes toward the employee's monthly membership fee is subject to federal and state income tax withholding.
- Use of the YMCA membership by a City of Berkeley employee, as provided for in this Agreement, is non-compensable, is not a part of the employee's work-related duties, is not required for employment and is not condoned as part of a physical fitness program, or required to remain top physical conditioning for the employee's job performance.
- The City of Berkeley or its Claims Administrator may not be liable for any injury that arises out of a City of Berkeley employee's participation in and use of a YMCA membership.

36.9 Annual Performance Evaluation

The City may implement a program of annual performance evaluation. Such evaluations shall be conducted by the employee's immediate supervisor and reviewed by additional levels of supervision. Each employee may make written comments on the evaluation, which shall be made a part of the employee's personnel records.

ARTICLE 8 - PUBLIC EMPLOYEES RETIREMENT SYSTEM

SECTION 37: PAYMENT OF EMPLOYEES' PERS CONTRIBUTIONS

37.1 Miscellaneous Designation

The City shall continue participation under the Miscellaneous Employees Plan of the California Public Employees' Retirement System, (CalPERS).

37.2 "Classic Employees" Definition

Classic Employees are defined as current employees and future employees who do not qualify as "New Members" under the California Public Employees' Pension Reform Act of 2013 (PEPRA).

37.3 "New Member" Definition

"New Members are as defined in the Public Employees' Pension Reform Act of 2013 (PEPRA), Government Code Section 7522.04(f).

37.4 CalPERS Retirement Formula for "New Members" as Defined Under the Public Employees' Pension Reform Act of 2013 (PEPRA)

"New Members" as defined by PEPRA who are hired by the City on or after January 1, 2013 shall be entitled to the retirement formula set forth in PEPRA.

37.5 CalPERS Retirement Formula and Employer Paid Member Contribution for Classic Employees (i.e. current employees and employees who do not qualify as "New Members" under PEPRA)

Current employees and other employees who do not qualify as "New Members" under PEPRA shall continue to be entitled to the 2.7% at age 55 retirement formula benefit, and the City shall continue the contribution of eight percent (8%) to CalPERS on behalf of the employee.

37.6 CalPERS New Members Normal Share of Cost

New Members as defined by PEPRA who are hired on or after January 1, 2013 shall pay 50% of the normal share of cost required by PEPRA. New Members shall receive any other additional optional CalPERS benefits that the City provides to Classic Employees as allowed by PEPRA.

37.7 Reporting of Contributions

Effective July 3, 1994, contributions made pursuant to Section 37.5 above have been reported to CalPERS as "special compensation" as provided in Government Code Section 20636(c)(4) pursuant to Section 20691. Said contributions shall not apply in the case of temporary or provisional employees.

- The member contributions made by the City shall not be considered as a part of an employee's salary for the purpose of computing straight time earnings, compensation for overtime worked, or education incentive pay; nor shall such contribution be taken into account in determining the level of any other benefit which is a function of or percentage of salary. Pursuant to California Government Code Section 20691, the City shall not increase, reduce or eliminate payments of the normal contributions on behalf of the employee without engaging in the meet and confer process with the Union.
- The City will not treat these contributions as compensation subject to income tax withholding unless the Internal Revenue Service or Franchise Tax Board indicates that such contributions are taxable income subject to withholding. Each employee shall be solely and personally responsible for any federal, state or local tax liability of the employee that may arise out of the implementation of this Section or any penalty that may be imposed therefore.

37.8 City Contracts with CalPERS

The City's contract with CalPERS includes the following optional benefits:

- 37.8.1 Classic Employees One Year of Final Compensation: Classic Employees as defined in Section 37.2 ("Classic Employees" Definition) above and as provided in Government Code Section 20042 (July 9, 1978).
- 37.8.2 **New Members Three Years Final Compensation**: Provided further that New Members as defined by PEPRA hired on or after January 1, 2013 shall be eligible to receive retirement allowance based on three (3) highest consecutive years of compensation under the plan as provided under the California Public Employees' Pension Reform Act of 2013, or as subsequently amended.
- 37.8.3 Post Retirement Survivor Allowance as provided in Sections 21624, 21626 and 21628 (December 16, 1973).
- 37.8.4 Post Retirement Survivor Allowance to Continue after Remarriage as provided in Section 21635 (July 18, 1986).
- 37.8.5 Credit for Unused Sick Leave as provided in Section 20965 (June 26, 1988).
- 37.8.6 1959 Survivor Benefits to Surviving Spouse at Age 60 as provided in Section 21580 (December 16, 1973).

- 37.8.7 Third Level of 1959 Survivor Benefits as provided in Section 21573 (November 28, 1996).
- 37.8.8 2% @ 55 for Local Miscellaneous Members as provided in Section 21354 (June 30, 1992).
- 37.8.9 Military Service Credit as Public Service as provided in Section 21024 (April 9, 1999).
- 37.8.10 Public Service Credit for Peace Corps or AmeriCorps/VISTA Service as Provided in Section 21023.5 (April 14, 2000).

37.9 Index Level 1959 Survivor Benefit

The City intends to amend its contract with CalPERS to provide the Indexed Level of the 1959 Survivor Benefit as provided in Section 21574.5 when administratively feasible.

37.10 Classic Members' Pension Contribution through 20516 a Contract Amendment

37.10.1 **July 3, 2016:** Effective July 3, 2016, employees will contribute one and one-quarter percent (1.25%) toward the City's CalPERS employer share of pension through a 20516 CalPERS amendment that allows such contributions via automatic payroll deduction on a pre-tax basis, in exchange for the City granting the salary increase set forth in Section 11.1.5 (July 3, 2016 - Salary/PERS Exchange) of this MA. Such employee deductions by the City shall be used towards the City's CalPERS required contributions.

37.11 New Members' Pension Contributions through 20516 a Contract Amendment

- 37.11.1 **July 3, 2016**: Effective July 3, 2016, in addition to the contributions in Section 37.4 (CalPERS Retirement Formula for "New Members" as Defined Under the Public Employees' Pension Reform Act of 2013 (PEPRA)), New Members will contribute one and one-quarter percent (1.25%) towards the City's CalPERS employer share of pension through a 20516 CalPERS amendment that allows such contributions as pre-tax via automatic payroll deduction, in exchange for the City granting the salary increase set forth in Section 11.1.5 (July 3, 2016 Salary/PERS Exchange) of this MA. Such employee deductions by the City shall be used towards the City's CalPERS required contributions.
- 37.11.2 The parties recognize that the CalPERS 20516 amendment that allows employee contributions toward the employer rate is in addition to the

required 50% of the normal cost of "New Members" benefits and made in consideration of additional salary increases in Section 11.1.5 (July 3, 2016 - Salary/PERS Exchange) above (a total of 1% salary increases in exchange for employees paying a total of 1.25% towards CalPERS pension cost).

As soon as administratively possible, following council approval of this agreement, the additional contribution of 1.25%, as indicated in Section 37.11.1, shall be eliminated through a contract amendment with CalPERS. The start of the first full pay period following completion of the contract amendment shall be the effective date of the reduction.

SECTION 38: SUPPLEMENTAL RETIREMENT/DISABILITY INSURANCE PLAN

Effective January 1, 1983, the majority of miscellaneous employees under the City's contract with the State of California Public Employees Retirement System who were covered by the integrated Social Security Program voted to withdraw from participation in the Federal Social Security Program. In lieu of Social Security payments, the City has agreed to pay an amount equal to that percent of individual pay (6.7% payable on the first \$32,400 of salary paid in the calendar year) which had been paid by the City to Social Security as of December 31, 1982 to a Supplemental Retirement and Income and Long Term Disability Insurance Plan for those employees previously covered under the integrated PERS/Social Security Plan. Provisions of this plan are described in Berkeley Municipal Code Chapter 04.36.101 (Supplemental Retirement and Income Plan I) as amended and Berkeley Municipal Code Chapter 04.38.101 (Supplemental Retirement and Income Plan II) as amended. All employees hired after July 22, 1988 are in SRIP II. All employees hired prior to July 22, 1988 are in SRIP I unless they chose to enroll in SRIP II prior to December 17, 1988.

ARTICLE 9 - LAYOFF PROCEDURE

SECTION 39: LAYOFF

The layoff policy for the City of Berkeley is intended to provide the maximum employment protection of the City staff should a layoff become necessary. The policy also aims to minimize the impact such a layoff might have on the City's affirmative action accomplishments.

39.1 Announcement of Layoff

- 39.1.1 **Notification:** The City Council, City Manager, and department head shall make every reasonable effort to manage and budget the City's resources effectively and to plan for the delivery of City services in a manner which will avoid the necessity of laying off career City employees. If a reduction in the work force for more than thirty (30) calendar days is necessitated by, but not limited to, the following: a material change in duties and organization, adverse working conditions, return of employee from leave of absence, or shortage of work or funds, the City Manager shall notify the Director of Human Resources of the intended action and the reason for the layoff.
- 39.1.2 Freezing of Vacancies: Immediately following a decision which may involve the potential layoff of career City employees, the City Manager shall freeze all current City vacancies in the competitive service in similar and related classifications to those likely to be targeted for layoff, as well as all related full-time, benefited, temporary positions which are expected to last six (6) months or more, and notify all department heads that such current and anticipated vacancies will be frozen until further notice in order to implement the provisions of Section 39.6 (Flexible Placement Program).

39.2 Seniority Service Date

39.2.1 All service in the employ of the City shall be counted toward the establishment of an employee's seniority service date, including, for example, permanent, probationary, provisional, temporary (full-time and intermittent), seasonal, and exempt employment, as well as leaves of absence for obligatory military service while an employee of the City. Less than full-time service will be consolidated in equivalences of full-time service for the purpose of establishing the seniority service date. Time off as a result of formal disciplinary action will be subtracted from the seniority service date.

39.2.2 The auditor's office will maintain up-to-date and current seniority dates for all City employees holding probationary and permanent appointments.

39.3 Establishment of Seniority Lists

- 39.3.1 Whenever a layoff of one or more career employees becomes necessary, as defined above, such layoffs shall be made according to City-wide classification seniority lists. Upon receiving notification that the City Manager must proceed with a possible reduction in the work force, and following receipt of information concerning the specific positions, programs, and departments involved, the Human Resources Department will immediately establish separate probationary and permanent seniority lists for each classification targeted for layoff.
- 39.3.2 The names of all City employees holding permanent and probationary appointments in a given classification will be listed on the appropriate list in descending order by seniority service date. Employees on both lists shall be laid off on the basis of their seniority service date, i.e., employees with the least amount of total service shall be laid off first. All emergency, temporary, and provisional employees working in classifications similar to those identified for layoff must be terminated prior to the layoff of probationary or permanent employees. Employees on the probationary seniority list for a specific classification will be laid off prior to employees on the permanent seniority list for that classification.
- 39.3.3 Probationary or permanent employees temporarily acting out of classification and holding a provisional appointment in another classification will only be listed on a seniority list of the classification in which they hold permanent or probationary status targeted for layoff.
- 39.3.4 If two (2) or more employees on a seniority list have an identical seniority service date, the tie shall be broken in the following order:
 - 39.3.4.1 Time in classification the employee having least time in the classification shall be released first;
 - 39.3.4.2 By lot.

39.4 Employee Retreat Rights

- 39.4.1 Before an employee with permanent or probationary status may be released from employment with the City of Berkeley, the Human Resources Department must consider the employee's right to retreat to lower level classifications through which he or she was originally promoted or any subsequently created intermediate level career classification which provides normal progression through the classification series. Retreat rights shall also extend to employees who have not previously been promoted through a classification but for whom the classification is a natural progression or beginning in the classification series.
- 39.4.2 In the process of retreating, the same rules concerning the length of service, classification seniority lists, etc., apply as in the first stage of the layoff process. In order to retreat, the targeted employee must be higher on the seniority list for the classification into which he or she is retreating than at least one of the incumbents on the probationary or permanent seniority list for that classification.
- 39.4.3 If an employee is qualified for retreat into more than one classification with comparable salary ranges, or if a vacancy exists in a classification to which an employee is entitled to retreat, the options shall be discussed with the employee and due consideration shall be given to the employee's preferences. However, it is the prerogative of the City Manager to determine the final placement offer to the employee.
- 39.4.4 The retreating employee has a right to be retained in the highest salary range possible which is equal to or less than his or her present salary range. An employee involved in layoff does not have a right of mandatory placement to positions with a higher salary range, i.e., promotion.

39.5 Employee Notification

- 39.5.1 Emergency, temporary, intermittent, seasonal, etc., employees shall be notified individually, in writing, of pending layoff as soon as possible, but no definite time period is required. However, at least two (2) weeks notification is desirable if possible.
- 39.5.2 Provisional employees shall be notified individually, in writing, of pending layoff as soon as possible, with no less than fifteen (15) calendar days notification if targeted for release or reassignment.
- 39.5.3 Permanent, probationary, and career-exempt employees should be notified individually in writing of pending layoffs as soon as possible, with no less

than thirty (30) calendar days notification if targeted for release or reassignment.

- If an employee fails to accept a bona fide offer of reassignment within ten (10) calendar days after the offer has been made, he or she forfeits further right to employment retention. Acceptance of a reassignment does not remove the right of appeal under Section 39.9 (Appeal Procedure).
- 35.5.4 An employee who is transferred in lieu of layoff when his or her position has been eliminated shall have automatic return rights to the previous position if it is restored within one (1) year of the date of the transfer. If an employee with a full time position is offered a reduction in hours in that position or in a lower classification, the employee may elect to be targeted for layoff for purpose of consideration under Section 39.6 (Flexible Placement Program.) If there is no flexible placement available for the employee, the employee may accept the reduction in hours, in lieu of layoff.

39.6 Flexible Placement Program

- 39.6.1 In order to minimize the negative impact of layoff, the City Manager will, as previously stated in Section 39.1.2 (Freezing of Vacancies), impose a citywide freeze on all appropriate vacancies as soon as it has been determined that a layoff of career City employees may be necessary.
- 39.6.2 Following the release of all emergency, temporary, and provisional employees in classes similar to those targeted for layoff, and as soon as employees targeted for layoff have been identified and the provisions under Sections 39.4 (Employee Retreat Rights) have been carried out, the Human Resources Department will review and identify the frozen vacant classifications into which employees ultimately targeted for layoff may be placed on the basis of total experience and education. In making this decision, a waiver of minimum qualification standards, and/or the substitution of related experience and education may be made, with an understanding on the part of management and supervisory personnel that adequate on-the-job training, which can be completed within no more than six (6) months, will be provided to facilitate job adjustment and to compensate for the waiver of qualification standards if that has occurred.

This shall be called the qualification period. The employee shall be advised of their progress after two months and four and six months in the new classification. If at the end of the qualification period the department head notifies the Human Resources Department that the employee is

unable to adequately perform the assignment or fails a test for the position administered by Human Resources then the employee shall be subject to the layoff process.

- 39.6.3 Assignments under the Flexible Placement Program shall be limited to positions in the same or lesser salary range as the classification from which the employee is to be laid off, except that the City Manager may authorize the offer of a flexible placement to position with a maximum salary of no more than five (5) percent above the salary range as the classification from which layoff is targeted, when it is in the best interest of the City service to do so. Whenever flexible placement is made to a classification with a greater salary range, the appointment shall be probationary, in accordance with the terms of that classification.
- 39.6.4 Offers to positions under the Flexible Placement Program shall be made according to seniority service date and in accordance with the probationary and permanent seniority list certification process outlined in Section 39.3 (Establishment of Seniority Lists). All offers and placements made under this provision of the layoff policy shall be documented in detail, with records available for audit and review at all times.
- 39.6.5 If an employee fails to accept a bona fide written offer of an alternative job within ten (10) calendar days after the offer has been made, he or she forfeits further rights to employment retention. Acceptance of an alternative job under the Flexible Placement Program in no way jeopardizes an employee's standing on the reemployment priority lists on which his or her name has been placed in accordance with Section 39.7 (Reemployment Lists).

39.7 Reemployment Lists

- 39.7.1 The names of all probationary and permanent employees released from positions in the competitive service as a result of layoff must be placed on reemployment priority lists for those classifications from which they were separated, as well as all other classifications to which they have retreat rights in accordance with Section 39.4 (Employee Retreat Rights).
- 39.7.2 A reemployment priority list shall remain in effect for three (3) years.
- 39.7.3 Departments with vacancies in any classification for which there is an active reemployment priority list must use the reemployment priority list to fill their positions and may not use any other recruitment or appointment

method to fill a vacancy until appropriate reemployment lists have been exhausted.

- 39.7.4 When a vacancy occurs in a class for which there is a reemployment priority list, the name of the employee on the appropriate reemployment priority list with the highest seniority date shall be certified to the selecting official. Employees so certified from the reemployment priority list must be appointed to the existing vacancy.
- 39.7.5 If a former employee fails to accept a bona fide written offer of reemployment within fifteen (15) calendar days, his or her name will be removed permanently from the reemployment priority list from which the offer was made. Failure to accept an offer of reemployment to the classification with the highest salary range for which the employee is eligible for reemployment will result in automatic removal from all reemployment priority lists. However, the employee may decline (or accept) reemployment to lower salary range classifications without jeopardizing his or her standing on the reemployment priority list for the classification from which he or she was originally terminated.
- 39.7.6 Upon reappointment to the classification from which the employee was originally separated or demoted, the employee has the right to be placed at the step of the salary range which the employee held at the time of layoff or demotion.

39.8 Career Exempt Employees

Only those employees holding full-time, benefited, exempt positions, who in the past have achieved permanent status and have been continuously employed without a break in service between their career and exempt appointment, have the right to retreat to previously held career classifications, placement on the reemployment priority lists, and all other provisions governing layoff procedures. For the purpose of layoff, such employees shall be referred to as "career-exempt."

39.9 Appeal Procedures

Any permanent, probationary, or career-exempt employee who is laid off, demoted, or reassigned as a result of layoff who believes that the layoff procedure has been improperly administered as it pertains to the employee's case may appeal the action under Section 35 (Grievances). In addition, employees may, at all times before, during, and subsequent to layoff, review all records, including seniority lists, reemployment priority lists, documentation pertaining to appointments under the Flexible Placement Program, etc., which pertain to their classification and their rights under the provisions of the layoff policy.

Page 170 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

39.10 Reemployment Priority Rights

If it is determined that a vacancy has been filled by a non-reemployment priority list eligible in a classification for which a reemployment priority list existed and which included available applicants at the time, the former employee with reemployment rights shall be hired and given retroactive pay from the date that the vacancy occurred. The employee who was originally hired to fill the vacancy shall continue to be retained in City employment, provided he or she has completed the probationary period.

Page 171 of 183 2022-2025 Memorandum Agreement

City of Berkeley

IBEW Local 1245

SIGNATURE PAGE

representatives whose signatures appear bel	
FOR INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS, LOCAL 1245	FOR THE CITY OF BERKELEY
Bob Dean, Business Manager	Dee Williams-Ridley, City Manager
Charley Souders, Business Representative (Chief Negotiator)	Donald Ellison, Interim Director of Human Resources
Greg Marwick, Steward	Dawud Brewer, Employee Relations Manager (Chief Negotiator)
Ryan Schaffler, Negotiation Committee	Joy Brown, Operations Manager Public Works
	Amani Manning, Assistant Human Resources Analyst

EXHIBIT A - Hourly Salary Schedule

As of August 7, 2022 (3.0% Salary Increase)

Job	Don	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
	Rep	Classification ritle	FLJA	Steh I	Step 2	step s	Step 4	Steh 5
Code	Unit							
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	48.2945	50.6173	52.9270
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	35.3125	37.5934	39.5648	40.5647
51030	С	ELECTRICIAN	N	0	0	48.2945	50.6173	52.9270
51100	С	LEAD COMMUNICATION TECHNICIAN	N	0	0	51.6314	53.9685	56.5598
51080	С	LEAD ELECTRICIAN	N	0	0	51.6314	53.9685	56.5598

As of July 7, 2023 (3.0% Salary Increase)

Job Code	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	49.74.33	52.1358	54.5148
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	36.3719	38.7212	40.7517	41.7816
51030	С	ELECTRICIAN	N	0	0	49.7433	52.1358	54.5148
51100	С	LEAD COMMUNICATION TECHNICIAN	N	0	0	53.1803	55.5876	58.2566
51080	С	LEAD ELECTRICIAN	N	0	0	53.1803	55.5876	58.2566

Page 173 of 183 2022-2025 Memorandum Agreement

IBEW Local 1245

As of July 9, 2024 (1.0% Salary Increase)

Job Code	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	50.2407	52.6572	55.0599
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	36.7356	39.1084	41.1592	42.1994
51030	С	ELECTRICIAN	N	0	0	50.2407	52.6572	55.0599
51100	С	LEAD COMMUNICATION TECHNICIAN	N	0	0	53.7121	56.1435	58.8392
51080	С	LEAD ELECTRICIAN	N	0	0	53.7121	56.1435	58.8392

Note: Employee's monthly salary will depend upon the number of hours worked. Employees hired before January 9, 2000, have an option (discussed in detail in Exhibit E (Hours and Days of Work) to work either a 37.5-hour work week or a 40-hour work week. Employees hired after January 9, 2000 work a 40-hour per week schedule.

EXHIBIT C

CITY OF BERKELEY MUNICIPAL CODE CHAPTER 4.04.120

4.04.120 Exemption from Career Service (At-Will Employees).

The provisions of this chapter shall apply to all positions in the service of the city except:

- A. The city manager, assistant city manager, deputy city manager, assistant to the city manager, police review commission officer, police review commission investigator, and assistant to the mayor;
- B. Assistant, associate and senior management analyst when appointed to the city manager's department or to the Budget Unit of the Management and Administrative Services Agency; and secretary to the mayor, administrative secretary and secretary when appointed to the mayor's office;
- C. All department heads, health officer, and supervising psychiatrist;
- D. Persons employed seasonally in the summer camps;
- E. Persons employed as reserve or emergency employees during the period of national emergency as provided in the ordinance creating such employment;
- F. All persons who are paid at an hourly rate with the exception of library aides. Persons appointed as hourly Library Aides shall be part of the career service with the exception of those hired to fill temporary positions of six months or less.

Persons appointed to these positions which are exempt from the career service shall serve at the pleasure of the appointing authority and may be demoted, suspended, or otherwise rejected at any time without cause and without right of appeal or hearing in any manner. Except that, any employee in the career service promoted or transferred to a position not included in the career service and made exempt from the career service shall be reinstated to his or her career appointment from which he or she was promoted or transferred if within six months of appointment to a position not included in the career service, action is taken to dismiss him or her, unless charges are filed and the employee is discharged in accordance with this chapter and the rules established hereunder for positions in

Page 175 of 183 2022-2025 Memorandum Agreement

City of Berkeley IBEW Local 1245

the career service. All employments designated in this Section shall be entitled to only those benefits provided for at-will employees in the personnel rules and regulations or in applicable memoranda agreements.

EXHIBIT D – Service Vacation Leave

Years During Which Service Rendered	Number of Years of Service	AUTHORIZED ANNUAL VACATION LEAVE (IN CALENDAR WEEKS)
All years prior to 01/01/1950		2
01/01/1950 through 12/31/1956	First 10 years of service Years of service in excess of 10	2 3
01/01/1957 through	First 5 years of service	2
12/31/1961	6 through 25 years of service Years in excess of 25	3 4
01/01/1962 through 12/31/1965	First 5 years of service 6 through 20 years of service	2 3 4
01/01/1966 through	Years of service in excess of 20 First 5 years of service 6 through 20 years of service	
06/30/1970	21 through 25 years of service Years of service in excess of 25	4 5
Subsequent to 06/30/1970	First 5 years of service 6 through 15 years of service 16 through 25 years of service	2 3 4
	Years of service in excess of 25	5
Effective 01/01/1981	First 4 years of service 5 through 12 years of service 13 through 20 years of service	2 3 4
	First 3 years of service 4 through 11 years of service	2 3
Effective 07/01/1984	12 through 17 years of service 18 and subsequent years of service	4 5
	First 3 years of service 4 through 11 years of service	2 3
Effective 12/08/1987	12 through 17 years of service 18 through 24 years of service 25 and subsequent years of service	4 5 6

EXHIBIT E - HOURS AND DAYS OF WORK

The hours and days of work applicable to employees in Representation Unit C, as set by the City Manager, are presently as follows:

1. Hours per Day

Field Operations: Working arrangements may vary as to daily schedules under the

reduced workweek.

Flexible Hours: On a job-by-job basis, with prior approval of the Senior Electrical

Supervisor, a crew may schedule a workday, or workdays, other than

the normal workday schedule of 7:00 a.m. to 3:30 p.m.

2. Hours per Week (Basic)

Field Operations: 40 hours. However, employees in Representation Unit C as of January 9, 2000 have the option of retaining a 37.5-hour per week work schedule. An employee will have an option to change his work schedule to 40 hour per week at the beginning of a payroll period. An employee will have a one-time option to revert to a 37.5 hour per week work schedule at the beginning of a payroll period. After exercising this one time option, if the employee chooses to change his work schedule to 40 hours per week, the employee must maintain this work schedule for the balance of his career in this Representation Unit. All persons hired on or after January 9, 2000 will work a 40-hour per week schedule and will not be permitted to change his or her work schedule to 37.5 hours per week.

The parties may agree in writing to alternative work schedules other than those set forth in this Agreement, such as a 9-80 work schedule which produces an 80-hour pay period with the hours worked in nine work days. The City will take into consideration its operational needs, including the impact of potential overtime, in deciding whether to approve an alternate work schedule. However, such alternate work schedule will not be unreasonably denied.

3. Work Week

The work week will begin at 12:01 a.m., Sunday and end at Saturday midnight. Regular days off will be considered to be Saturday and Sunday.

City of Berkeley

IBEW Local 1245

EXHIBIT F - IMPLEMENTING THE REDUCED WORK SCHEDULE

NOTE: This Exhibit F applies only to employees hired before January 9, 2000 who are permitted to work a 37.5 hour per week work schedule subject to the conditions set forth in Exhibit E (Hours and Days of Work).

The City and the Union agree that the implementation of the reduced work schedule which is to be effective June 29, 1986 will be as follows:

That one-fourth of the Electrical Division employees will be off every fourth Monday on a staggered basis. This time would be treated as a scheduled day off and could not be adjusted or moved. A yearly schedule would be set up at the beginning of the year so that each employee would know what days he would have off. Since the agreed upon reduced work schedule does not calculate evenly into one day off every four weeks, it is proposed that an additional day off (Friday) be given once every 16 weeks. This agreement very closely approximates the reduced work schedule. Any discrepancies would be credited to the employee's vacation at the end of the contract year.

For Fair Labor Standards Act purposes, the workweek is defined as 40 hours.

Sick leave, vacation, and other compensable leaves of absences shall be based on the 40 hour work week.

When a scheduled day off falls on a holiday, the next regularly scheduled work day shall be observed as the scheduled day off.



EXHIBIT G

DIRECTIVE PROHIBITING UNAUTHORIZED CONSTRUCTION

Office of the City Manager

July 31, 2001

To:

Department Directors

From:

Weldon Rucker City Manager

Subject:

DIRECTIVE PROHIBITING UNAUTHORIZED CONSTRUCTION OR FACILITY MAINTENANCE WORK IN CITY FACILITIES

It has come to my attention that some construction, building maintenance and electrical work is being improperly performed at various facilities that are either owned or leased by the City. Because this work can affect the health and safety of all City employees and can result in a significant potential liability for the City, it is absolutely essential that any such work be done by competent and experienced individuals. Even if licensed contractors are brought in to do the work, they may not be fully aware of how their work will impact other City projects or facilities. Furthermore, already scarce City funds are wasted when Public Works has to correct improper or hazardous work performed by unauthorized individuals or contractors. Accordingly, such construction and maintenance work must only be performed by properly trained and supervised Public Works maintenance staff or contractors selected by, and under the direction of, the Public Works Department.

Therefore, I am reminding and directing all department directors to ensure that absolutely no construction, building maintenance or electrical work be performed in any City owned or leased facilities by contractors or anyone else not under the direction of Public Works. All construction and facility maintenance work is to be referred to the Public Works Department.

For routine maintenance, minor repairs, or minor physical modifications, contact the Public Works Customer Service representative at 644-6620. For larger construction projects, contact Public Works Capital Improvement Projects at 981-6330. Public Works can then make appropriate decisions as to whether construction or maintenance work should be performed by City employees or by outside contractors, utilities, or agencies.

Thank you, in advance, for your full compliance with this direction.

cc: City Manager's Office

2180 Milvia Street, Berkeley, CA 94704 Tel: 510.981.7000 TDD: 510.981.6903 Fax: 510.981.7099 E-mail: manager@ci.berkeley.ca.us

Page 180 of 183

RESOLUTION NO. ##,###-N.S.

APPROVING A NEW CLASSIFICATION AND SALARY RESOLUTION FOR INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS, LOCAL 1245 AND RESCINDING RESOLUTION NO. 69,962-N.S.

WHEREAS, the City Council has approved a new three-year Memorandum Agreement with International Brotherhood of Electrical Workers, Local 1245 which includes salary increases; and

WHEREAS, it is necessary for the City Council to adopt a new Classification and Salary Resolution to reflect the salary adjustments reflected in the new Memorandum Agreement.

NOW THEREFORE, BE IT RESOLVED that the Council of the City of Berkeley adopts a new Classification and Salary Resolution for employees in Representation Unit C (Electrical Occupations) effective July 1, 2022 through June 30, 2025 to incorporate changes to the salary schedule as shown in Exhibit A (August 7, 2022) attached hereto and made a part thereof.

BE IT FURTHER RESOLVED that Resolution No. 69,962-N.S. is hereby rescinded.

Exhibit

A: International Brotherhood of Electrical Workers, Local 1245 Classification/Salary Resolution Period: August 7, 2022 through June 30, 2025

EXHIBIT A - Hourly Salary Schedule

As of August 7, 2022 (3.0% Salary Increase)

Job Code	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	48.2945	50.6173	52.9270
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	35.3125	37.5934	39.5648	40.5647
51030	С	ELECTRICIAN	N	0	0	48.2945	50.6173	52.9270
51100	С	LEAD COMMUNICATION TECHNICIAN	N	0	0	51.6314	53.9685	56.5598
51080	С	LEAD ELECTRICIAN	N	0	0	51.6314	53.9685	56.5598

As of July 7, 2023 (3.0% Salary Increase)

Job Code	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	49.7433	52.1358	54.5148
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	36.3719	38.7212	40.7517	41.7816
51030	С	ELECTRICIAN	N	0	0	49.7433	52.1358	54.5148
51100	С	LEAD COMMUNICATION TECHNICIAN	N	0	0	53.1803	55.5876	58.2566
51080	С	LEAD ELECTRICIAN	N	0	0	53.1803	55.5876	58.2566

As of July 9, 2024 (1.0% Salary Increase)

Job Code	Rep Unit	Classification Title	FLSA	Step 1	Step 2	Step 3	Step 4	Step 5
32020	С	COMMUNICATIONS TECHNICIAN	N	0	0	50.2407	52.6572	55.0599
52180	С	ELECTRICAL PARTS TECHNICIAN	N	0	36.7356	39.1084	41.1592	42.1994
51030	С	ELECTRICIAN	N	0	0	50.2407	52.6572	55.0599
51100	С	LEAD COMMUNICATION TECHNICIAN	N	0	0	53.7121	56.1435	58.8392
51080	С	LEAD ELECTRICIAN	N	0	0	53.7121	56.1435	58.8392

DocuSign Envelope ID: 56C1B0AA-F09F-4304-A5F7-7269E0A355D5 Page 182 of 183

CITY OF BERKELEY / IBEW, Local 1245 **2022 MOU NEGOTIATIONS** CITY COMPREHENSIVE PROPOSAL #1 – July 13, 2022

The Parties have reached a Tentative Agreement on the terms below and agree that all other MOU provisions will remain the same. This Tentative Agreement is subject to the ratification of the IBEW, Local 1245 membership and adoption by the City Council.

SALARIES <u>l.</u>

11.1 Salary Rates

Salary rates for the period of June 30, 2022 July 1, 2022 through June 30, 20252, for all classes of positions in Representation Unit C shall be set according to the classifications and salary ranges assigned to those classifications listed in Exhibit "A" and attached hereto and made part hereof.

Salary Increase upon Council Approval: Effective July 1, 2022, t The salary ranges for 11.1.1 theose classification covered by this Agreement as listed in Exhibit A will receive a salary increase of eight percent (8.0%) and shall be show in Exhibit A. as follows:

Effective the first full pay period following council approval, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of three percent (3.0%).

Effective July 9, 2023, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of three percent (3.0%).

Effective July 7, 2024, the salary ranges for the classification covered by this Agreement as listed in Exhibit A will receive a salary increase of one percent (1.0%).

11.1.2 One-Time Payment: A one-time non-pensionable lump sum payment of \$1,000.001000.00 for each career bargaining unit employee in this unit shall be conferred and paid the first full pay period after council approval [TBD]on July 13, 2022. Eligible employees must be current City employees in paid status for the full pay period this lump sum is conferred and shall not apply to unit members who are in unpaid status. Such payment shall be subject to required payroll tax deductions. The parties agree that tThis lump sum provision does not create or bind the City to any precedent or past practice.

> If during any fiscal year during the term of this agreement, the City reaches agreement with the largest bargaining unit or extends the unrepresented employees an across the board Cost of Living Adjustment (COLA) increase, wage increase, lump sum payment, or other benefit increase the City agrees to apply these same increases to the salary ranges and/or benefits for those classifications covered by this MA as listed in Exhibit В.

II. State Electrician Certification Premium

11.20 State Certification Premium (New Section)

An employee who possesses and maintains a "General Electrician" state certification from the Department of Industrial Relations shall receive an additional three percent (3%) to their base rate of pay. The City shall retain the right to add the general electrician state certification as a minimum qualification for the Electrician and Lead Electrician classifications. If the certification is made a minimum qualification, active employees, employed in classifications within this unit at the time of implementation, shall be exempt from this qualification for the remainder of their employment.

III. 37.11 New Members' Pension Contributions through 20516 a Contract Amendment

- 37.11.1 July 3, 2016: Effective July 3, 2016, in addition to the contributions in Section 37.4 (CalPERS Retirement Formula for "New Members" as Defined Under the Public Employees' Pension Reform Act of 2013 (PEPRA)), New Members will contribute one and one-quarter percent (1.25%) towards the City's CalPERS employer share of pension through a 20516 CalPERS amendment that allows such contributions as pre-tax via automatic payroll deduction, in exchange for the City granting the salary increase set forth in Section 11.1.5 (July 3, 2016 - Salary/PERS Exchange) of this MA. Such employee deductions by the City shall be used towards the City's CalPERS required contributions.
- 37.11.2 The parties recognize that the CalPERS 20516 amendment that allows employee contributions toward the employer rate is in addition to the required 50% of the normal cost of "New Members" benefits and made in consideration of additional salary increases in Section 11.1.5 (July 3, 2016 -Salary/PERS Exchange) above (a total of 1% salary increases in exchange for employees paying a total of 1.25% towards CalPERS pension cost).

As soon as administratively possible, following council approval of this agreement, the additional contribution of 1.25%, as indicated in Section 37.11.1, shall be eliminated through a contract amendment with CalPERS. The start of the first full pay period following completion of the contract amendment shall be the effective date of the reduction.

For the City of Berkeley For IBEW, Local 1245 DocuSianed by: DocuSigned by: Dawud Brewer Charley Souders 7/20/2022 7/20/2022 **Dawud Brewer Charley Souders** Date Date **Employee Relations Manager Business Representative**